Mathematical Engineering, Manufacturing, and Management Sciences



Advances in Sustainable Machining and Manufacturing Processes

^{Edited by} Kishor Kumar Gajrani Arbind Prasad Ashwani Kumar

DR-.304 04 DR-5 .095 08 DR-383

21 51 DR-

Advances in Sustainable Machining and Manufacturing Processes

Mathematical Engineering, Manufacturing, and Management Sciences

Series Editor: Mangey Ram, Professor, Assistant Dean (International Affairs), Department of Mathematics, Graphic Era University, Dehradun, India

The aim of this new book series is to publish the research studies and articles that bring up the latest development and research applied to mathematics and its applications in the manufacturing and management sciences areas. Mathematical tool and techniques are the strength of engineering sciences. They form the common foundation of all novel disciplines as engineering evolves and develops. The series will include a comprehensive range of applied mathematics and its application in engineering areas such as optimization techniques, mathematical modelling and simulation, stochastic processes and systems engineering, safety-critical system performance, system safety, system security, high assurance software architecture and design, mathematical modelling in environmental safety sciences, finite element methods, differential equations, reliability engineering, etc.

Differential Equations in Engineering

Research and Applications Edited by Nupur Goyal, Piotr Kulczycki, and Mangey Ram

Sustainability in Industry 4.0

Challenges and Remedies Edited by Shwetank Avikal, Amit Raj Singh, and Mangey Ram

Applied Mathematical Modeling and Analysis in Renewable Energy

Edited by Manoj Sahni and Ritu Sahni

Swarm Intelligence: Foundation, Principles, and Engineering Applications

Abhishek Sharma, Abhinav Sharma, Jitendra Kumar Pandey, and Mangey Ram

Advances in Sustainable Machining and Manufacturing Processes

Edited by Kishor Kumar Gajrani, Arbind Prasad, and Ashwani Kumar

Advanced Materials for Biomechanical Applications *Edited by Ashwani Kumar, Mangey Ram, and Yogesh Kumar Singla*

Biodegradable Composites for Packaging Applications *Edited by Arbind Prasad, Ashwani Kumar, and Kishor Kumar Gajrani*

Computing and Stimulation for Engineers

Edited by Ziya Uddin, Mukesh Kumar Awasthi, Rishi Asthana, and Mangey Ram

For more information about this series, please visit: https://www.routledge.com/ Mathematical-Engineering-Manufacturing-and-Management-Sciences/book-series/ CRCMEMMS

Advances in Sustainable Machining and Manufacturing Processes

Edited by Kishor Kumar Gajrani Arbind Prasad Ashwani Kumar



CRC Press is an imprint of the Taylor & Francis Group, an **informa** business

Cover image: Getty

First edition published 2022 by CRC Press 6000 Broken Sound Parkway NW, Suite 300, Boca Raton, FL 33487-2742

and by CRC Press 4 Park Square, Milton Park, Abingdon, Oxon, OX14 4RN

CRC Press is an imprint of Taylor & Francis Group, LLC

© 2022 selection and editorial matter, Kishor Kumar Gajrani, Arbind Prasad and Ashwani Kumar, individual chapters, the contributors

Reasonable efforts have been made to publish reliable data and information, but the author and publisher cannot assume responsibility for the validity of all materials or the consequences of their use. The authors and publishers have attempted to trace the copyright holders of all material reproduced in this publication and apologize to copyright holders if permission to publish in this form has not been obtained. If any copyright material has not been acknowledged please write and let us know so we may rectify in any future reprint.

Except as permitted under U.S. Copyright Law, no part of this book may be reprinted, reproduced, transmitted, or utilized in any form by any electronic, mechanical, or other means, now known or hereafter invented, including photocopying, microfilming, and recording, or in any information storage or retrieval system, without written permission from the publishers.

For permission to photocopy or use material electronically from this work, access www.copyright.com or contact the Copyright Clearance Center, Inc. (CCC), 222 Rosewood Drive, Danvers, MA 01923, 978-750-8400. For works that are not available on CCC please contact mpkbookspermissions@tandf.co.uk

Trademark notice: Product or corporate names may be trademarks or registered trademarks and are used only for identification and explanation without intent to infringe.

ISBN: 978-1-032-08165-6 (hbk) ISBN: 978-1-032-25687-0 (pbk) ISBN: 978-1-003-28457-4 (ebk)

DOI: 10.1201/9781003284574

Typeset in Times by SPi Technologies India Pvt Ltd (Straive) This book is dedicated to all mechanical, production, manufacturing, and aerospace engineers



Contents

Preface	xi
Acknowledgment	xiii
Editors	xv
Contributors	xvii
Introduction	xxi

PART I Sustainable Machining

Chapter 1	Challenges in Machining of Advanced Materials
	M. N. Mathabathe and A. S. Bolokang
Chapter 2	Machining by Advanced Ceramics Tools: Challenges and Opportunities
	P. Kour, S. K. Pradhan, Amit Kumar, and Manoranjan Kar
Chapter 3	Characterization and Evaluation of Eco-Friendly Cutting Fluids 31
	Phaneendra Kiran Chaganti and Madan Mohan Reddy Nune
Chapter 4	Advances in Textured Cutting Tools for Machining51
	Anand C. Petare, Neelesh Kumar Jain, I. A. Palani, and S. Kanmani Subbu
Chapter 5	Advances in MQL Machining71
	Milon Selvam Dennison, Kirubanidhi Jebabalan, M. Neelavannan, and M. Abisha Meji
Chapter 6	Nanofluids Application for Cutting Fluids91
	Mechiri Sandeep Kumar and V. Vasu
Chapter 7	Nanofluids for Machining in the Era of Industry 4.0 121
	Wei Li, Ahmed Mohamed Mahmoud Ibrahim, Yinghui Ren, and Ahmed Moustafa Abd-El Nabi

Chapter 8	Ionic Liquids as a Potential Sustainable Green Lubricant for Machining in the Era of Industry 4.0
	Upendra Maurya and V. Vasu
Chapter 9	Sustainable Electrical Discharge Machining Process: A Pathway
	Satish Mullya and Tribeni Roy
Chapter 10	Sustainable Abrasive Jet Machining173
	Şenol Bayraktar and Cem Alparslan
Chapter 11	Artificial Neural Networks for Machining
	Şenol Bayraktar and Cem Alparslan
Chapter 12	Machining and Vibration Behavior of Ti-TiB Composites Processed through Powder Metallurgy Techniques
	M. Selvakumar, T. Ramkumar, S. Vinoth Kumar, P. Chandramohan, and R. Ganesh
Chapter 13	Numerical Analysis of Machining Forces and Shear Angle during Dry Hard Turning215
	Anupam Alok, Amit Kumar, Manjesh Kumar, Manas Das, and Kishor Kumar Gajrani
Chapter 14	Machining Performance Evaluation of Titanium Biomaterial, Ti6Al4V in CNC Cylindrical Turning Using CBN Insert
	Raju Pawade, Nikhil Khatekar, Sameer Ghanvat, and Nitesh Gothe
PART II	Manufacturing Processes
Chapter 15	Industrial Internet of Things in Manufacturing251
	Hridayjit Kalita and Kaushik Kumar

Chapter 16	Improvement in Forming Characteristics Resulted in	
	Incremental Sheet Forming	263
	Saurabh Thakur and Parnika Shrivastava	

Contents

Chapter 17	Deformation Mechanism of Polymers, Metals, and Their Composites in Dieless Forming Operations	277
	Saurabh Thakur and Parnika Shrivastava	
Chapter 18	Sustainable Polishing of Directed Energy Deposition– Based Cladding Using Micro-Plasma Transferred Arc Pravin Kumar, Neelesh Kumar Jain, and Abhay Tiwari	289
Index		303



Preface

Humans impact the environment in numerous ways. Human civilization needs a healthy environment to ensure the survival of our planet. The balance between environment and technology is the need of today's world. Technology has made our life comfortable, but our environment is paying price. Nowadays, it has been realized that a healthy environment is the necessity of humans' and this planet's survival. Therefore, efforts are increasing toward adapting green and sustainable technology to reduce detrimental environmental impacts.

Sustainable machining and manufacturing processes are the need of today's world. Among various manufacturing processes, machining is one of the widely used. Hence, it has to be made sustainable. The chapters in this book are categorized under two broad sections: (1) sustainable machining and (2) manufacturing processes. Part I, includes the work of numerous researchers as a review of modeling and experimental work. Chapter 1 discusses challenges in machining advanced materials using conventional and nonconventional machining processes. Chapter 2 covers the challenges and probable opportunities during machining by advanced ceramic materials. Chapter 3 discusses various ways to characterize and evaluate eco-friendly cutting fluids. Chapter 4 covers dry machining using advanced textured cutting tools. Chapter 5 discusses advances in one of the important techniques of near-dry machining, namely, minimum quantity lubrication, its need, its significance, and economics and environmentally friendly ways for machining. Chapter 6 covers the application of nanofluids as cutting fluids during machining. Chapter 7 discusses using nanofluids for machining in the era of Industry 4.0 and its effect on environmental sustainability. Chapter 8 explores the use of ionic liquids as a potential sustainable green lubricant for machining in the era of Industry 4.0. The structure of various ionic liquids, their relative machining performance, and their overall environmentally sustainable aspects are covered. Chapter 9 discusses sustainable electrical-discharge machining and using sustainable dielectric while maintaining similar accuracy and precision during the process. Chapter 10 covers the effects of water jet pressure, flow rate, standoff distance, and abrasive grit size on depth of penetration, cutting rate, surface roughness, taper cut ratio, and top kerf width during sustainable abrasive jet machining. Chapter 11 explores the use of artificial neural networks to successfully predict various responses, such as surface roughness, cutting force, and tool wear, during machining. Chapter 12 discusses the machining and vibration behavior of Ti-TiB composites processed through powder metallurgy techniques. In Chapter 13, the numerical analysis of machining forces and shear angle during dry hard turning of AISI 4640 steel using Al₂O₃-coated tungsten-based cemented carbide cutting inserts is conducted to predict cutting force and shear angle. Chapter 14 explores the machining performance evaluation of titanium biomaterial alloys in computer numerical control turning using a cubic boron nitride tool insert.

Part II of this book discusses sustainability aspects in various manufacturing processes. Chapter 15 discusses the use of Industrial Internet of Things in manufacturing, various communication protocols, data management techniques, and software design models focusing on the Fourth Industrial Revolution. Chapter 16 explores the ways to improve forming characteristics in incremental sheet forming. The chapter aims to highlight and systematically review the recent strategies related to numerical techniques, such as finite element analyses, computer-aided design, tool path development, experimental setups, and hybrid techniques that have been proved to increase the quality of the formed ISF parts. Chapter 17 covers the similarities and differences in deformation mechanisms of polymers, metals, and their composites in dieless forming operations. Chapter 18 discusses the sustainable polishing of directed energy deposition–based cladding using micro-transferred arc.

This book will work as a reference book for researchers, practicing machine shop engineers, and managers. This book can also be used as a textbook for the postgraduate level and as an elective course book for the undergraduate level. *Advances in Sustainable Machining and Manufacturing Processes* provides a foundational link to more specialized research work in the domain of sustainable manufacturing.

Dr. Kishor Kumar Gajrani Dr. Arbind Prasad Dr. Ashwani Kumar

Acknowledgment

We would like to thank all the authors of various book chapters for their contributions. Our heartfelt gratitude to CRC Press (Taylor & Francis Group) and the editorial team for their support during the completion of this book. We are sincerely grateful to reviewers for their suggestions and illuminating views on each book chapter presented in book *Advances in Sustainable Machining and Manufacturing Processes*.



Editors

Dr. Kishor Kumar Gajrani is Assistant Professor in the Department of Mechanical Engineering at the Indian Institute of Information Technology, Design and Manufacturing, Kancheepuram, Chennai, India. He has an M.Tech and a PhD degree from the Department of Mechanical Engineering at the Indian Institute of Technology Guwahati. Thereafter, he worked as a postdoctoral researcher at the Indian Institute of Technology Bombay. He has been published in 27 international journals and book chapters of repute, as well as attended numerous conferences. His research interests are broadly related to the advancement of sustainable machining processes, advanced materials, tribology, coatings, green lubricants and coolants, and food packaging.

Dr. Arbind Prasad has completed his PhD in mechanical engineering from the Indian Institute of Technology Guwahati, Assam, India. He has filled four patents out of his research work. He has numerous international journal articles, book chapters, and reputed conference papers to his credit. Dr. Arbind Prasad has obtained various prestigious awards, such as the Sponsored Research Industrial Consultancy (SRIC) award from IIT Kanpur, the Best Oral Presentation from the American Chemical Society, and Best Paper awards from IIT Guwahati during the Research Conclave. He has been invited to deliver talks at various organizations of repute. He has coordinated various faculty development programs, short-term courses, symposiums, and national seminars and has completed research projects sponsored under various government schemes in India. He is currently working as Assistant Professor and Head (Mechanical Engineering) in the Department of Science and Technology, Government of Bihar, posted at Katihar Engineering College, Katihar, Bihar, India. His main areas of interest include manufacturing, machining, polymer composites, biomaterials, materials processing, and orthopedic biomedical applications.

Dr. Ashwani Kumar received a PhD (mechanical engineering) in the area of mechanical vibration and design. He is currently working as Senior Lecturer, Mechanical Engineering (Gazetted Officer Class II) at Technical Education Department, Uttar Pradesh (Government of Uttar Pradesh) India since December 2013. He has worked as Assistant Professor in the Department of Mechanical Engineering, Graphic Era University, Dehradun, India, from July 2010 to November 2013. He has more than 11 years of research and academic experience in mechanical and materials engineering. He is the series editor of book series *Advances in Manufacturing, Design and Computational Intelligence Techniques* published by CRC Press (Taylor & Francis). He is the associate editor for the *International Journal of Mathematical, Engineering and Management Sciences*, indexed in ESCI/Scopus and the Directory of Open Access Journals.

He is an editorial board member of four international journals and acts as a review board member of 20 prestigious (indexed in SCI/SCIE/Scopus) international journals with high impact factors, such as *Applied Acoustics*, *Measurement*, the *Journal* of Engineering Science and Technology, the American Journal of Science & Engineering, Journal of Mechanical Engineering, and the Latin American Journal of Solids and Structures. In addition, he has published 85 research articles in journals, book chapters and conferences. He has authored/co-authored cum edited SV-JME of mechanical and materials engineering. He is associated with international conferences as an invited speaker/advisory board/review board member. He has delivered many invited talks in webinars, faculty development program, and workshops. He has been awarded as Best Teacher for excellence in academics and research. He has successfully guided 12 B.Tech, M.Tech, and PhD candidates through their theses.

In administration he is working as coordinator for the All India Council for Technical Education, extension of approval (E.O.A.), a nodal officer for PMKVY-TI Scheme (Government of India), and internal coordinator for CDTP scheme (Government of Uttar Pradesh). He is currently involved in the research area of machine learning, advanced materials, machining and manufacturing techniques, biodegradable composites, heavy vehicle dynamics, and Coriolis mass flow sensors.

Contributors

Ahmed Moustafa Abd-El Nabi

Production Engineering and Mechanical Design Department Faculty of Engineering Minia University Minia, Egypt

Anupam Alok

National Institute of Technology Patna Patna, India

Cem Alparslan

Faculty of Engineering and Architecture Department of Mechanical Engineering Recep Tayyip Erdoğan University Rize, Turkey

Şenol Bayraktar

Faculty of Engineering and Architecture Department of Mechanical Engineering Recep Tayyip Erdoğan University Rize, Turkey

A. S. Bolokang

Council for Scientific and Industrial Research Manufacturing Cluster, Advanced

Materials Engineering Pretoria, South Africa

Phaneendra Kiran Chaganti

Department of Mechanical Engineering Birla Institute of Technology and Science (BITS-Pilani) Hyderabad, India

P. Chandramohan

Department of Mechanical Engineering Sri Ramakrishna Engineering College Coimbatore, India

Manas Das

Indian Institute of Technology Guwahati, India

Milon Selvam Dennison

Kampala International University Western Campus Uganda

Kishor Kumar Gajrani

Department of Mechanical Engineering Indian Institute of Information Technology, Design and Manufacturing Kancheepuram, India

and

Centre for Smart Manufacturing Indian Institute of Information Technology, Design and Manufacturing Kancheepuram, India

R. Ganesh

Department of Production Engineering Dr. Mahalingam College of Engineering and Technology Pollachi, India

Sameer Ghanvat

Department of Mechanical Engineering Dr. Babasaheb Ambedkar Technological University Lonere, India

Nitesh Gothe

Department of Mechanical Engineering Dr. Babasaheb Ambedkar Technological University Lonere, India

Ahmed Mohamed Mahmoud Ibrahim

College of Mechanical and Vehicle Engineering Hunan University Changsha, China and Faculty of Engineering Production Engineering and Mechanical Design Department Minia University Minia, Egypt

Neelesh Kumar Jain

Department of Mechanical Engineering Indian Institute of Technology Indore Indore, India

Kirubanidhi Jebabalan

Division of Materials Technology Technical University of Liberec Czech Republic

Hridayjit Kalita

Birla Institute of Technology Ranchi, India

Manoranjan Kar

Department of Physics Indian Institute of Technology Patna Patna, India

Nikhil Khatekar

Department of Mechanical Engineering PCT's A. P. Shah Institute of Technology Thane, India

P. Kour

Department of Physics Birla Institute of Technology Mesra Patna, India

Amit Kumar

Department of Mechanical Engineering National Institute of Technology Patna Patna, India Kaushik Kumar Birla Institute of Technology Ranchi, India

Manjesh Kumar

Indian Institute of Technology Guwahati, India

Mechiri Sandeep Kumar

B V Raju Institute of Technology Narsapur, India

Pravin Kumar

Department of Mechanical Engineering Indian Institute of Technology Indore Simrol, 453 552 (MP), India

S. Vinoth Kumar

Key Laboratory of High-Efficiency and Clean Mechanical Manufacture National Demonstration Center for Experimental Mechanical Engineering Education School of Mechanical Engineering Shandong University Jinan, China

Wei Li

College of Mechanical and Vehicle Engineering Hunan University Changsha, China

M. N. Mathabathe

Council for Scientific and Industrial Research Manufacturing Cluster, Advanced Materials Engineering Pretoria, South Africa

M. Abisha Meji

Kampala International University Western Campus Uganda

Contributors

Satish Mullya

Annasaheb Dange College of Engineering and Technology, Ashta, India

M. Neelavannan

Dr. Mahalingam College of Engineering and Technology Coimbatore, India

Madan Mohan Reddy Nune

Department of Mechanical Engineering Anurag University Hyderabad, India

Upendra Maurya

Department of Mechanical Engineering National Institute of Technology Warangal, India

Raju Pawade

Department of Mechanical Engineering Dr. Babasaheb Ambedkar Technological University Lonere, India

Anand C. Petare

Central Workshop Indian Institute of Technology Indore Indore, India

I. A. Palani

Department of Mechanical Engineering Indian Institute of Technology Indore Indore, India

S. K. Pradhan

Department of Mechanical Engineering Birla Institute of Technology Mesra Patna, India

T. Ramkumar

Department of Mechanical Engineering Dr. Mahalingam College of Engineering and Technology Pollachi, India

Yinghui Ren

College of Mechanical and Vehicle Engineering Hunan University Changsha, China

Tribeni Roy

Birla Institute of Technology and Science Pilani (BITS-Pilani) Pilani, India

and

School of Engineering London South Bank University London, United Kingdom

M. Selvakumar

Department of Automobile Engineering Dr. Mahalingam College of Engineering and Technology Pollachi, India

Parnika Shrivastava

Department of Mechanical Engineering National Institute of Technology Hamirpur, India

S. Kanmani Subbu

Department of Mechanical Engineering Indian Institute of Technology Palakkad Palakkad, India

Saurabh Thakur

Department of Mechanical Engineering National Institute of Technology Hamirpur, India

Abhay Tiwari

Department of Mechanical Engineering Indian Institute of Technology Indore Simrol, MP, India

V. Vasu

Department of Mechanical Engineering National Institute of Technology Warangal, India



Introduction

The ever-increasing trends of upcoming technology also lead to many concerns related to the environment, society, and economics in every field of engineering. Sustainability in any organization implies that something can be sustained indefinitely. Recently, governments and world leaders have been advocating for green and sustainable manufacturing initiatives. To support and promote such initiatives, researchers, scientists, engineers, and academic institutions have a responsibility to introduce educational programs related to sustainable manufacturing to prepare the future generation. Sustainable machining and manufacturing processes are a need of today's world. *Sustainable machining* is defined as the creation of products by cutting material that uses processes that are environmentally friendly, economically sound, and safe for employees, consumers, and communities, as well as can conserve energy.

To introduce these multidimensional machining and manufacturing processes into industries, as well as into the curriculum of future generations of researchers and engineers, is the ambition behind this book. This goal can be balanced by developing adequate economic, environmental, and social criteria, with analysis of their independencies and application of that analysis for guiding technological innovation in respective economic, environmental, and societal frameworks.

This book provides a lucid way for readers to understand the advances in sustainable techniques for machining and manufacturing applications. The book consists of 18 chapters dedicated to the advances in sustainable machining and manufacturing processes. The chapters discuss the challenges faced when machining advanced materials, the use of eco-friendly cutting fluids, and how they affect the machined component characteristics and our environment. This book also covers topics such as dry and near-dry machining, machining with advanced textured cutting tools, minimum quantity lubrication, nanofluids and ionic fluids in the era of Industry 4.0, sustainable electrical-discharge machining, abrasive jet machining, and artificial neural network–based machining. This book includes machining and vibration behavior of composites and finite element analysis during machining with advanced ceramic coated carbide tools. Furthermore, the use of the Industrial Internet of Things in manufacturing, sheet forming, and deformation mechanism, as well as sustainable polishing using micro-plasma transferred arc, is discussed.

This book addresses the challenges and solutions for sustainable machining and manufacturing processes. It discusses prevailing trends and suggests research findings for industries to move toward sustainable development by improving economic and social perspectives, as well as reducing the detrimental effects to the environment. Overall, the aim of this book is to catalogue the latest achievements in the modern machining and manufacturing industry that can be helpful for future generations.

> Dr. Kishor Kumar Gajrani Dr. Arbind Prasad Dr. Ashwani Kumar



Part I

Sustainable Machining



1 Challenges in Machining of Advanced Materials

M. N. Mathabathe and A. S. Bolokang

Council for Scientific and Industrial Research, Manufacturing Cluster, Advanced Materials Engineering, Pretoria, South Africa

CONTENTS

1.1	Introduction				
1.2	1.2 Machining Process and Materials				
	1.2.1 Cutting Tool	4			
	1.2.2 Material Selection	4			
	1.2.3 Types of Machining Techniques	5			
1.3	Tool Wear/Life Span and Commercial Metal Cutting	5			
1.4	Machinability				
1.5	Machining Process Selection	8			
	1.5.1 Challenges Related to Machining	8			
	1.5.2 Practical Aspects and Developments	8			
1.6	Conclusion				
Refe	erences				

1.1 INTRODUCTION

The fuel efficiency and performance improvement in the aerospace application has led to the development of advanced materials, namely, from steel to nickel to titanium alloys; materials with lower density, for example, aluminum to carbon fiber composites; and ceramic and metal matrix composites (CMCs and MMCs, respectively), which are restituting part of elevated temperature alloys subject to engine application en route to the end of the 20th century [1].

Advanced structural materials can be defined as complex shapes or materials combinations to attain properties linking to functionality, for example, smart materials. These are materials designed for good mechanical, electrical, or thermal properties; high-efficiency energy conversion; materials with embedded sending systems for reliability and safety; and smart materials vehicles and large space structures are subject to a high-strength-per-mass ratio [2]. Researchers studied the design and fabrication of specific structures to enhance materials property three-dimensional (3D) printed structures, such as curved [3], honeycomb [4], cell shapes [5] or hexachirales [6].

The thermal, mechanical and electrical properties, which also enhance the performance of polymer matrix composites, make carbon nanotubes (CNTs) attractive for industry use. However, identifying prospective applications for space use is a challenge for these materials [7]. On one hand, the chief challenges formed by today's establishments involve comprehending and assimilating the rapidly advancing technological machining methods fabricated by industry. The integration and application of these advances prepare the industry for the next set of leading technical improvements. The present work focuses on a vast area of machining techniques and their challenges faced by advanced materials.

1.2 MACHINING PROCESS AND MATERIALS

1.2.1 CUTTING TOOL

Advanced materials, such as superalloys, are a challenge for machining due to their inherent high hardness, low thermal conductivity, and great resistance to shearing. However, they require elevated cutting temperatures and high cutting forces to avoid severe tool wear [8].

The major tools employed in the fabrication of Ni-based superalloys are ceramic, carbide tools, and polycrystalline cubic boron nitride (PCBN). The latter is commonly used due to its superior machining capabilities for "difficult to cut" materials [9]. The PCBN is designed without a chip breaker, which is a flat rake, resulting in more resistance to breaking the chip during the cutting process. However, its short-comings are that the surface of the workpiece is effortlessly damaged by the chip winding workpiece [9]. It was suggested that applying high-pressure cooling may overcome these drawbacks [10–11], while the importance of tool failure is attributed to analyzing the damage of tool material [12].

1.2.2 MATERIAL SELECTION

Advanced ceramic materials, for example, are clustered into two groups, that is, (1) conductive—such as the typical zirconium diboride (ZrB₂), metal nitrides (TiN/ZrN), boron carbide (B₄C), titanium diboride (TiB₂), and other similar materials—and (2) nonconductive silicon nitride (Si₃N₄), alumina (Al₂O₃), zirconia (ZrO₂), and silicon carbide (SiC). However, adding advanced ceramic materials or conductive particles to the latter, such as B₄C, TiC, Si, CaO, TiB₂, Si₃N₄ + TiC, Si₃N₄ + TiN, ZrO₂ + Cao, Al₂O₃ + TiC, ZrO₂ + Y₂O₃ + TiN, and Al₂O₃ + TiN, makes them conductive advanced materials [13–14]. However, the conventional machining shortcomings of advanced ceramic materials are attributable to their elevated hardness and brittleness [13]. Additionally, their processing/manufacturing performance is not cost-effective, particularly the expense incurred during the polishing stage [13]. Therefore, the surface of advanced materials may suffer damage during machining, resulting in stress concentration and cracks, thus affecting the mechanical strength of components [14].

Ceramic phases, that is, Al₂O₃/Al₂O₃ or SiC/SiC, in the same structure of CMCs mitigate the aggregate residual stresses emanating from the manufacturing process. Further, the coefficient of thermal expansion (CTE) mismatch and processing

temperature affects the residual stresses amid matrix and fibers, characterized by X-ray diffraction (XRD), Raman spectroscopy, or micro-hardness tests [15].

1.2.3 Types of Machining Techniques

Many machining techniques effectuated unrelenting tool wear associated with longer cutting time, the high cutting force resulting in higher machining cost, thus bearing machining challenges of advanced materials [12]. Furthermore, when processing advanced ceramics, for example, by abrasive machining and grinding, the product components experience plastic deformation and considerable residual stress, friable layers, and surface cracks [16, 17]. Table 1.1 outlines some machining techniques, both conventional and nonconventional.

The susceptible factors in the performance of electrical-discharge machining (EDM) are a low material removal rate (MRR) and high tool wear, which may be conquered by other advantages when compared with other nonconventional techniques [18].

On one hand, dry machining is conducted without the facilitation of cutting fluids. This process has gained widespread over the past years and is increasingly used by manufacturers that fabricate products with metals [19]. However, the technique exhibits drawbacks, such as (1) equipped for materials workpiece and machining approaches, which do not yield favorable results; (2) elevated temperatures at the tool–workpiece interface on the cutting edge; (3) the tool heating up due to the amplified temperature and losing its hardness [19]; (4) the surface integrity and dimensional accuracy of the workpiece are altered [20]; and (5) only materials that pose good machinability can be employed for dry machining [21].

1.3 TOOL WEAR/LIFE SPAN AND COMMERCIAL METAL CUTTING

Some advanced materials, such as the 40CrNi₂Si₂MoVA (300 M) steel, employed in aerospace because of its good strength, fracture toughness, elevated transverse plasticity, good corrosion resistance, and fatigue performance have their inherent drawbacks. The 300 M steel suffers poor thermal conductivity, resulting in a large cutting force and an elevated temperature, difficult chip control, and easy tool wear during the cutting process [30]. However, Zhang et al. [31] used cryogenic minimum quantity of lubricant (CMQL) technology to analyze the varying tool wear lubricants and cutting force amid high-speed turning of 300 M steel. The results indicated that this method reduced tool wear and prolonged tool life by the reduction of cutting force, cutting temperature at the knifepoint and friction amid the tool and workpiece. Furthermore, the authors studied the genetic algorithm under CMQL conditions which optimized cutting parameters and presented theoretical attribution for tool wear control at high-speed cutting 300 M steel. The ultimate tool wear prediction is demonstrated in Equation 1.1:

$$VB = 0.0362v^{0.1054} f^{0.0567} a_p^{0.0119}, (1.1)$$

TABLE 1.1Feature/Operation of Ceramic Metal Matrix Composites on the Machining Techniques

Machining Techniques Conventional	Information Machined	Challenges	Concluding Remarks	References
Orthogonal cutting	 Defined cutting edges Fiber-reinforced orientations, i.e. parallel, across, and transverse 	Ductile and brittle behavior transpire due to ceramic machining at small uncut chip thickness leading to grain fracture and slip-plane mechanisms	Roughness/coarseness of machined surface is susceptible to microcracking mechanisms of the particles, leading to residual stress as the scratch load increases	[22]
Milling	 Surface finish – roughness and morphology 	Observed ductile to brittle transition in the matrix during machining. Fiber removal due to pullout mechanism.	Concluded that surface roughness increased at a penetration depth of $>4 \mu m$, resulting in larger grooves due to brittle fracture	[23]
Drilling	The tool rotates along its axis	Entry and exit delaminations are induced machining damage due to high thrust forces employed-conventional drilling (CD)	Rotary ultrasonic machining (RUM) had an average reduced thrust force (~10%–15%), resulting in less significant exit delamination than CD	[24]
Grinding	Favored finishing operation for hard/brittle materials to achieve dimensional accuracies	Three grinding methods showed different results performed on C/Si materials CG provided a surface roughness (Ra) ~2–4 times lower than IG, while the UAG produced much higher values of Ra, as a result of the induced impact on the abrasive grains -caused cracks to propagate	Grinding holes >1-mm diameter are useful for successful machining of slots and surface finish employing the typical conventional mechanical techniques using cubic boron nitride and diamond tools	[25]
Nonconventional Abrasive waterjet (AWJ)	Cut and shape hard metals	Reduced surface quality at the jet exit because of kerf taper angle effect and brittle fracture removal due to loss of energy of the jet	The technique can be effective in machining slots, holes, and through cuts in Al ₂ O ₃ /Al ₂ O ₃ and SiC/ SiC CMCs, provided the operating parameters are optimized	[26]
Pulsed laser ablating (PLA)	Hole making in ultra-hard materials	The main drawback is heat-affected zone (HAZ)	Minimization of thermal damage, the PLA is recommended provided the pulse duration is enhanced, i.e., milli-, nano-, pico-, or femtosecond laser ablation.	[27, 28]
Electrical-discharge machining (EDM)	Cuts and holes in electrically conductive materials difficult to cut	Temperature built up due to process sparks, affecting the machined surface resulting in residual stress levels amid the fibers and matrix.	Debris removal is important to prevent damage to the workpiece. Additionally, deep flushing and tool vibration enhanced surface quality.	[29]

where VB is the amount of flank wear and v, f, and a_p are the cutting speed, feed, and depth of cutting, respectively. However, if the influence of a factor is not significant on the dependent variable, then the coefficient of the factor should be zero, that is, bi = 0, utilizing a t test as shown in Equation 1.2:

$$ti = \frac{bi / \sqrt{cii}}{\sqrt{Q / (n - m - 1)}}, i = 1, 2, \dots, m.$$
(1.2)

The tool wear coefficient level as determined by Zhang et al. [31] was $0.05(\alpha = 0.05)$, and $t(\alpha/2) (n - p - 1)$, and the ultimate results showed that t0.025(n - p - 1) = 2.22814. This indicated the significant effect of the cutting speed on the forecast value, with subsequent feed, and the depth of cutting.

To achieve economic tool life with subsequent surface conditions, cutting parameters must be carefully chosen. For example, on powder metal Ni alloys [32], these are generally

- Strain < 0.01 mm,
- Surface roughness <0.8 µm,
- Nonparent material required,
- · No redeposited material or layer, and
- No light contrast amorphous or recast layer.

For the cutting process completion, cutting strategies, namely, cutting parameters and tool geometries, must be properly analyzed. Wear evaluation is subject to an assessment conducted on cutting tools that are worn displaying wear features. In particular, flank wear and chipping of cutting edges cannot be utilized to measure tool wear according to ISO 3685 standard [32]. However, Abele et al. [33] employed statistical experiments/five-axis milling that involved a merger between the flank or tip and the cutting tool radius. Key process variables to consider using this method are axial depth of cut, cutting speed or feed rate, and to predict tool wear behavior.

A lot of research in the cutting process covers tool wear of tungsten carbide and carbide tools, but little has been covered on polycrystalline cubic boron nitride (PCBN) tool wear under high-pressure cooling. This was described by [34], that the workpiece deploys increased pressure on the tool flank, while its contact area is small, resulting in the flank being worn out. However, the flank face cutting area poses challenges because of insufficient coolant while the wear is escalating affecting the machined surface integrity. A study by Wu et al. [34] demonstrated the wear morphology and profile of the PCBN tool. Conversely, the wear model from the study provided a source to cutting superalloys subjected to cooling at high pressures to minimize machining and tool wear. The model was also subsequently reciprocal to cutting parameters, cutting time, and cooling pressure.

1.4 MACHINABILITY

Machinability of materials is clearly expressed as the effortless practice of materials removal (chips) by cutting tool edge, employing conventional machining operations to yield a suitable cost-effective surface quality [35].

The machinability of NiTi shape memory alloy (SMA) is a challenge due to the existence of intermetallics, resulting in strain hardening effects and poor surface texture [36]. Two varying approaches to machine NiTi SMAs have been adopted by Kong et al. [37], the results indicated that abrasive water jet machining was superior to plain water jet machining regarding a more controlled depth and surface texture. Likewise, Frotscher et al. [38] outlined that abrasive water jet machining was an appropriate machining process compared to the micromachining process, which was able to decrease the thermal effect and cutting time of the machining process. Some other unconventional machining processes such as the wire-electrical-discharge machining (WEDM), laser beam and EDM could be utilized for machining the NiTi alloys but have drawbacks, such as heat-affected zone, microvoids, and recast layer. To overcome these drawbacks, Manjaiah et al. [39] consolidated the parameters of the water jet machining to improve surface roughness and the kerf angle of the composites. It was concluded that the reinforcement of the composite in wt.% improved the integrity of the surface machined by clearing particles from the surface matrix [39]. Table 1.2 shows the conducted machinability studies of some of the advanced materials.

1.5 MACHINING PROCESS SELECTION

Machining process selection and technologies have become demanding for advanced materials. The prerequisites to consider are precision machining and reduced surface roughness (quality), large material removal rate (productivity), decreased tool wear (tool cost). The study reported by Feucht et al. [45] indicated the ultrasonic technology influence and the integrated machining of hard-to-machine advanced materials. Table 1.3 lists the summary of some of the simulated model approaches used for cutting mechanisms.

1.5.1 CHALLENGES RELATED TO MACHINING

Prerequisites in the CMC structures affected by the machining process are elevated tensile or compressive residual stresses, processing temperature, and CTE mismatch [50]. On one hand, the machining of CMCs is challenging because of their (1) brittle behavior, (2) high hardness, (3) heterogeneous structure composed (matrix, fibers, porosities), and (4) orthotropic mechanical and thermal behavior [50].

1.5.2 PRACTICAL ASPECTS AND DEVELOPMENTS

Contemporarily, manufacturing industries are harnessing the utilization of limited quantity lubrication and dry machining, ascribed to their ecological and economic conveniences. The minimum quantity lubrication (MQL) assisted machining is used

TABLE 1.2Machinability Studies on Some of the Advanced Materials

Approach	Material	Application	Type of Electrode/ Cutting Fluids	Remarks	Reference
Electric discharge machining (EDM)	Ti-6Al-4V	Biomedical, automotive, and aerospace	Graphite, aluminum, copper, and brass	Discharge current elevated values cause coarser surface integrity. Unlike other electrodes, graphite resulted in high surface roughness, particularly at elevated discharge current values	[40]
High-pressure cooling (HPC)	Inconel 718	Nuclear reactors, pumps, spacecraft, gas turbines, and rocket motors	Vegetable oil-based	The utility of cutting oils viz. their environmental, economic, and societal pillars combined with the surface texture of tools of the HPC method can enhance the high-speed machinability and productivity of superalloys.	[41]
Wire electrical discharge (WED)	AA2024/Al ₂ O ₃ /BN hybrid composite	Automobile, structural and aerospace industries	Molybdenum	From the optimization method, it was concluded that the pulse on-time makes a significant impact on the desired performance measures during machinability of the composite	[42]
Wire electrical discharge (WED)	Hybrid metal matrix composite (HMMC): Al LM6 as matrix, silicon carbide, and dunite added as reinforcements	Military components, automotive, and aerospace	Brass	The machinability analysis on performance has shown that pulse ON duration is the prevalent variable for achieving the performance measures desired.	[43]
Wire electrical discharge (WED)	Al/AlCoCrFeNiMo _{0.5} MMC	Engineering materials for automotive industry	0.25 mm diameter copper wire	The surface roughness, KW, and MRR were minimal. The lower effectiveness range because of reinforcement over the MMCs machinability is a positive indicator for the contemporary utilization of the novel material.	[44]

KW = kerf width, MRR = material removal rate.

TABLE 1.3Simulated Model Approaches Used for Cutting Mechanisms

Machining Approach	Theoretical Model	Type of Material	Process Parameters	Remarks	Author
PCBN turning	Adaptive genetic algorithm	GH4169 superalloy	Initial population size, i.e., m , P_c , P_m	Surface roughness during the turning process can be estimated, and the maximum error amid measured and predicted value is 0.107	[46]
Femtosecond laser processing	COMSOL software	Diamond microgrooves	Inlet velocity and size cross-section shape with a depth of 1000 microns	It was concluded that rectangular microgrooves have good heat dissipation compared to triangular and trapezoidal structures	[47]
PCBN tooling	Chip breaking model	GH4169 superalloy	f = 0.15 mm/r, $ap = 0.4mm, v_c = 160 m/min$	Bending moment increases, and the crimp radius decreases, leading to a reduction of feed rate and depth of cutting during the high-pressure cooling process. Ultimately, the breaking performance of the chip is improved	[8]
CNC machining	Stereoscopic/spherical geometry method (SGM), and Projection method (PM)	For advanced materials	Position angle θ and orientation angle ϕ of drilling and milling modular fixture	The procedure and principle of solving spatial angle in the modular fixture by SGM, and PM was well executed	[48]
Intelligent machining combined with sensor- based control systems	Interactive search method (ISM), multi-objective genetic algorithm (MOGA), and genetic algorithms (GA)	AISI 1064 steel	Cutting speed, depth of cut, and feed rate	The results showed that ISM exhibited optimal outcomes in the field of manipulation of machining processes parameters	[49]
Abrasive water jet cutting	SPH algorithm and Lagrange model numerical simulation	Q235, X60, X80 and 304 stainless steels	Pressure 40 Mpa, target distance 5 mm, and 80 mesh garnet abrasive	The erosion effect was experienced during cutting due to stress and friction, impact deformation, resulting in the depth of cutting increasing with an increase in cutting pressure.	[18]

where f is the cutting force, ap is the cutting depth, and v_c is the deflection.

in auto parts, for example. It has been used in the crankshaft (drilling oil holes) and the block, which is typically a challenging approach as a result of diameter and large length [19].

A new novel concept of energy efficiency grade evaluation (EEGE) has been formulated by Ma et al. [51]. This approach is developed in various stages: (1) the enlisting of the inherent energy efficiency (IEE), (2) the advancement of quantitative approach in machining systems, and (3) creation of IEE evaluating indicator system from both the inherent specific energy (ISE) and inherent energy utilization. The EEGE method is a new tool for analyzing the energy efficiency of the machining system. However, future works objectives should be on establishing fundamental databases and interval threshold standards and discovering the application of the EEGE method to configure high-energy-efficiency machining systems [51].

Nanomachining technology using electron beam processing was studied by [52]. The structure of the self-organized surface of the nanomaterials was etched by engaging varying etching speeds, forces, depths, and probe cyclic times; however, the proportionality amid the etching depth and force, while the self-organized nanomaterials are progressively raised. The author indicated that selection of suitable parameters is a possibility to produce a linear structure with a width of approximately 60 nm and a depth of approximately 8 nm on the self-organized surface of the nanomaterial and a dotlike structure with a point spacing and height of 70 nm and 4 nm, respectively.

One of the complex systems is intelligent manufacturing systems in whereby enhancement approaches are combined with sensor-based control systems. Researchers have ascertained the viability of sensors transmitting networks with each other during the cutting operation. However, the objective was to devise a procedure that qualifies a CNC lathe spindle and smart feed drive to react to variations in signals fed to them from a series of external sensors, able to detect disturbances throughout the cutting process. Regrettably, no prototype was developed for this proposed system. Instead, this was achieved via a simulation of lathes created to substantiate logic and coding of the sensory communication network [53].

On the other hand, advanced engineering structures with embedded sensors form the basis of progressing attempts in structural health monitoring (SHM) systems [54]. However, there are certain dependencies not required for SHM designers which are treated as noises in the signals created by sensors. Some of these dependencies that affect sensing properties are (1) mechanical and thermal loading, (2) signal processing method, (3) integration configuration between material and surface-bonded, (4) fabrication process utilized for integration, and (5) base metal properties [54]. Albeit, piezoelectric materials offer high durability, providing good sensing mechanisms. Optical sensors, although costly, are efficient in quantifying strain and temperature simultaneously, resulting in overall superior performance. Generally, sensor fabrication is a countless in situ quantification of SHM variables but attributed to laboratory applications, for example, characterization and identification of complex structures in terms of deformation and failure mechanisms. Additionally, the emergence of multimaterial three-dimensional (3D) printing methods eases the integration of sensor materials into 3D printed composites. Consecutively, contributing greater understanding of the deformation interfacial mechanisms of composite materials used in the architecture industry [54].

1.6 CONCLUSION

Advanced materials for current and future aerospace applications, for example, are invented to withstand conditions such as environmental damage (oxidation and corrosion), creep strain, dwell crack growth, elevated temperature yield stress, and microstructure stability, exclusive of rises in cost and density. However, advanced materials, such as the advanced Ni alloys, solicit enhanced process of machining to acquire modified materials mechanical properties, including cost improvement outcome.

Tool failure mechanism research study as part of the challenges faced during machining is prevalent in the comprehension of tool structure and applications. Recently, tool failure exploration included both the macro problems of tool failure and microcrack propagation within the material before the appearance of tool surface crack defects. However, the intense cutting process outlining the damage mechanism of the carbide tool was explored, namely, (1) crack initiation, (2) crack propagation, (3) damage accumulation, and (4) tool breakage.

REFERENCES

- Klocke, F., Soo, S. L., Karpuschewski, B., Webster, J. A., Novovic, D., Elfizy, A., Axinte, D. A., & Tönissen, S. (2015). Abrasive machining of advanced aerospace alloys and composites. *CIRP Annals – Manufacturing Technology*, 64(2), 581–604. https://doi. org/10.1016/j.cirp.2015.05.004.
- Gates, T. S. H. (2003). Computational materials: Modeling and simulation of nanostructured materials and systems. In 44th American Institute of Aeronautics and Astronautics, Structures, Structural Dynamics and Materials Conference, Langley Research Center, Hampton, Virginia, US.(p. 1534).
- Galantucci, L. M., Lavecchia, F., & Percoco, G. (2008). Study of compression properties of topologically optimized FDM made structured parts. *CIRP Annals*, 57, 243–246.
- Abramovitch, H., Burgard, M., Edery-Azulay, L., Evans, K. E., Hoffmeister, M., Miller, W., Scarpa, F., Smith, C. W., & Tee, K. F. (2010). Smart tetrachiral and hexachiral honeycomb: Sensing and impact detection. *Composites Science and Technology*, 70, 1072–1079.
- Miller, W., Smith, C. W., Scarpa, F., & Evans, K. E. (2010). Flatwise buckling optimization of hexachiral and tetrachiral honeycombs. *Composites Science and Technology*, 70, 1049–1056.
- Prall, D., & Lakes, R. S. (1997). Properties of a chiral honeycomb with a poisson's ratio of—1. *International Journal of Mechanical Sciences*, 39, 305–314.
- Vartak, D. A., Satyanarayana, B., Munjal, B. S., Vyas, K. B., Bhatt, P., & Lal, A. K. (2020). Potential applications of advanced nano-composite materials for space payload. *Australian Journal of Mechanical Engineering*, 1–9. https://doi.org/10.1080/14484846. 2020.1733176.
- Wu, M., Li, L., Liu, X., Cheng, Y., & Yu, Y. (2018). Analysis and experimental study of the cutting mechanism in machined nickel-base superalloys GH4169 with PCBN tools under high pressure cooling. *Integrated Ferroelectrics*, 189(1), 105–120. https://doi.org /10.1080/10584587.2018.1456175
- Mamalis, A. G., Kundrák, J., & Horváth, M. (2002). Wear and tool life of CBN cutting tools. *The International Journal of Advanced Manufacturing Technology*, 20(7), 475–479.

- Kramar, D., Sekuli, M., Jurkovi, Z., & Kopa, J. (2013). The machinability of nickelbased alloys in high pressure jet assisted (HPJA) turning. *Metalurgija*, 52(4), 512–514.
- 11. Schwartz, M. (1992). Handbook of structural ceramics. New York: McGraw Hill.
- Tuersley, I. P., Jawaid, A., & Pashby, I. R. (1994). Review: Various methods of machining advanced ceramic materials. *Journal of Materials Processing Technology*, 42(4), 377–390.
- Ming, W., Jia, H., Zhang, H., Zhang, Z., Liu, K., Du, J., Shen, F., & Zhang, G. (2020). A comprehensive review of electric discharge machining of advanced ceramics. *Ceramics International*, 46(14), 21813–21838. https://doi.org/10.1016/j.ceramint.2020.05.207.
- Samant, A. N., & Dahotre, N. B. (2009). Laser machining of structural ceramics a review. *Journal of the European Ceramic Society*, 29(6), 969–993.
- Ayrikyan, A., Prach, O., Khansur, N. H., & Webber, K. G. (2018). Investigation of residual stress in lead-free BNT-based ceramic/ceramic composites. *Acta Materialia*, 148, 432–441.
- Zhang, B., Zheng, X. L., Tokura, H., & Yoshikawa, M. (2003). Grinding induced damage in ceramics. *Journal of Materials Processing Technology*, 132(1), 353–364.
- Kirchner, H. P. (2010). Damage penetration at elongated machining grooves in hotopressed Si₃N₄. *Journal of the American Ceramic Society*, 67(2), 127–132.
- Hua, W., Zhang, W., Li, J., Cai, K., & Jiang, Y. (2020). Feasibility of abandoned ammunition materials cold cutting experimental study and numerical prediction. *Integrated Ferroelectrics*, 207(1), 1–11. https://doi.org/10.1080/10584587.2020.1728660.
- Goindi, G. S., & Sarkar, P. (2017). Dry machining: A step towards sustainable machining – Challenges and future directions. *Journal of Cleaner Production*, 165, 1557–1571. https://doi.org/10.1016/j.jclepro.2017.07.235.
- Klocke, F., Gierlings, S., Brockmann, M., & Veselovac, D. (2011). Influence of temperature on surface integrity for typical machining processes in aero engine manufacture. *Procedia Engineering*, 19, 203–208.
- Thakur, A., & Gangopadhyay, S. (2016). Dry machining of nickel-based super alloy as a sustainable alternative using TiN/TiAlN coated tool. *Journal of Cleaner Production*, 129, 256–268.
- Spriet, P. (2014). CMC applications to gas turbines. In *Ceramic matrix composites: Materials, modeling and technology*. Bansal, N.P., & Lamon, J., Hoboken, NJ: John Wiley & Sons, Inc. (pp. 591–608).
- Yuan, S., Li, Z., Zhang, C., & Guskov, A. (2017). Research into the transition of material removal mechanism for C/SiC in rotary ultrasonic face machining. *The International Journal of Advanced Manufacturing Technology*, 95(5–8), 1751–1761.
- Ding, K., Fu, Y., Su, H., Chen, Y., Yu, X., & Ding, G. (2014). Experimental studies on drilling tool load and machining quality of C/SiC composites in rotary ultrasonic machining. *Journal of Materials Processing Technology*, 214(12), 2900–2907.
- Diaz, O. G., Axinte, D., Butler-Smith, P., & Novovic, D. (2018). On understanding the microstructure of SiC/SiC ceramic matrix composites (CMCs) after a material removal process. *Materials Science and Engineering*, 713(26), 1–11.
- Hashish, M., Kotchon, A., & Ramulu, M. (2015). Status of AWJ machining of CMCS and hard materials. In *International Technical Conference on Diamond, Cubic Boron Nitride and Their Applications*. Indianapolis, IN (pp. 1–14).
- Tuersley, I., & Hoult, T. (1998). The processing of SiC SiC ceramic matrix composites using a pulsed Nd-YAG laser: Part II the effect of process variables. *Journal of Materials Science*, 3, 963–967.
- Li, W., Zhang, R., Liu, Y., Wang, C., Wang, J., Yang, X., & Cheng, L. (2016). Effect of different parameters on machining of SiC/SiC composites via pico-second laser. *Applied Surface Science*, 364, 378–387.
- Wei, C., Zhao, L., Hu, D., & Ni, J. (2013). Electrical discharge machining of ceramic matrix composites with ceramic fiber reinforcements. *The International Journal of Advanced Manufacturing Technology*, 64(1–4), 187–194.
- Huiping, Z., Hongxia, Z., & Yinan, L. (2014). Surface roughness and residual stresses of high speed turning 300 M ultrahigh strength steel. *Advances in Mechanical Engineering*, 2014, 1–7. https://doi.org/10.1155/2014/859207.
- Zhang, H. P., Zhang, Z., Zheng, Z. Y., & Liu, E. (2020). Tool wear in high-speed turning ultra-high strength steel under dry and CMQL conditions. *Integrated Ferroelectrics*, 206(1), 122–131. https://doi.org/10.1080/10584587.2020.1728633.
- Kappmeyer, G., Hubig, C., Hardy, M., Witty, M., & Busch, M. (2012). Modern machining of advanced aerospace alloys-Enabler for quality and performance. *Procedia CIRP*, *1*(1), 28–43. https://doi.org/10.1016/j.procir.2012.04.005.
- 33. Abele, E., Lieder, M., Hölscher, R., & Hubig, C. (2012). Qualitative description of tool wear characteristic. *Wt-Online*.
- Wu, M., Wu, S., Chu, W., Liu, K., & Cheng, Y. (2020). Experimental study on tool wear in cutting superalloy under high-pressure cooling. *Integrated Ferroelectrics*, 207(1), 208–219. https://doi.org/10.1080/10584587.2020.1728681.
- Alabdullah, M., Polishetty, A., Nomani, J., & Littlefair, G. (2019). An investigation on machinability assessment of Al-6XN and AISI 316 alloys: An assessment study of machining. *Machining Science and Technology*, 23(2), 171–217. https://doi.org/10.108 0/10910344.2018.1486415.
- Lin, H. C., Lin, K. M., & Chen, Y. C. (2000). A study on the machining characteristics of TiNi shape memory alloys. *Journal of Materials Processing Technology*, 105, 327–332.
- Kong, M. C., Axinte, D., & Voice, W. (2011). Challenges in using waterjet machining of NiTi shape memory alloys: An analysis of controlled depth milling. *Journal of Materials Processing Technology*, 211(6), 959–971.
- Frotscher, M., Kahleyss, F., Simon, T., Biermann, D., & Eggeler, G. (2011). Achieving small structures in thin NiTi sheets for medical applications with water jet and micro machining: A comparison. *Journal of Materials Engineering and Performance*, 15, 776–782.
- Manjaiah, M., Narendranath, S., Basavarajappa, S., & Gaitonde, V. N. (2014). Wire electric discharge machining characteristics of titanium nickel shape memory alloy. *Transactions of Nonferrous Metals Society of China (English Edition)*, 24(10), 3201–3209.
- Ahmed, N., Ishfaq, K., Moiduddin, K., Ali, R., & Al-Shammary, N. (2019). Machinability of titanium alloy through electric discharge machining. *Materials and Manufacturing Processes*, 34(1), 93–102. https://doi.org/10.1080/10426914.2018.1532092.
- Mahesh, K., Philip, J. T., Joshi, S. N., & Kuriachen, B. (2021). Machinability of Inconel 718: A critical review on the impact of cutting temperatures. *Materials and Manufacturing Processes*, 36(7), 753–791. https://doi.org/10.1080/10426914.2020.184 3671.
- Naidu, B. V. V., Varaprasad, K. C., & Prahlada Rao, K. (2021). Machinability analysis on wire electrical discharge machining of stir casted AA2024/Al₂O₃/BN hybrid composite for aerospace applications. *Materials and Manufacturing Processes*, 36(6), 730–743. https://doi.org/10.1080/10426914.2020.1854466.
- Manikandan, N., Balasubramanian, K., Palanisamy, D., Gopal, P. M., Arulkirubakaran, D., & Binoj, J. S. (2019). Machinability analysis and ANFIS modelling on advanced machining of hybrid metal matrix composites for aerospace applications. *Materials and Manufacturing Processes*, 34(16), 1866–1881. https://doi.org/10.1080/10426914.2019. 1689264.

- Karthik, S., Prakash, K. S., Gopal, P. M., & Jothi, S. (2019). Influence of materials and machining parameters on WEDM of Al/AlCoCrFeNiMo 0.5 MMC. *Materials and Manufacturing Processes*, 34(7), 759–768. https://doi.org/10.1080/10426914.2019.159 4250.
- Feucht, F., Ketelaer, J., Wolff, A., Mori, M., & Fujishima, M. (2014). Latest machining technologies of hard-to-cut materials by ultrasonic machine tool. *Procedia CIRP*, 14, 148–152. https://doi.org/10.1016/j.procir.2014.03.040.
- Li, L., Wu, M., Liu, X., Cheng, Y., & Yu, Y. (2017). The prediction of surface roughness of PCBN turning GH4169 based on adaptive genetic algorithm. *Integrated Ferroelectrics*, *180*(1), 118–132. https://doi.org/10.1080/10584587.2017.1338881.
- Dou, J., Cui, J., Fang, X., Dong, X., Ullah, N., & Xu, M. (2020). Theoretical and experimental study on machining rectangular microgroove of diamond by femtosecond laser. *Integrated Ferroelectrics*, 208(1), 104–116. https://doi.org/10.1080/10584587.2020.172 8722.
- Xu, L., Chen, K., & Du, W. W. (2020). The application and the calculation of deflection angle of the adjustable modular fixture which has space double angle. *Integrated Ferroelectrics*, 210(1), 175–188. https://doi.org/10.1080/10584587.2020.1728858.
- Radovanović, M. (2019). Multi-objective optimization of multi-pass turning AISI 1064 steel. *The International Journal of Advanced Manufacturing Technology*, 100(1–4), 87–100.
- Diaz, O. G., Luna, G. G., Liao, Z., & Axinte, D. (2019). The new challenges of machining ceramic matrix composites (CMCs): Review of surface integrity. *International Journal of Machine Tools and Manufacture*, 139, 24–36. https://doi.org/10.1016/j. ijmachtools.2019.01.003.
- Ma, F., Zhang, H., Gong, Q., & Hon, K. K. B. (2021). A novel energy efficiency grade evaluation approach for machining systems based on inherent energy efficiency. *International Journal of Production Research*, 59(19), 6022–6033. https://doi.org/10.10 80/00207543.2020.1799104.
- 52. Li, D. (2020). Surface structure etching law of self-organizing nanomaterials based on AFM and SEM. *Integrated Ferroelectrics*, 207(1), 49–61. https://doi.org/10.1080/1058 4587.2020.1728664.
- Imad, M., Hopkins, C., Hosseini, A., Yussefian, N. Z., & Kishawy, H. A. (2021). Intelligent machining: A review of trends, achievements and current progress. *International Journal of Computer Integrated Manufacturing*, 1–29. https://doi.org/10. 1080/0951192X.2021.1891573.
- Montazerian, H., Rashidi, A., Milani, A. S., & Hoorfar, M. (2020). Integrated sensors in advanced composites: A critical review. *Critical Reviews in Solid State and Materials Sciences*, 45(3), 187–238. https://doi.org/10.1080/10408436.2019.1588705.

Challenges in Machining of Advanced Materials

Klocke, F., Soo, S. L., Karpuschewski, B., Webster, J. A., Novovic, D., Elfizy, A., Axinte, D. A., & Tönissen, S. (2015). Abrasive machining of advanced aerospace alloys and composites. CIRP Annals – Manufacturing Technology, 64(2), 581–604. https://doi.org/10.1016/j.cirp.2015.05.004. Gates, T. S. H. (2003). Computational materials: Modeling and simulation of nanostructured materials and systems. In 44th American Institute of Aeronautics and Astronautics, Structures, Structural Dynamics and Materials Conference, Langley Research Center, Hampton, Virginia, US.(p. 1534).

Galantucci, L. M., Lavecchia, F., & Percoco, G. (2008). Study of compression properties of topologically optimized FDM made structured parts. CIRP Annals, 57, 243–246.

Abramovitch, H., Burgard, M., Edery-Azulay, L., Evans, K. E., Hoffmeister, M., Miller, W., Scarpa, F., Smith, C. W., & Tee, K. F. (2010). Smart tetrachiral and hexachiral honeycomb: Sensing and impact detection. Composites Science and Technology, 70, 1072–1079. Miller, W., Smith, C. W., Scarpa, F., & Evans, K. E. (2010). Flatwise buckling optimization of hexachiral and tetrachiral honeycombs. Composites Science and Technology, 70, 1049–1056. Prall, D., & Lakes, R. S. (1997). Properties of a chiral honeycomb with a poisson's ratio of—1. International Journal of Mechanical Sciences, 39, 305–314.

Vartak, D. A., Satyanarayana, B., Munjal, B. S., Vyas, K. B., Bhatt, P., & Lal, A. K. (2020). Potential applications of advanced nano-composite materials for space payload. Australian Journal of Mechanical Engineering, 1–9. https://doi.org/10.1080/14484846.2020.1733176. Wu, M., Li, L., Liu, X., Cheng, Y., & Yu, Y. (2018). Analysis and experimental study of the cutting mechanism in machined nickel-base superalloys GH4169 with PCBN tools under high pressure cooling. Integrated Ferroelectrics, 189(1), 105–120.

https://doi.org/10.1080/10584587.2018.1456175

Mamalis, A. G., Kundrák, J., & Horváth, M. (2002). Wear and tool life of CBN cutting tools. The International Journal of Advanced Manufacturing Technology, 20(7), 475–479.

Kramar, D., Sekuli, M., Jurkovi, Z., & Kopa, J. (2013). The machinability of nickel-based alloys in high pressure jet assisted (HPJA) turning. Metalurgija, 52(4), 512–514.

Schwartz, M. (1992). Handbook of structural ceramics. New York: McGraw Hill.

Tuersley, I. P., Jawaid, A., & Pashby, I. R. (1994). Review: Various methods of machining

advanced ceramic materials. Journal of Materials Processing Technology, 42(4), 377–390.

Ming, W., Jia, H., Zhang, H., Zhang, Z., Liu, K., Du, J., Shen, F., & Zhang, G. (2020). A comprehensive review of electric discharge machining of advanced ceramics. Ceramics Interpretional, 46(14), 21912, 21929, https://doi.org/10.1016/j.acramitt.2020.05.207

International, 46(14), 21813–21838. https://doi.org/10.1016/j.ceramint.2020.05.207.

Samant, A. N., & Dahotre, N. B. (2009). Laser machining of structural ceramics a review. Journal of the European Ceramic Society, 29(6), 969–993.

Ayrikyan, A., Prach, O., Khansur, N. H., & Webber, K. G. (2018). Investigation of residual stress in lead-free BNT-based ceramic/ceramic composites. Acta Materialia, 148, 432–441.

Zhang, B., Zheng, X. L., Tokura, H., & Yoshikawa, M. (2003). Grinding induced damage in ceramics. Journal of Materials Processing Technology, 132(1), 353–364.

Kirchner, H. P. (2010). Damage penetration at elongated machining grooves in hotcpressed Si3N4. Journal of the American Ceramic Society, 67(2), 127–132.

Hua, W., Zhang, W., Li, J., Cai, K., & Jiang, Y. (2020). Feasibility of abandoned ammunition materials cold cutting experimental study and numerical prediction. Integrated Ferroelectrics, 207(1), 1–11. https://doi.org/10.1080/10584587.2020.1728660.

Goindi, G. S., & Sarkar, P. (2017). Dry machining: A step towards sustainable machining – Challenges and future directions. Journal of Cleaner Production, 165, 1557–1571. https://doi.org/10.1016/j.jclepro.2017.07.235.

Klocke, F., Gierlings, S., Brockmann, M., & Veselovac, D. (2011). Influence of temperature on surface integrity for typical machining processes in aero engine manufacture. Procedia Engineering, 19, 203–208.

Thakur, A., & Gangopadhyay, S. (2016). Dry machining of nickel-based super alloy as a sustainable alternative using TiN/TiAIN coated tool. Journal of Cleaner Production, 129, 256–268.

Spriet, P. (2014). CMC applications to gas turbines. In Ceramic matrix composites: Materials, modeling and technology. Bansal, N.P., & Lamon, J., Hoboken, NJ: John Wiley & Sons, Inc. (pp. 591–608).

Yuan, S., Li, Z., Zhang, C., & Guskov, A. (2017). Research into the transition of material removal mechanism for C/SiC in rotary ultrasonic face machining. The International Journal of Advanced Manufacturing Technology, 95(5–8), 1751–1761.

Ding, K., Fu, Y., Su, H., Chen, Y., Yu, X., & Ding, G. (2014). Experimental studies on drilling tool load and machining quality of C/SiC composites in rotary ultrasonic machining. Journal of Materials Processing Technology, 214(12), 2900–2907.

Diaz, O. G., Axinte, D., Butler-Smith, P., & Novovic, D. (2018). On understanding the microstructure of SiC/SiC ceramic matrix composites (CMCs) after a material removal process. Materials Science and Engineering, 713(26), 1–11.

Hashish, M., Kotchon, A., & Ramulu, M. (2015). Status of AWJ machining of CMCS and hard materials. In International Technical Conference on Diamond, Cubic Boron Nitride and Their Applications. Indianapolis, IN (pp. 1–14).

Tuersley, I., & Hoult, T. (1998). The processing of SiC – SiC ceramic matrix composites using a pulsed Nd-YAG laser: Part II the effect of process variables. Journal of Materials Science, 3, 963–967.

Li, W., Zhang, R., Liu, Y., Wang, C., Wang, J., Yang, X., & Cheng, L. (2016). Effect of different parameters on machining of SiC/SiC composites via pico-second laser. Applied Surface Science, 364, 378–387.

Wei, C., Zhao, L., Hu, D., & Ni, J. (2013). Electrical discharge machining of ceramic matrix composites with ceramic fiber reinforcements. The International Journal of Advanced Manufacturing Technology, 64(1–4), 187–194.

Huiping, Z., Hongxia, Z., & Yinan, L. (2014). Surface roughness and residual stresses of high speed turning 300 M ultrahigh strength steel. Advances in Mechanical Engineering, 2014, 1–7. https://doi.org/10.1155/2014/859207.

Zhang, H. P., Zhang, Z., Zheng, Z. Y., & Liu, E. (2020). Tool wear in high-speed turning ultrahigh strength steel under dry and CMQL conditions. Integrated Ferroelectrics, 206(1), 122–131. https://doi.org/10.1080/10584587.2020.1728633.

Kappmeyer, G., Hubig, C., Hardy, M., Witty, M., & Busch, M. (2012). Modern machining of advanced aerospace alloys-Enabler for quality and performance. Procedia CIRP, 1(1), 28–43. https://doi.org/10.1016/j.procir.2012.04.005.

Abele, E., Lieder, M., Hölscher, R., & Hubig, C. (2012). Qualitative description of tool wear characteristic. *Wt-Online*.

Wu, M., Wu, S., Chu, W., Liu, K., & Cheng, Y. (2020). Experimental study on tool wear in cutting superalloy under high-pressure cooling. Integrated Ferroelectrics, 207(1), 208–219. https://doi.org/10.1080/10584587.2020.1728681.

Alabdullah, M., Polishetty, A., Nomani, J., & Littlefair, G. (2019). An investigation on machinability assessment of Al-6XN and AISI 316 alloys: An assessment study of machining. Machining Science and Technology, 23(2), 171–217. https://doi.org/10.1080/10910344.2018.1486415.

Lin, H. C., Lin, K. M., & Chen, Y. C. (2000). A study on the machining characteristics of TiNi shape memory alloys. Journal of Materials Processing Technology, 105, 327–332.

Kong, M. C., Axinte, D., & Voice, W. (2011). Challenges in using waterjet machining of NiTi shape memory alloys: An analysis of controlled depth milling. Journal of Materials Processing Technology, 211(6), 959–971.

Frotscher, M., Kahleyss, F., Simon, T., Biermann, D., & Eggeler, G. (2011). Achieving small structures in thin NiTi sheets for medical applications with water jet and micro machining: A comparison. Journal of Materials Engineering and Performance, 15, 776–782.

Manjaiah, M., Narendranath, S., Basavarajappa, S., & Gaitonde, V. N. (2014). Wire electric discharge machining characteristics of titanium nickel shape memory alloy. Transactions of Nonferrous Metals Society of China (English Edition), 24(10), 3201–3209.

Ahmed, N., Ishfaq, K., Moiduddin, K., Ali, R., & Al-Shammary, N. (2019). Machinability of titanium alloy through electric discharge machining. Materials and Manufacturing Processes, 34(1), 93–102. https://doi.org/10.1080/10426914.2018.1532092.

Mahesh, K., Philip, J. T., Joshi, S. N., & Kuriachen, B. (2021). Machinability of Inconel 718: A critical review on the impact of cutting temperatures. Materials and Manufacturing Processes, 36(7), 753–791. https://doi.org/10.1080/10426914.2020.1843671.

Naidu, B. V. V., Varaprasad, K. C., & Prahlada Rao, K. (2021). Machinability analysis on wire electrical discharge machining of stir casted AA2024/Al2O3/BN hybrid composite for aerospace applications. Materials and Manufacturing Processes, 36(6), 730–743.

https://doi.org/10.1080/10426914.2020.1854466.

Manikandan, N., Balasubramanian, K., Palanisamy, D., Gopal, P. M., Arulkirubakaran, D., & Binoj, J. S. (2019). Machinability analysis and ANFIS modelling on advanced machining of hybrid metal matrix composites for aerospace applications. Materials and Manufacturing Processes, 34(16), 1866–1881. https://doi.org/10.1080/10426914.2019.1689264.

Karthik, S., Prakash, K. S., Gopal, P. M., & Jothi, S. (2019). Influence of materials and machining parameters on WEDM of Al/AlCoCrFeNiMo 0.5 MMC. Materials and Manufacturing Processes, 34(7), 759–768. https://doi.org/10.1080/10426914.2019.1594250.

Feucht, F., Ketelaer, J., Wolff, A., Mori, M., & Fujishima, M. (2014). Latest machining technologies of hard-to-cut materials by ultrasonic machine tool. Procedia CIRP, 14, 148–152. https://doi.org/10.1016/j.procir.2014.03.040.

Li, L., Wu, M., Liu, X., Cheng, Y., & Yu, Y. (2017). The prediction of surface roughness of PCBN turning GH4169 based on adaptive genetic algorithm. Integrated Ferroelectrics, 180(1), 118–132. https://doi.org/10.1080/10584587.2017.1338881.

Dou, J., Cui, J., Fang, X., Dong, X., Ullah, N., & Xu, M. (2020). Theoretical and experimental study on machining rectangular microgroove of diamond by femtosecond laser. Integrated Ferroelectrics, 208(1), 104–116. https://doi.org/10.1080/10584587.2020.1728722.

Xu, L., Chen, K., & Du, W. W. (2020). The application and the calculation of deflection angle of the adjustable modular fixture which has space double angle. Integrated Ferroelectrics, 210(1), 175–188. https://doi.org/10.1080/10584587.2020.1728858.

Radovanović, M. (2019). Multi-objective optimization of multi-pass turning AISI 1064 steel. The International Journal of Advanced Manufacturing Technology, 100(1–4), 87–100.

Diaz, O. G., Luna, G. G., Liao, Z., & Axinte, D. (2019). The new challenges of machining ceramic matrix composites (CMCs): Review of surface integrity. International Journal of Machine Tools and Manufacture, 139, 24–36. https://doi.org/10.1016/j.ijmachtools.2019.01.003. Ma, F., Zhang, H., Gong, Q., & Hon, K. K. B. (2021). A novel energy efficiency grade evaluation approach for machining systems based on inherent energy efficiency. International Journal of Production Research, 59(19), 6022–6033. https://doi.org/10.1080/00207543.2020.1799104. Li, D. (2020). Surface structure etching law of self-organizing nanomaterials based on AFM and SEM. Integrated Ferroelectrics, 207(1), 49–61.

https://doi.org/10.1080/10584587.2020.1728664.

Imad, M., Hopkins, C., Hosseini, A., Yussefian, N. Z., & Kishawy, H. A. (2021). Intelligent machining: A review of trends, achievements and current progress. International Journal of Computer Integrated Manufacturing, 1–29. https://doi.org/10.1080/0951192X.2021.1891573. Montazerian, H., Rashidi, A., Milani, A. S., & Hoorfar, M. (2020). Integrated sensors in advanced composites: A critical review. Critical Reviews in Solid State and Materials Sciences, 45(3), 187–238. https://doi.org/10.1080/10408436.2019.1588705.

Machining by Advanced Ceramics Tools

Hrelja, M., Klancnik, S., Irgolic, T., Paulic, M., Balic, J., Brezocnik, M. (2014). Turning parameters optimization using particle swarm optimization. J. Proc. Eng., 69, 670–677. El-Hofy, H.A. (2005). Advanced machining processes. McGrawHill, New York, USA. Baptista, R., Simoes, J.F.A. (2000). Three and five axes milling of sculptured surfaces. J. Mat. Process. Technol., 103, 398–403.

Hamdy, K., Mohamed, M.K., Abouelmagd, G. (2013). New electrode profile for machining of internal cylindrical surfaces by electrochemical drilling. Int. J. Control. Autom. Syst., 2, 2165–8277.

Taylor, F.W. (1907). On the art of cutting metals. Trans. ASME, 28, 31–350.

Kronenberg, M. (1966). Machining science and applications, Pergamon Press, Oxford, UK. Oxley, P.L.B. (1963). Mechanics of metal cutting for a material of variable flow stress. ASME. J. Eng. Ind., 85, 339–345.

Gutiérrez-González, C.F., Suarez, M., Pozhidaev, S., Rivera, S., Peretyagin, P., Solís, W., Díaz, L.A., Fernandez, A., Torrecillas, R. (2016). Effect of TiC addition on the mechanical behaviour of Al2O3-SiC whiskers composites obtained by SPS. J. Eur. Ceram. Soc., 36, 2149–2152.

Zhao, B., Liu, H., Huang, C., Wang, J., Cheng, M. (2017). Fabrication and mechanical properties of Al2O3-SiCw TiCnp ceramic tool material. Ceram. Int., 43, 10224–10230. Mohammadpour, M., Abachi, P., Pourazarang, K. (2012). Effect of cobalt replacement by nickel on functionally graded cemented carbonitrides. Int. J. Refract. Met. Hard Mater, 30, 42–47. Grzesik, W. (2017). Cutting tool materials. Adv. Mach. Processes Metallic Mater., 2 35–63. Kitagawa, T., Kubo, A., Maekawa, K. (1997). Temperature and wear of cutting tools in high-speed machining of Inconel 718 and Ti-6A1-6V-2Sn. Wear, 202, 142–148.

Li, L., He, N., Wang, M., Wang, Z.G. (2002). High-speed cutting of Inconel 718 with coated carbide and ceramic inserts. J. Mater. Process. Technol., 129, 127–130.

Schneider Jr., G. (2009). Cutting Tools Application.

http://www.Toolingandproduction.com/web/home.php

Zhao, J. (2014). 25-the use of ceramic matrix composites for metal cutting applications. Adv. Ceram. Matrix Compos, 2, 623–654.

Jack, D.H. (1986). Ceramic cutting tool materials. Mater. Des., 7, 267–273.

Gonzalez-Julian, J., Schneider, J., Miranzo, P., Osendi, M.I., Belmonte, M. (2011). Enhanced tribological performance of silicon nitride-based materials by adding carbon nanotubes. J. Am. Ceram. Soc., 94, 2542–2548.

Cheng, M., Lia, H., Zhao, B., Huang, C., Yao, P., Wang, B. (2017). Mechanical properties of two types of Al2O3/TiC ceramic cutting tool material at room and elevated temperatures. Ceram. Int., 43, 13869–13874.

Whitney, E.D. (1995). Ceramic cutting tools: Materials, development, and performance, William Andrew Publishing, Norwich, USA.

Stephenson, D.A., Agapiou, J.S. (2016). Metal cutting theory and practice. 3rd edn., CRC Press Taylor & Francis Group, New York, USA.

Grigoriev, S.N., Fedorov, S.V., Hamdy, K. (2019). Materials, properties, manufacturing methods and cutting performance of innovative ceramic cutting tools – A review. Manufacturing Rev. 6, 19, 2330.

Wang, D., Xue, C., Cao, Y., Zhao, J. (2017). Fabrication and cutting performance of an Al2O3/TiC/TiN ceramic cutting tool in turning of an ultra-high-strength steel. Int. J. Adv. Manuf. Technol., 91, 1967–1976.

Smirnov, B.I., Nikolaev, V.I., Orlova, T.S., Shpeizman, V.V., Arellano-Lopez, A.R., Goretta, K.C., Singh, D., Routbort, J.L. (1998). Mechanical properties and microstructure of an Al2O3-SiC-TiC composite. Mat. Sci. Eng. A, 242, 292–295.

Liu, W., Chu, Q., He, R., Huang, M., Wu, H., Jiang, Q., Chen, J., Deng, X., Wu, S. (2018). Preparation and properties of TiAIN coatings on silicon nitride ceramic cutting tools. Ceram. Int., 44, 2209–2215.

Švec, P., Brusilová, A., Kozánková, J. (2008). Effect of microstructure and mechanical properties on wear resistance of silicon nitride ceramics. Mater. Eng., 16, 34–40.

Cheng, Y., Zhang, Y., Wan, T., Yin, Z., Wang, J. (2017). Mechanical properties and toughening mechanisms of graphene platelets reinforced Al2O3/TiC composite ceramic tool materials by microwave sintering. Mat. Sci. Eng. A,680, 190–196.

Gandotra, S., Singh, J., Gill, S.S. (2011). Investigation of wear behaviour on coated and noncoated carbide inserts subjected to low temperature treatment. J. Metall. Eng., 1, 1–16.

Kumar, S., Khedkar, N.K., Jagtap, B., Singh, T.P. (2017). The effects of cryogenic treatment on cutting tools. IOP Conf. Series: Mater. Sci. Eng., 225, 1–9.

Dusza, J., Morgiel, J., Duszová, A., Kvetková, L., Nosko, M., Kun, P., Balázsi, C. (2012). Microstructure and fracture toughness of Si3N4+ graphene platelet composites. J. Eur. Ceram. Soc., 32, 3389–3397.

Chung-Cheng, T., Hong, H. (2002). Comparison of the tool life of tungsten carbide coated by multi-layer TiCN and TiALCN for end mills using Taguchi method. J. Mater. Process. Technol., 123, 1–4.

Bouzakis, K.-D., Michailidis, N., Skordaris, G., Bouzakis, E., Biermann, D., Saoubi, R.M. (2012). cutting with coated tools: Coating technologies, characterization methods and performance optimization. CIRP Ann. Manuf. Technol., 61, 703–723.

Vereschaka, A.S., Grigoriev, S.N., Tabakov, V.P., Sotova, E.S., Vereschaka, A.A., Kulikov, M.Y. (2014). Improving the efficiency of the cutting tool made of ceramic when machining hardened steel by applying nano-dispersed multi-layered coatings. Key Eng. Mater., 581, 68–73.

Gurdial Blugan, G., Strehler, C., Vetterli, M., Ehrle, B., Duttlinger, R., Blösch, P., Kuebler, J. (2017). Performance of lightweight coated oxide ceramic composites for industrial high speed wood cutting tools: A step closer to market. Ceram. Int., 43, 8735–8742.

Long, Y., Zeng, J., Wu, S. (2014). Cutting performance and wear mechanism of Ti-Al-N/Al-Cr-O coated silicon nitride ceramic cutting inserts. Ceram. Int., 40, 9615–9620.

Gajrani, K.K., Sankar, M.R. (2017). State of the art on micro to nano-textured cutting tools. Mater. Today Proc., 4, 3776–3785.

Feng, Y., Zhang, J., Wang, L., Zhang, W., Tian, Y., Kong, X. (2017). Fabrication techniques and cutting performance of micro textured self-lubricating ceramic cutting tools by in-situ forming of Al2O3-TiC. Int. J. Refract. Met. Hard Mater., 68, 121–129.

Sugihara, T., Enomoto, T. (2012). Improving anti-adhesion in aluminum alloy cutting by micro strip texture. Prec. Eng., 36, 229–237.

Lian, Y., Deng, J., Yan, G., Cheng, H., Zhao, J. (2013). Preparation of tungsten disulfide (WS2) soft-coated nano-textured self-lubricating tool and its cutting performance. Int. J. Adv. Manuf. Technol., 68, 2033–2042.

Xing, Y., Deng, J., Zhao, J., Zhang, G., Zhang, K. (2014). Cutting performance and wear mechanism of nanoscale and microscale textured Al2O3/TiC ceramic tools in dry cutting of hardened steel. Int. J. Refract. Met. Hard Mater., 43, 46–58.

Rathod, P., Aravindan, S., Venkateswara, R.P. (2016). Performance evaluation of novel microtextured tools in improving the machinability of aluminum alloy (Al 6063). Proc. Technol., 23, 296–303.

Wu, G., Xu, C., Xiao, G., Yi, M., Chen, Z., Xu, L. (2016). Self-lubricating ceramic cutting tool material with the addition of nickel coated CaF2 solid lubricant powders. Int. J. Refract. Met. Hard Mater., 56, 51–58.

Cui, X., Guo, J., Zheng, J. (2016). Optimization of geometry parameters for ceramic cutting tools in intermittent turning of hardened steel. Mater. Des., 92, 424–437.

Cinert, J. (2018). Study of mechanisms of the spark plasma sintering technique, PHD Thesis, Czech Technical University, Prague, Czech Republic.

Matizamhuka, W.R. (2016). Spark plasma sintering (SPS) an advanced sintering technique for structural nanocomposite materials. J. S. Afr. Inst. Min. Metall., 116, 1171–1180.

Saheb, N., Iqbal, Z., Khalil, A., Hakeem, A.S., Al-Aqeeli, N., Laoui, T., Al-Qutub, A., Kirchner, R. (2012). Spark plasma sintering of metals and metal matrix nanocomposites: A review. J. Nanomater. DOI:10.1155/2012/983470.

Munir, Z.A., Anselmi-Tamburini, U., Ohyanagi, M. (2006). The effect of electric field and pressure on the synthesis and consolidation of materials: A review of the spark plasma sintering method. J. Mater. Sci., 41, 763–777.

Borrell, A., Salvador, M.D. (2018). Chapter 1: Advanced ceramic materials sintered by microwave technology. In Sintering technology method and application, Intech Open., 3–24, DOI: 10.5772/intechopen.78831

Gołombek, K., Domrzalski, L.A. (2007). Hard and wear resistance coatings for cutting tools. J. Achiev. Mater. Manuf. Eng., 24, 107–110.

Gevorkyan, E., Lavrynenko, S., Rucki, M., Siemiatkowski, Z., Kislitsa, M. (2017). Ceramic cutting tools out of nanostructured refractory compounds. Int. J. Refract. Met. Hard Mater., 68, 142–144.

Li, M., Huang, C., Zhao, B., Liu, H., Wang, J., Liu, Z. (2017). Mechanical properties and microstructure of Al2O3-TiB2- TiSi2 ceramic tool material. Ceram. Int., 43, 14192–14199. Zou, B., Huang, C., Song, J., Liu, Z., Liu, L., Zhao, Y. (2012). Effects of sintering processes on mechanical properties and microstructure of TiB2-TiC + 8 wt% nano-Ni composite ceramic cutting tool material. Mater. Sci. Eng. A 540, 235–244.

Yin, Z., Huang, C., Zou, B., Liu, H., Zhu, H., Wang, J. (2014). Effects of particulate metallic phase on microstructure and mechanical properties of carbide reinforced alumina ceramic tool materials. Ceram. Int., 40, 2809–2817.

Tiwari, D., Basu, B., Biswas, K. (2009), Simulation of thermal and electric field evolution during spark plasma sintering. Ceram. Int., 35, 699–708.

Ramirez, C., Miranzo, P., Belmonte, M., Osendi, M.I., Poza, P., Vega-Diaz, S.M., Terrones, M. (2014). Extraordinary toughening enhancement and flexural strength in Si3N4 composites using graphene sheets. J. Eur. Ceram. Soc., 34, 161–169.

Mishra, S.K., Pathak, L.C. (2009). Self-propagating high-temperature synthesis (SHS) of advanced high-temperature ceramics. Key Eng. Mater. 395, 15–38.

Kim, H., Kawahara, M., Tokita, M. (2000). Specimen temperature and sinterability of Ni powder by spark plasma sintering. J. Jpn. Soc. Powder Powder Metal., 47, 887–891.

Levashov, E.A., Mukasyan, A.S., Rogachev, A.S., Shtansky, D.V. (2017). Self-propagating high-temperature synthesis of advanced materials and coatings. Int. Mat. Rev., 62, 203–239. Borovinskaya, I., Gromov, A., Levashov, E., Maksimov, Y., Mukasyan, A., Rogachev, A. (2017). Concise encyclopedia of self-propagating high-temperature synthesis, 1st edn, Elsevier Science, Amsterdam, Netherlands.

Yi, H.C., Moore, J.J. (1990). Self-propagating high-temperature (combusting) synthesis (SHS) of powder-compacted materials. J. Mater. Sci., 25, 1159–1168.

Matli, P.R., Abdul-Shakoor, R., Mohamed, A.M.A., Gupta, M. (2016). Microwave rapid sintering of Al-metal matrix composites: A review on the effect of reinforcements, Microstructure. Mech. Prop. Metals, 6, 143–162.

Keblouti, O., Boulanouar, L., Azizi, M.W., Yallese, M.A. (2017). Effects of coating material and cutting parameters on the surface roughness and cutting forces in dry turning of AISI 52100 steel. Struct. Eng. Mech., 61, 519–526.

Ucun, I., Aslantasx, K., Gokcxe, B. Bedir, F. (2014). Effect of tool coating materials on surface roughness in micromachining of Inconel 718 super alloy. Proc. IMechE. Part B: J. Eng. Manuf., 228, 1–13.

Wang, B., Liu, Z. (2016). Cutting performance of solid ceramic end milling tools in machining hardened AISI H13 steel. Int. J. Refract. Met. Hard Mater., 55, 24–32.

Jianxin, D., Wenlong, S., Hui, Z., Pei, Y., Aihua, L. (2011). Friction and wear behaviours of the carbide tools embedded with solid lubricants in sliding wear tests and in dry cutting processes. Wear, 270, 666–674.

Sugihara, T., Tanaka, H., Enomoto, T. (2017). Development of novel CBN cutting tool for high speed machining of Inconel 718 focusing on coolant behaviours. Proc. Manuf., 10, 436–442.

Characterization and Evaluation of Eco-Friendly Cutting Fluids

Ningsih Y R (2017), Utilization of sulphurized palm oil as cutting fluid base oil for broaching process, IOP Conference Series: Earth and Environment Science, 60, 012008.

National Institute for Occupational Safety (1998), Criteria for a recommended standard: Occupational exposure to metalworking fluids, 98–102.

King N, Keranen L, Gunter L, Sutherland J (2001), Wet Versus Dry Turning: A Comparison of Machining Costs, Product Quality, and Aerosol Formation, SAE Technical Paper 2001-01-0343, ISSN:0148-7191.

Feng S C, Hattori M (2000), Cost and Process Information Modeling for Dry Machining, Proceedings of the International Workshop on Environment and Manufacturing,

https://tsapps.nist.gov/publication/get_pdf.cfm?pub_id=821121 (Accessed August 18, 2021. Fratila D, Evaluation of near-dry machining effects on gear milling process efficiency, Journal of Cleaner Production, 17, 839–845.

Nandgaonkar S, Gupta T V K, Joshi S (2016), Effect of water oil mist spray (WOMS) cooling on drilling of Ti6Al4V alloy using ester oil based cutting fluid, Procedia Manufacturing, 6, 71–79. US Dept of labor, Safety and Health Topics (2018), Metalworking fluids – safety and health best

practices manual, Occupational Safety and Health Administration.

Yu Y, Guo Y, Wang L, Tang E (2010), Development of environmentally friendly water-based synthetic metal-cutting fluid, Modern Applied Science, 1, 53–58.

Brown M, Fotheringham J D, Hoyes T J, Mortier R M, Orszulik S T, Randles S J, Stroud P M (1997), Synthetic base fluids, in R M Mortier, M F Fox, and S T Orszulik, eds., Chemistry and technology of lubricants, Dordrecht: Springer Netherlands, 35–74.

Tripathi A, Vinu R (2015), Characterization of thermal stability of synthetic and semi-synthetic engine oils, Lubricants, 3, 54–79.

Debnath S, Reddy M M, Yi Q S (2014), Environmental friendly cutting fluids and cooling techniques in machining: A review, Journal of Cleaner Production, 83, 33–47.

Ramanaa M, Srinivasulub K, Raoc G (2011), Performance evaluation and selection of optimal parameters in turning of Ti-6AI-4V alloy under different cooling conditions, International Journal of Innovative Technology and Creative Engineering, 1, 10–21.

Jayal A D, Balaji A K, Sesek R, Gaul A, Lillquist D R (2007), Machining performance and health effects of cutting fluid application in drilling of A390.0 cast aluminum alloy, Journal of Manufacturing Process, 9, 137–146.

Byers J P (2017), Metalworking fluids, 3rd ed. CRC Press.

Pusavec F, Krajnik P, Kopac J (2010), Transitioning to sustainable production – Part I: Application on machining technologies, Journal of Cleaner Production, 18, 174–184.

Rubio E M, Agustina B, Marín M, Bericua A (2015), Cooling systems based on cold compressed air: A review of the applications in machining processes, Procedia Engineering, 132, 413–418. Jiang C-C, Cao Y-K, Xiao G-Y, Zhu R-F, Lu Y-P (2017), A review on the application of inorganic nanoparticles in chemical surface coatings on metallic substrates, RSC Advances, 7, 7531–7539.

Xie H, Jiang B, He J, Xia X, Pan F (2016), Lubrication performance of MoS2 and SiO2 nanoparticles as lubricant additives in magnesium alloy-steel contacts, Tribology International, 93, 63–70.

Naves V T G, Da Silva M B, Da Silva F J, Evaluation of the effect of application of cutting fluid at high pressure on tool wear during turning operation of AISI 316 austenitic stainless steel, Wear, 302, 1201–1208.

Amrita M, Srikant R R, Sitaramaraju A V (2014), Evaluation of cutting fluid with nanoinclusions, Journal of Nanotechnology Engineering Medicine, 4, 031007.

Singh R K, Sharma A K, Dixit A R, Mandal A, Tiwari A K (2017), Experimental investigation of thermal conductivity and specific heat of nanoparticles mixed cutting fluids, Materials Today Proceedings, 4, 8587–8596.

Donnet C., Erdemir A (2004), Solid lubricant coatings: Recent developments and future trends, Tribology Letters, 17, 389–397.

Sharma A K, Tiwari A K, Dixit A R (2016), Characterization of TiO2, Al2O3 and SiO2 nanoparticle based cutting fluids, Materials Today Proceedings, 3, 1890–1898.

Najiha M S, Rahman M M, and Kadirgama K, Performance of water-based TiO2 nanofluid during the minimum quantity lubrication machining of aluminium alloy, AA6061-T6, Journal of Cleaner Production, 135, 1623–1636.

Tosum N, Rostam S, Rasul S (2016), Use of Nano Cutting Fluid in Machining, Fourth International Conference on Advanced Mechanical Automation Engineering – MAE 2016, Rome, Italy, 17–21.

Xie H, Jiang B, Hu X, Peng C, Guo H, Pan F (2017), Synergistic effect of MoS2 and SiO2 nanoparticles as lubricant additives for magnesium alloy-steel contacts, Nanomaterials 7, 1–16. Aureli F, D'Amato M, De Berardis B, Raggi A, Turco A C, Cubadda F (2012), Investigating agglomeration and dissolution of silica nanoparticles in aqueous suspensions by dynamic reaction cell inductively coupled plasma-mass spectrometry in time resolved mode, Journal of Analytical Atomic Spectrometry, 27, 1540.

Gilbert Y, Veillette M, Duchaine C (2010), Metalworking fluids biodiversity characterization, Journal of Applied Microbiology, 108, 437–449.

Nune M M R, Kiran C P (2019), Development, characterization, and evaluation of novel ecofriendly metal working fluid, Measurement, 137, 401–416.

Singh R K, Sharma A K, Dixit A R, Mandal A, Tiwari A K (2017), Experimental investigation of thermal conductivity and specific heat of nanoparticles mixed cutting fluids, Materials Today: Proceedings, 4, 8587–8596.

Nune M M R, Kiran C P (2020), Performance evaluation of novel developed biodegradable metal working fluid during turning of AISI 420 material, Journal of Brazilian Society of Mechanical Science and Engineering, 42, 1–16.

Adams M C, Gannon J E, Bennett E O (1979), BOD and COD studies of synthetic and semisynthetic cutting fluids, Water Air Soil Pollution, 11, 105–113.

Gajrani K K, Ram D, Ravi Sankar M (2017), Biodegradation and hard machining performance comparison of eco-friendly cutting fluid and mineral oil using flood cooling and minimum quantity cutting fluid techniques, Journal of Cleaner Production, 165, 1420–1435.

Huang J, Wang X, Long Q, Wen X, Zhou Y, Li L (2009), Influence of pH on the stability characteristics of nanofluids, 2009 Symposium on Photonics and Optoelectronics, Wuhan,

China, 1–4.

Lin J, Wang L, Chen G (2011), Modification of graphene platelets and their tribological properties as a lubricant additive, Tribology Letters, 41, 209–215.

Husnawan M, Saifullah M G, Masjuki H H (2007), Development of friction force model for mineral oil base stock containing palmolein and anti-wear additive, Tribology International, 40, 74–81.

Ing T C, Mohammed Rafiq A K, Azli Y, Syahrullail S (2012), The effect of temperature on the tribological behavior of RBD palm stearin, Tribology Transactions, 55, 539–548.

Yunus R, Fakhru'l-Razi A, Ooi T L, Iyuke S E, Perez J M (2004), Lubrication properties of trimethylolpropane esters based on palm oil and palm kernel oils, European Journal of Lipid Science Technology, 106, 52–60.

Sharif M Z, Azmi W H, Redhwan A A M, Zawawi N N M, Mamat R (2017), Improvement of nanofluid stability using 4-Step UV-vis spectral absorbency analysis, Journal of Mechanical Engineering, 4, 233–247.

Nandanwar R, Singh P, Haque F Z, Lee S H (2015), Synthesis and characterization of SiO2 nanoparticles by sol-gel process and its degradation of methylene blue, Chemical Science Journal, 5, 1–10.

Angayarkanni S A, Philip J (2013), Role of adsorbing moieties on thermal conductivity and associated properties of nanofluids, Journal of Physical Chemistry C, 117, 9009–9019.

Vidal F A C, Ávila A F (2014), Tribological investigation of nanographite platelets as additive in anti-wear lubricant: A top-down approach, Journal of Tribology, 136, 031603.

Kaufman Y, Chen, S Y, Mishra H, Schrader A M, Lee D W, Das S, Donaldson S H, Israelachvili J N (2017), Simple-to-apply wetting model to predict thermodynamically stable and metastable contact angles on textured/rough/patterned surfaces, Journal of Physical Chemistry C, 121, 5642–5656.

Eshraghi E, Kazemzadeh Y, Qahramanpour M, Kazemi A (2017), Investigating effect of SiO2 nanoparticle and sodium-dodecyl-sulfate surfactant on surface properties: Wettability alteration and IFT reduction, Journal of Petroleum Environment Biotechnology, 8, 6–10.

Weston J S, Jentoft R E, Grady B P, Resasco D E, Harwell J H (2015), Silica nanoparticle wettability: Characterization and effects on the emulsion properties, Industrial Engineering Chemistry Research, 54, 4274–4284.

Pedišić L, Šarić M, Bielen S (2003), Application possibilities of new AW/EP additive types in watermiscible metalworking fluids, Industrial Lubrication Tribology, 55, 23–31.

Advances in Textured Cutting Tools for Machining

Mills, B. 1996. "Recent developments in cutting tool materials". Journal of Materials Processing Technology, 56(1), 16–23. doi: 10.1016/0924-0136(95)01816-6

Shokrani, A., Dhokia, V., and Newman, S. T. 2012. "Environmentally conscious machining of difficult-to-machine materials with regard to cutting fluids". International Journal of Machine Tools and Manufacture, 57, 83–101. doi: 10.1016/j.ijmachtools.2012.02.002

Jedrzejewski, J., and Kwasny, W. 2017. "Development of machine tools design and operational properties". The International Journal of Advanced Manufacturing Technology, 93(1), 1051–1068. doi: 10.1007/s00170-017-0560-2

Deshpande, S., and Deshpande, Y. 2019. "A review on cooling systems used in machining processes". Materials Today: Proceedings, 18, 5019–5031. doi: 10.1016/j.matpr.2019.07.496 Gajrani, K. K., and Sankar, M. R. 2017. "State of the art on micro to nano textured cutting tools". Materials Today: Proceedings, 4(2, Part A), 3776–3785. doi: 10.1016/j.matpr.2017.02.274 Astakhov, V. P. 2011. "Machining of Hard Materials – Definitions and Industrial Applications". In Machining of Hard Materials. ed. J. P. Davim. 1–32. London: Springer London.

Kligerman, Y., Etsion, I., and Shinkarenko, A. 2005. "Improving tribological performance of piston rings by partial surface texturing". Journal of Tribology, 127(3), 632–638. doi: 10.1115/1.1866171

Tala-Ighil, N., Maspeyrot, P., Fillon, M., and Bounif, A. 2007. "Effects of surface texture on journal-bearing characteristics under steady-state operating conditions". Proceedings of the Institution of Mechanical Engineers, Part J: Journal of Engineering Tribology, 221(6), 623–633.

doi: 10.1243/13506501JET287

Pou, P., Riveiro, A., del Val, J., Comesaña, R., Penide, J., Arias-González, F., and Pou, J. 2017. "Laser surface texturing of Titanium for bioengineering applications". Procedia Manufacturing, 13, 694–701. doi: 10.1016/j.promfg.2017.09.102

Ribeiro, F. S. F., Lopes, J. C., Bianchi, E. C., and de Angelo Sanchez, L. E. 2020. "Applications of texturization techniques on cutting tools surfaces—A survey". The International Journal of Advanced Manufacturing Technology, 109(3), 1117–1135. doi: 10.1007/s00170-020-05669-0 Sawant, M. S., Jain, N. K., and Palani, I. A. 2018. Investigations on Additive Manufacturing of Metallic Materials by Micro-Plasma Transferred Arc Powder Deposition Process. Doctoral dissertation, Department of Mechanical Engineering, IIT Indore.

Jhavar, S., Jain, N. K., and Paul, C. P. 2014. "Development of micro-plasma transferred arc (µ-PTA) wire deposition process for additive layer manufacturing applications". Journal of Materials Processing Technology, 214(5), 1102–1110. doi: 10.1016/j.jmatprotec.2013.12.016 Pratap, A., Patra, K., and Dyakonov, A. A. 2019a. "A comprehensive review of micro-grinding: emphasis on toolings, performance analysis, modeling techniques, and future research directions". The International Journal of Advanced Manufacturing Technology, 104(1), 63–102. doi: 10.1007/s00170-019-03831-x

4M Association. 2008. Metals processing: Microgrinding. http://www.4m-association.org/content/Microgrinding.

Pratap, A., Patra, K., and Dyakonov, A. A. 2016. "Manufacturing miniature products by microgrinding: A review". Procedia Engineering, 150, 969–974. doi: 10.1016/j.proeng.2016.07.072 Zhang, B., Liu, X., Brown, C. A., and Bergstrom, T. S. 2002. "Microgrinding of nanostructured material coatings". CIRP Annals, 51(1), 251–254. doi: 10.1016/S0007-8506(07)61510-8 Xie, J., Luo, M.-J., He, J.-L., Liu, X.-R., and Tan, T.-W. 2012. "Micro-grinding of micro-groove array on tool rake surface for dry cutting of titanium alloy". International Journal of Precision Engineering and Manufacturing, 13(10), 1845–1852. doi: 10.1007/s12541-012-0242-9 Dhage, S., Jayal, A. D., and Sarkar, P. 2019. "Effects of surface texture parameters of cutting tools on friction conditions at tool-chip interface during dry machining of AISI 1045 steel". Procedia Manufacturing, 33, 794–801. doi: 10.1016/j.promfg.2019.04.100

Jain, N. K., and Chaubey, S. K. 2017. Review of miniature gear manufacturing. In Comprehensive Materials Finishing, ed. M. S. J. Hashmi, 504–538. Oxford: Elsevier, Oxford. Schubert, A., Zeidler, H., Kühn, R., and Hackert-Oschätzchen, M. 2015. "Micro-electrical discharge machining: a suitable process for machining ceramics". Journal of Ceramics, 2015, 470801. doi: 10.1155/2015/470801

Joshi, S. 2014. "Micro-Electrical Discharge Micromachining". In Introduction to Micromachining, ed. V. K. Jain, 11.1–11.29. New Delhi, India: Narosa Publishing House, ISBN: 978-81-8487-361-0

Lindquist, J. M., Young, R. J., and Jaehnig, M. C. 1993. Recent advances in application of focused ion beam technology. Microelectronic Engineering, 21(1), 179–185. doi: 10.1016/0167-9317(93)90051-6

Jain, V. K. 2014. Advanced Machining Processes (2nd ed.). Allied Publishers Private Limited, New Delhi, India

Nanophysics. 2021. Focused Ion Beam Technology Capabilities and Applications.

https://www.nanophys.kth.se/nanolab/fei-nova200/FejFIB1.pdf

Orso, S. 2005. Application of a FIB system to biological samples. https://elib.uni-stuttgart.de > 02_Orso_Bio_FIB

Jain, N. K., and Jain, V. K. 2001. Modeling of material removal in mechanical type advanced machining processes: a state-of-art review. International Journal of Machine Tools and Manufacture, 41(11), 1573–1635. doi: 10.1016/S0890-6955(01)00010-4

Xu, S., Shimada, K., Mizutani, M., and Kuriyagawa, T. 2014. Fabrication of hybrid micro/nanotextured surfaces using rotary ultrasonic machining with one-point diamond tool. International Journal of Machine Tools and Manufacture, 86, 12–17. doi: 10.1016/j.ijmachtools.2014.06.005 Elias, J. V., Venkatesh N. P., Lawrence, K. D., and Mathew, J. 2021. Tool texturing for microturning applications – an approach using mechanical micro indentation. Materials and Manufacturing Processes, 36(1), 84–93. doi: 10.1080/10426914.2020.1813899

Wu, J., Yu, A., Chen, Q., Wu, M., Sun, L., and Yuan, J. 2020. Tribological properties of bronze surface with dimple textures fabricated by the indentation method. Proceedings of the Institution of Mechanical Engineers, Part J: Journal of Engineering Tribology, 234(10), 1680–1694. doi:

10.1177/1350650120940126

Arslan, A., Masjuki, H. H., Kalam, M. A., Varman, M., Mufti, R. A., Mosarof, M. H., and Quazi, M. M. 2016. "Surface texture manufacturing techniques and tribological effect of surface texturing on cutting tool performance: A review". Critical Reviews in Solid State and Materials Sciences, 41(6), 447–481. doi: 10.1080/10408436.2016.1186597

Kawasegi, N., Sugimori, H., Morimoto, H., Morita, N., and Hori, I. 2009. "Development of cutting tools with microscale and nanoscale textures to improve frictional behavior". Precision Engineering, 33(3), 248–254. doi: 10.1016/j.precisioneng.2008.07.005

Sasi, R., Kanmani Subbu, S., and Palani, I. A. 2017. "Performance of laser surface textured high speed steel cutting tool in machining of Al7075-T6 aerospace alloy". Surface and Coatings Technology, 313, 337–346. doi: 10.1016/j.surfcoat.2017.01.118

Niketh, S., and Samuel, G. L. 2017. "Surface texturing for tribology enhancement and its application on drill tool for the sustainable machining of titanium alloy". Journal of Cleaner Production, 167, 253–270. doi: 10.1016/j.jclepro.2017.08.178

Riveiro, A., Maçon, A. L. B., del Val, J., Comesaña, R., and Pou, J. 2018. "Laser surface texturing of polymers for biomedical applications". Frontiers in Physics, 6(16). doi: 10.3389/fphy.2018.00016

Petare, A. C., Mishra, A., Palani, I. A., and Jain, N. K. 2018. "Study of laser texturing assisted abrasive flow finishing for enhancing surface quality and microgeometry of spur gears". The International Journal of Advanced Manufacturing Technology. doi: 10.1007/s00170-018-2944-3 Sawant, M. S., Jain, N. K., and Palani, I. A. 2018. "Influence of dimple and spot-texturing of HSS cutting tool on machining of Ti-6Al-4V". Journal of Materials Processing Technology, 261, 1–11. doi: 10.1016/j.jmatprotec.2018.05.032

Li, Y., Deng, J., Chai, Y., and Fan, W. 2016. "Surface textures on cemented carbide cutting tools by micro EDM assisted with high-frequency vibration". The International Journal of Advanced Manufacturing Technology, 82(9), 2157–2165. doi: 10.1007/s00170-015-7544-x Singh, B., Sasi, R., Kanmani Subbu, S., and Muralidharan, B. 2019. Electric discharge texturing of HSS cutting tool and its performance in dry machining of aerospace alloy. Journal of the Brazilian Society of Mechanical Sciences and Engineering, 41(3), 152. doi: 10.1007/s40430-019-1654-6

Pratap, A., Patra, K., and Dyakonov, A. A. 2019b. On-machine texturing of PCD micro-tools for dry micro-slot grinding of BK7 glass. Precision Engineering, 55, 491–502. doi: 10.1016/j.precisioneng.2018.11.004

Pratap, A., and Patra, K. 2020. Combined effects of tool surface texturing, cutting parameters and minimum quantity lubrication (MQL) pressure on micro-grinding of BK7 glass. Journal of Manufacturing Processes, 54, 374–392. doi: 10.1016/j.jmapro.2020.03.024

Kawasegi, N., Ozaki, K., Morita, N., Nishimura, K., and Yamaguchi, M. 2017. Development and machining performance of a textured diamond cutting tool fabricated with a focused ion beam and heat treatment. Precision Engineering, 47, 311–320. doi:

10.1016/j.precisioneng.2016.09.005

Elias, J. V., Venkatesh, N. P., Lawrence, K. D., and Mathew, J. 2021. Tool texturing for microturning applications – an approach using mechanical micro indentation. Materials and Manufacturing Processes, 36(1), 84–93. doi: 10.1080/10426914.2020.1813899

Wu, J., Yu, A., Chen, Q., Wu, M., Sun, L., and Yuan, J. 2020. Tribological properties of bronze surface with dimple textures fabricated by the indentation method. Proceedings of the Institution of Mechanical Engineers, Part J: Journal of Engineering Tribology, 234(10), 1680–1694. doi: 10.1177/1350650120940126

Zhang, K., Deng, J., Xing, Y., Li, S., and Gao, H. 2015. "Effect of microscale texture on cutting performance of WC/Co-based TiAIN coated tools under different lubrication conditions". Applied Surface Science, 326, 107–118. doi: 10.1016/j.apsusc.2014.11.059

Sun, J., Zhou, Y., Deng, J., and Zhao, J. 2016. "Effect of hybrid texture combining micro-pits and micro-grooves on cutting performance of WC/Co-based tools". The International Journal of Advanced Manufacturing Technology, 86(9), 3383–3394. doi: 10.1007/s00170-016-8452-4 Duan, R., Deng, J., Ai, X., Liu, Y., and Chen, H. 2017. "Experimental assessment of derivative cutting of micro-textured tools in dry cutting of medium carbon steels". The International Journal of Advanced Manufacturing Technology, 92(9), 3531–3540. doi: 10.1007/s00170-017-0360-8 Sugihara, T., and Enomoto, T. 2017. "Performance of cutting tools with dimple textured surfaces: A comparative study of different texture patterns". Precision Engineering, 49, 52–60. doi: 10.1016/j.precisioneng.2017.01.009 Ye, D., Lijun, Y., Bai, C., Xiaoli, W., Yang, W., and Hui, X. 2018. "Investigations on femtosecond laser-modified microgroove-textured cemented carbide YT15 turning tool with promotion in cutting performance". The International Journal of Advanced Manufacturing Technology, 96(9), 4367–4379. doi: 10.1007/s00170-018-1906-0

Hao, X., Chen, X., Xiao, S., Li, L., and He, N. 2018. "Cutting performance of carbide tools with hybrid texture". The International Journal of Advanced Manufacturing Technology, 97(9), 3547–3556. doi: 10.1007/s00170-018-2188-2

Alagan, N. T., Zeman, P., Hoier, P., Beno, T., and Klement, U. 2019. "Investigation of microtextured cutting tools used for face turning of alloy 718 with high-pressure cooling". Journal of Manufacturing Processes, 37, 606–616. doi: 10.1016/j.jmapro.2018.12.023

Vignesh, G., Barik, D., Aravind, S., Ragupathi, P., and Arun, M. 2021. "An experimental study on machining of AISI H – 13 steel using dimple-textured and non-textured tungsten carbide cutting tools". IOP Conference Series: Materials Science and Engineering, 1017, 012021. doi: 10.1088/1757-899x/1017/1/012021

Sugihara, T., Kobayashi, R., and Enomoto, T. 2021. "Direct observations of tribological behavior in cutting with textured cutting tools". International Journal of Machine Tools and Manufacture, 168, 103726. doi: 10.1016/j.ijmachtools.2021.103726

Wang, X., Popov, V. L., Yu, Z., Li, Y., Xu, J. and Yu, H. 2021. Study on cutting performance of SiCp/Al composite using textured tool. Research Square. doi: 10.21203/rs.3.rs-302368/v1 Devaraj, S., Malkapuram, R., and Singaravel, B. 2021. Performance analysis of micro textured cutting insert design parameters on machining of Al-MMC in turning process. International Journal of Lightweight Materials and Manufacture, 4(2), 210–217. doi: 10.1016/i.jilmm.2020.11.003

Lu, Y., Deng, J., Sun, Q., Ge, D., Wu, J., and Zhang, Z. 2021. Effect of micro textures on the cutting performance of circular saw blade. The International Journal of Advanced Manufacturing Technology, 115(9), 2889–2903. doi: 10.1007/s00170-021-07348-0

Wang, Q., Yang, Y., Yao, P., Zhang, Z., Yu, S., Zhu, H., and Huang, C. 2021. Friction and cutting characteristics of micro-textured diamond tools fabricated with femtosecond laser. Tribology International, 154, 106720. doi: 10.1016/j.triboint.2020.106720

Advances in MQL Machining

Meisenzahl, R.R. and Mokyr, J., 2012. 9. The Rate and Direction of Invention in the British Industrial Revolution (pp. 443–482). University of Chicago Press.

Selvam, M.D. and Senthil, P., 2016. Investigation on the effect of turning operation on surface roughness of hardened C45 carbon steel. Australian Journal of Mechanical Engineering, 14(2), pp. 131–137.

Singh, G., Aggarwal, V. and Singh, S., 2020. Critical review on ecological, economical and technological aspects of minimum quantity lubrication towards sustainable machining. Journal of Cleaner Production, 271(22), p. 122185.

Li, L., 2018. China's manufacturing locus in 2025: With a comparison of "Made-in-China 2025" and "Industry 4.0". Technological Forecasting and Social Change, 135, pp. 66–74.

Hamran, N.N., Ghani, J.A., Ramli, R. and Haron, C.C., 2020. A review on recent development of minimum quantity lubrication for sustainable machining. Journal of Cleaner Production, 268, p. 122165.

Selvam, M.D., Senthil, P. and Sivaram, N.M., 2017. Parametric optimisation for surface roughness of AISI 4340 steel during turning under near dry machining condition. International Journal of Machining and Machinability of Materials, 19(6), pp. 554–569.

Naseem, M. and Naseem, S., 2020. Environmental Law in India. Kluwer Law International BV. Dennison, M.S., Sivaram, N.M., Barik, D. and Ponnusamy, S., 2019. Turning operation of AISI 4340 steel in flooded, near-dry and dry conditions: a comparative study on tool-work interface temperature. Mechanics and Mechanical Engineering, 23(1), pp. 172–182.

Selvam, M.D. and Sivaram, N.M., 2020. A comparative study on the surface finish achieved during turning operation of AISI 4340 steel in flooded, near-dry and dry conditions. Australian Journal of Mechanical Engineering, 18(3), pp. 457–466.

Abukhshim, N.A., Mativenga, P.T. and Sheikh, M.A., 2006. Heat generation and temperature prediction in metal cutting: A review and implications for high speed machining. International Journal of Machine Tools and Manufacture, 46(7–8), pp. 782–800.

Wickramasinghe, K.C., Sasahara, H., Abd Rahim, E. and Perera, G.I.P., 2020. Green Metalworking Fluids for sustainable machining applications: A review. Journal of Cleaner Production, 257, p. 120552.

Sarhan, A.A., Sayuti, M. and Hamdi, M., 2012. Reduction of power and lubricant oil consumption in milling process using a new SiO2 nanolubrication system. The International Journal of Advanced Manufacturing Technology, 63(5–8), pp. 505–512.

Dennison, M.S., Meji, M.A., Nelson, A.J.R., Balakumar, S. and Prasath, K., 2019. A comparative study on the surface finish achieved during face milling of AISI 1045 steel components using eco-friendly cutting fluids in near dry condition. International Journal of Machining and Machinability of Materials, 21(5–6), pp. 337–356.

Krolczyk, G.M., Maruda, R.W., Krolczyk, J.B., Wojciechowski, S., Mia, M., Nieslony, P. and Budzik, G., 2019. Ecological trends in machining as a key factor in sustainable production–A review. Journal of Cleaner Production, 218, pp. 601–615.

Rajarajan, S., Ramesh Kannan, C. and Dennison, M.S., 2020. A comparative study on the machining characteristics on turning AISI 52100 alloy steel in dry and microlubrication condition. Australian Journal of Mechanical Engineering, pp. 1–12. doi:10.1080/14484846.2019.1711019 Race, A., Zwierzak, I., Secker, J., Walsh, J., Carrell, J., Slatter, T. and Maurotto, A., 2021. Environmentally sustainable cooling strategies in milling of SA516: Effects on surface integrity of dry, flood and MOL machining. Journal of Cleaner Production, 288, p. 125580.

Said, Z., Gupta, M., Hegab, H., Arora, N., Khan, A.M., Jamil, M. and Bellos, E., 2019. A comprehensive review on minimum quantity lubrication (MQL) in machining processes using nano-cutting fluids. The International Journal of Advanced Manufacturing Technology, 105(5), pp. 2057–2086.

Dennison, M.S. and Umar, M.M., 2020. Data-set collected during turning operation of AISI 1045 alloy steel with green cutting fluids in near dry condition. Data in Brief, 32, p. 106215. https://www.epa.gov/sustainability/sustainable-manufacturing (as on 01/09/2021).

Ekinovic, S., Prcanovic, H. and Begovic, E., 2015. Investigation of influence of MQL machining parameters on cutting forces during MQL turning of carbon steel St52-3. Procedia Engineering, 132, pp. 608–614.

Sangwan, K.S., Bhakar, V. and Digalwar, A.K., 2018. Sustainability assessment in manufacturing organizations: Development of assessment models. Benchmarking: An International Journal, 25(3), pp. 994–1027.

Benedicto, E., Carou, D. and Rubio, E.M., 2017. Technical, economic and environmental review of the lubrication/cooling systems used in machining processes. Procedia Engineering, 184, pp. 99–116.

Sharma, A.K., Tiwari, A.K. and Dixit, A.R., 2016. Effects of Minimum Quantity Lubrication (MQL) in machining processes using conventional and nanofluid based cutting fluids: A comprehensive review. Journal of Cleaner Production, 127, pp. 1–18.

Sharif, M.N., Pervaiz, S. and Deiab, I., 2017. Potential of alternative lubrication strategies for metal cutting processes: a review. The International Journal of Advanced Manufacturing Technology, 89(5–8), pp. 2447–2479.

Boswell, B., Islam, M.N., Davies, I.J., Ginting, Y.R. and Ong, A.K., 2017. A review identifying the effectiveness of minimum quantity lubrication (MQL) during conventional machining. The International Journal of Advanced Manufacturing Technology, 92(1), pp. 321–340.

Li, K., Aghazadeh, F., Hatipkarasulu, S. and Ray, T.G., 2003. Health risks from exposure to metal-working fluids in machining and grinding operations. International Journal of Occupational Safety and Ergonomics, 9(1), pp. 75–95.

Zheng, F., Kang, R., Dong, Z., Guo, J., Liu, J. and Zhang, J., 2018. A theoretical and experimental investigation on ultrasonic assisted grinding from the single-grain aspect. International Journal of Mechanical Sciences, 148, pp. 667–675.

Valase, K. and Raut, D.N., 2019. Mediation analysis of multiple constructs in the relationship between manufacturing and technology and environmental constructs in structural equation model for sustainable manufacturing. The International Journal of Advanced Manufacturing Technology, 101(5), pp. 1887–1901.

Osman, K.A., Ünver, H.Ö. and Şeker, U., 2019. Application of minimum quantity lubrication techniques in machining process of titanium alloy for sustainability: A review. The International

Journal of Advanced Manufacturing Technology, 100(9), pp. 2311–2332.

Khanafer, K., Eltaggaz, A., Deiab, I., Agarwal, H. and Abdul-Latif, A., 2020. Toward sustainable micro-drilling of Inconel 718 superalloy using MQL-Nanofluid. The International Journal of Advanced Manufacturing Technology, 107(7), pp. 3459–3469.

Virdi, R.L., Chatha, S.S. and Singh, H., 2020. Machining performance of Inconel-718 alloy under the influence of nanoparticles based minimum quantity lubrication grinding. Journal of Manufacturing Processes, 59, pp. 355–365.

Singh, H., Singh, B. and Virdi, R.L., 2022. Exploration of Effectiveness of Ionic Liquid Adopted as an Additive to the Vegetable Oils. In Sustainable Machining Strategies for Better Performance (pp. 171–183). Springer, Singapore.

Babu, M.N., Anandan, V. and Babu, M.D., 2021. Performance of ionic liquid as a lubricant in turning inconel 825 via minimum quantity lubrication method. Journal of Manufacturing Processes, 64, pp. 793–804.

Shokrani, A. and Betts, J., 2020. A new hybrid minimum quantity lubrication system for machining difficult-to-cut materials. CIRP Annals, 69(1), pp. 73–76.

Shokrani, A., Al-Samarrai, I. and Newman, S.T., 2019. Hybrid cryogenic MQL for improving tool life in machining of Ti-6AI-4V titanium alloy. Journal of Manufacturing Processes, 43, pp. 229–243.

Lai, Z., Wang, C., Zheng, L., Lin, H., Yuan, Y., Yang, J. and Xiong, W., 2020. Effect of cryogenic oils-on-water compared with cryogenic minimum quantity lubrication in finishing turning of 17-4PH stainless steel. Machining Science and Technology, 24(6), pp. 1016–1036.

Yıldırım, Ç.V., 2020. Investigation of hard turning performance of eco-friendly cooling strategies: Cryogenic cooling and nanofluid based MQL. Tribology International, 144, p. 106127.

Sugiantoro, B., 2018. Performance evaluation of using water and bio oil-based nanocutting fluids under minimum quantity lubrication with compressed cold air during milling operations of steel. In IOP Conference Series: Materials Science and Engineering (Vol. 403, p. 012098). IOP Publishing.

Khalil, A.N.M., Azmi, A.I., Murad, M.N., Annuar, A.F. and Ali, M.A.M., 2019. Coupled effects of vortex tube hybrid cooling with minimal quantity reinforced nanoparticle lubricants in turning NiTi alloys. The International Journal of Advanced Manufacturing Technology, 105(7), pp. 3007–3015.

De Bartolomeis, A. and Shokrani, A., 2020. Electrohydrodynamic atomization for minimum quantity lubrication (EHDA-MQL) in end milling Ti6Al4V titanium alloy. Journal of Manufacturing and Materials Processing, 4(3), p. 70.

Nanofluids Application for Cutting Fluids

United Nation, World Commission on Environment and Development "Our Common Future, Chapter 2: Towards Sustainable Development," 1987. Available at www.un-documents.net/ocf-02.html.

Department of Commerce, United States of America, how does Commerce define Sustainable Manufacturing. Available at

http://www.trade.gov/competitiveness/sustainablemanufacturing/how_doc_defines-SM.asp. Machine Tools in China to 2014-Demand and Sales Forecasts, Market Share, Market Size, Market Leader, 2011. Available at www.freedoniagroup.com/Machine-Tools-In-hina.html Last Trends in the European Machine Tool Industry, 2011. Available at www.cecimo.eu/machine-tools/datastatistics/latesttrend.html.

Sutherland J., Gunter K., Allen D., Bauer D., Bras B., Gutowski T., Murphy C., Piwonka T., Sheng P., Thuston D., and Wolff E., A global perspective on the environmental challenges facing the automotive industry: state-of-the-art and directions for the future, Int. J. Veh. Des., 2004, 35(1), 86–110

Quinn M.M., Kriebel D., Geiser K., and Moure-Eraso R., Sustainable production: a proposed strategy for the work environment, Am. J. Ind. Med., 1998, 34, 297–304

Hewstone R.K., Environmental health aspects of lubricant additives, Sci. Total Environ., 1994, 156, 243–254

Greaves I.A., Eisen E.A., Smith T.J., Pothier L.J., Kriebel D., Woskie S.R., Kennedy S.M., Shalat S., and Monson R.R., Respiratory health of automobile workers exposed to metal-working fluid aerosols: respiratory symptoms, Am. J. Ind. Med., 1997, 32, 450–459

Lim C.-H., Yu I.J., Kim H.-Y., Lee S.-B., Kang M.-G., Marshak D.R., and Moon C.-K. Respiratory effect of acute and sub-acute exposure to endotoxincontaminated metal working fluid (MWF) aerosols on Sprague-Dawley rats Arch. Toxicol., 2005, 79, 321–329

Godderis L., Deschuyffeleer T., Roelandt H., Veulemans H., and Moens G., Exposure to metalworking Xuids and respiratory and dermatological complaints in a secondary aluminium plant, Int. Arch. Occup. Environ. Health, 2008, 81, 845–853

Cassin C., and Boothroyd G., Lubricating action of cutting fluids, J. Mech. Eng. Sci., 1965, 1(7), 67–81

D'Errico G.E., Guglielmi E., and Rutelli G., A study of coatings for end mills in high speed metal cutting, J. Mater. Process. Technol., 1999, 92–93, 251–256

Guleryuz C.G., Krzanowski J.E., Veldhuis S.C., and Fox-Rabinovich G.S., Machining performance of TiN coatings incorporating indium as solid lubricant, Surf. Coat. Technol., 2009, 203, 3370–3376

Hanyu H., Murakami Y., Kamiya S., and Saka M., New diamond coating with finely crystallized smooth surface for the tools to achieve fine surface finish of non-ferrous metals, Proc. Front. Surf. Eng., 2003, 169–170, 258–261

Renevier N.M., Oosterling H., Konig U., Dautzenberg H., Kim B.J., Geppert L., Koopmans F.G.M., and Leopold J., Performance and limitations of MoS2/Ti composite coated inserts, Surf. Coat. Technol., 2003, 172(1), 13–23

Dos Santos G.R., Da Costa D.D., Amorim F.L., and Torres R.D., Characterization of DLC thin film and evaluation of machining forces using coated inserts in turning of Al-Si alloys, Surf. Coat. Technol., 2007, 202, 1029–1033

Mativenga P.T., and Hon K.K., A study of cutting forces and surface finish in high-speed machining of AISI H13 tool steel using carbide tools with TiAIN based coatings, Proc. Inst. o Mech. Eng., 2003, 217, 143–151

Dudzenski D., Devillez A., Moufki A., Larrouquere D., Zerrouki V., and Vigneau J., A review of developments toward dry and high speed machining of Inconel 718 alloy, Int. J. Mach. Tools Manuf., 2004, 44, 439–456

Diniz A.E., and Micaroni R., Cutting conditions for finish turning process aiming: the use of dry cutting, Int. J. Mach. Tools Manuf., 2002, 42, 899–904

Galanis N.I., Study of the performance of the turning and drilling of austenitic stainless steels using two coolant techniques. Int. J. Mach. Mach. Mater., 2008, 3, 1–2

Shefelbine W., and Dornfeld D., The effect of dry machining on burr size, 2004. Available at http://repositories.edlib.org/lma/eodef/wendy_1_03.

Outeiro J.C., Pina J.C., M'Saoubi R., Pusavec F., and Jawahir I.S., Analysis of residual stresses induced by dry turning of difficult-to-machine materials, CIRP Ann. Manuf. Technol., 2008, 57, 77–80

Klocke F., and Eisenblatter G., Dry cutting, Ann. CIRP Manuf. Technol., 1997, 46(2), 519–526 Sreejith P.S., and Ngoi B.K.A., Dry machining: machining of the future. J. Mater. Process. Technol., 2000, 101, 287–291

Obikawa T., Kamata Y., Asano Y., Nakayama K., and Otieno A.W., Microliter lubrication machining of Inconel 718. Int. J. Mach. Tools Manuf., 2008, 48(15), 1605–1612

Stoll A., and Furness R., Near-dry machining (MQL) is a key technology for driving paradigm shift in machining operations, Mach. Technol., Soc. Manuf. Eng., Fourth Quarter, 2006, 17(4), 1–22

Astakhov V.P., Ecological Machining: Near-Dry Machining, in P.J. Davim (Ed.): Machining Fundamentals and Recent Advances, Springer, London, 2008, pp. 195–223

Min S., Inasaki I., Fujimura S., Wakabayashi T., and Suda S., Investigation of adsorption behaviour of lubricants in near-dry machining, Proc. Inst. Mech. Eng. Part B: J. Eng. Manuf., 2005, 219, 665–671

Ding Y., and Hong S.Y., Improvement in chip breaking in machining a low carbon steel by cryogenically precooling the workpiece, Trans. ASME, J. Manuf. Sci. Eng., 1997, 120(1), 76–83 Yildiz Y., and Nalbant M., A review of cryogenic cooling in machining processes, Int. J. Mach. Tools Manuf., 2008, 48, 947–964

Dhananchezian M., and Kumar M.P., Experimental investigation of cryogenic cooling by liquid nitrogen in the orthogonal machining of Aluminium 6061-T6 alloy, Int. J. Mach. Mater., 2010, 7(3–4), 274–285

Paul S., Dhar N.R., and Chattopadhyay A.B., Beneficial effects of cryogenic cooling over dry and wet machining on tool wear and surface finish in turning AISI 1060 steel, J. Mater. Process. Technol., 2001, 116, 44–48

Dhar N.R., Paul S., and Chattopadhyay A.B., The influence of cryogenic cooling on tool wear, dimensional accuracy and surface finish in turning AISI 1040 and E4340C steel, Wear, 2002, 249, 932–942

Ezuqwu E.O. Key improvements in the machining of difficult-to-cut aerospace superalloys, Int J Mach Tool Manu, 2005, 45, 1353–1367

Wang Z.Y., and Rajurkar K.P., Cryogenic machining of hard-to-cut materials, Wear, 2000, 239, 168–175

Bermingham M.J., Kirsch J., Sun S., Palanisamy S., and Dargusch M.S., New observations on tool life, cutting forces, and chip morphology in cryogenic machining Ti-6Al-4V, Int. J. Mach. Tools Manuf., 2011, 51, 500–511

Hong S.Y., and Ding Y., Cooling approaches and cutting temperatures in cryogenic machining of Ti-6Al-4V, Int. J. Mach. Tools Manuf., 2001, 41, 1417–1437

Kopac J., Achievements of sustainable manufacturing by machining, J. Achiev. Mater. Manuf. Eng., 2009, 34(2), 180–187

Pusavec F., Krajnik P., and Kopac J., Transitioning to sustainable production – Part 1: Application on machining technologies, J. Clean. Prod., 2010, 18, 174–184

Varadarajan A.S., Philip P.K., and Ramamoorthy B., Investigations on hard turning with minimal cutting fluid application (HTMF) and its comparison with dry and wet turning, Int. J. Mach. Tools Manuf., 2002, 42, 193–200

Wertheim R., Rotbery J., and Ber A., Influence of high-pressure flushing through the rake face of the cutting tool, Annals of the CIRP, 1992, 4(1), 101–106

Rahman M., Kumar A.S., and Choudhury M.R., Identification of effective zones for high pressure coolant in milling, Ann. CIRP, 2000, 49(1), 47–52

Kovacevic R., Cherukuthota C., and Mazurkiewicz M., High pressure waterjet cooling/lubrication to improve machining efficiency in milling, Int. J. Mach. Tools Manuf., 1995, 35(10), 1459–1473 Klocke F., Sangermann H., Kramer A., and Lung D., Influence of a high-pressure lubricoolant supply on thermo-mechanical tool load and tool wear behavior in the turning of aerospace materials, Proc. Inst. Mech. Eng. B J. Eng. Manuf., 2011, 225, 52

Timmerhaus K.D., and Reed R.P., Cryogenic Engineering: Fifty Years of Progress, Springer Verlag, Berlin, 2007.

Shokrani A., Dhokia V., and Newman S.T., Environmentally conscious machining of difficult-tomachine materials with regard to cutting fluids, Int. J. Mach. Tools Manuf., 2012, 57, 83–101 Sadeghi M.H., Hadad M.J., Tawakoli T., Vesali A., and Emami M., An investigation on surface grinding of AISI 4140 hardened steel using minimum quantity lubrication-MQL technique, Int. J. Mater. Form., 2010, 3(4), 241–251

Da Silva L.R., Bianchi E.C., Fusse R.Y., Catai R.E., Franc T.V., and Aguiar P.R., Analysis of surface integrity for minimum quantity lubricant—MQL in grinding, Int. J. Mach. Tools Manuf., 2007, 47, 412–418

Zeilmann R.P., and Weingaertner W.L., Analysis of temperature during drilling of Ti6Al4V with minimal quantity of lubricant, J. Mater. Process. Technol., 2006, 179, 124–127

Heinemann R., Hinduja S., Barrow G., and Petuelli G., Effect of MQL on the tool life of small twist drills in deep-hole drilling, Int. J. Mach. Tools Manuf., 2006, 46, 1–6

Davim J.P., Sreejith P.S., Gomes R., and Peixoto C., Experimental studies on drilling of aluminium (AA1050) under dry, minimum quantity of lubricant, and flood-lubricated conditions, Proc. IMechE Part B: J. Eng. Manuf., 2006, 220, 1605–1611

Astakhov V.P., Ecological Machining: Near-Dry Machining, in P.J. Davim (Ed.): Machining Fundamentals and Recent Advances, Springer, London, 2008, pp. 195–223

Rohrer H., The nanoworld: Chances and challenges. Microelectron. Eng., 1996, 32, 5–14 Eastman J.A., Phillpot S.R., Choi S.U.S., and Keblinski P., Thermal transport in nanofluids, Annu. Rev. Mater. Res., 2004, 34, 219–246

Lienhard I.V., and John H. A Heat Transfer Text Book, 3rd edn., Phlogiston Press, Cambridge, 2004.

Zhang Y., Zeng G., Piprek J., Bar-Cohen A., and Shakouri A., Superlattice microrefrigerators flip-clip bonded with optoelectronic devices, Journal IEEE Trans. Computer Packing Tech., 28(4), 2005, 658–666

Vanam K., Junghans J., Barlow F., Selvam R.P., Balda J.C., and Elshabini A., A novel packaging methodology for spray cooling of power semiconductor devices using dielectric liquids, in Applied Power Electronics Conf. Exposition, APEC2005, Vol. 3, 2005, pp. 2014–2018 Peterson G.P., An Introduction to Heat Pipes Modeling, Testing, and Applications, John Wiley & Sons, New York, 1994.

Kumar D.H., Patel H.E., Kumar V.R., Sundararajan T., Pradeep T., and Das S.K., Model for heat conduction in nanofluids, Phys. Rev. Lett., 2004, 93(14), 144301.

Xie H., Wang T., Xi T., and Ai F., Thermal conductivity enhancement of suspensions containing Nanosized Alumina particles, J. Appl. Phys., 2002, 91, 4568–4572

Koo J., and Kleinstreuer C., Impact analysis of nanoparticles motion mechanisms on the thermal conductivity of nanofluids, Int. Commun. Heat Mass Transfer, 2005, 32(9), 1111–1118 Prasher R., Bhattacharya P., and Phelan P.E., Thermal conductivity of Nanoscale colloidal solutions (Nanofluids), Phys. Rev. Lett., 2005, 94, 025901.

Nan C.W., Liu G., Lin Y., and Li M., Interface effective thermal conductivity of carbon nanotube composites, Appl. Phys. Lett., 2004, 85, 3549–3551

Prasher R., Applications of Nanotechnology in Electronics Cooling, Seminar, Troy, New York, March 17, 2006.

Suresh S., Venkitaraj K.P., Selvakumar P., and Chandrasekar M., Synthesis of Al2O3-Cu/water hybrid nanofluids using two step method and its thermo physical properties, Colloids Surf. A: Physicochem. Eng. Asp., 2011, 388, 41–48

Vasu V., and Reddy G.P.K., Effect of minimum quantity lubrication with Al2O3 nanoparticles on surface roughness, tool wear and temperature dissipation in machining Inconel 600 alloy, Proc. Inst. Mech. Eng. Part N: J. Nanoeng. Nanosys., 2011, 225, 3–16

Amrita M., Srikant R., Sitaramaraju A., Prasad M., and Krishna P.V., Experimental investigations on influence of mist cooling using nanofluids on machining parameters in turning AISI 1040 steel, Proc. Ins. Mech. Eng., Part J: J. Eng. Trib. 2013, 227, 1334–1346

Rao D.N., Srikant R., Krishna P.V., and Subrahmanyam M., Nanocutting fluids in minimum quantity lubrication, in International Multi-Conference on Engineering and Technological Innovation (IMETI 2008), Orlando, FL, USA, 2008.

Rao S.N., Satyanarayana B., and Venkatasubbaiah K., Experimental estimation of tool wear and cutting temperatures in MQL using cutting fluids with CNT inclusion, Int. J. Eng. Sci. Technol., 2011, 3, 2928–2931

Prasad M., and Srikant R., Performance evaluation of nano graphite inclusions in cutting fluids with MQL technique in turning of AISI 1040 Steel, Int. J. Res. Eng. Tech., 2013, 2, 381–393 Kumar M.S., and Kumar K.S., An investigation multi walled carbon nanotubes based nano cutting fluids in turning of martensitic stainless steel by using Taguchi and Anova analysis, IOSR J. Mech. Civil Eng., 2010, 7, 8–15

Roy S., and Ghosh A., High-speed turning of AISI 4140 steel by multi-layered TiN top-coated insert with minimum quantity lubrication technology and assessment of near tool-tip temperature using infrared thermography, Proc. Ins. Mech. Eng., Part B: J. Eng. Man., 2014, 228(9), 1058–1067

Behera B.C., Ghosh S., and Rao P.V., Application of nanofluids during minimum quantity lubrication: A case study in turning process, Tribol. Int., 2016, 101, 234–246

Masuda H., Ebata A., Teramae K., and Hishinuma N., Alteration of thermal conductivity and viscosity of liquid by dispersing ultra-fine particles dispersion of y-A12O3, SiO2, and TiO2 ultra-fine particles, Netsu. Bussei., 1993, 4, 227–233

Lee S.P., Choi S.U.S., Li S., and Eastman J.A., Measuring thermal conductivity of fluids containing oxide nanoparticles, ASME J. Heat Transfer, 1999, 121, 280–289

Xie H., Wang J., Xi T., and Liu Y., Thermal conductivity of suspensions containing nanosized SiC particles, Int. J. Thermophys., 2002, 23(2), 571–580

Putra N., Roetzel W., and Das S.K., Natural convection of nanofluids, Heat Mass Transf., 2003, 39, 775–784

Chon C.H., Kihm K.D., Lee S.P., and Choi S.U.S., Empirical correlation finding the role of temperature and particle size for nanofluid (Al2O3) thermal conductivity enhancement, Appl. Phys. Lett., 2005, 87, 153107.

Wang X., Xu, X., and Choi, S.U.S., Thermal conductivity of nanoparticle-fluid mixture, J. Thermophys. Heat Transf., 1999, 13, 474–480

Lee J.H., Hwang K.S., Jang S.P., Lee B.H., Kim J.H., Choi S.U.S., and Choi C.J., Effective viscosities and thermal conductivities of aqueous nanofluids containing low volume concentrations of Al2O3 nanoparticles, Int. J. Heat Mass Transf., 2008, 51, 2651–2656 Zhang X., Gu H., and Fujii M., Effective thermal conductivity and thermal diffusivity of nanofluids containing spherical and cylindrical nanoparticles, Exp. Thermal Fluid Sci., 2006, 31, 593–599 Timofeeva E.V., Gavrilov A.N., McCloskey J.M., and Tolmachev Y.V., Thermal conductivity and particle agglomeration in alumina nanofluids: experiment and theory, Phys. Rev. E, 2007, 76, 061203

Wei Y., and Xie H., A review on nanofluids: preparation, stability mechanisms, and applications, J. Nanomater., 2012, 1–17

Zhou L.P., and Wang B.X., Experimental research on the thermophysical properties of nanoparticle suspensions using the quasi-steady method, *Annu. Proc. Chin. Eng. Thermophysics*, 2002, 889–892

Das S.K., Putra N., Thiesen P., and Roetzel W., Temperature dependence of thermal conductivity enhancement for nanofluids, ASME J. Heat Transfer, 2003, 125, 567–574 Murshed S.M.S., Leong K.C., and Yang C., Enhanced thermal conductivity of TiO2-water based nanofluids, Int. J. Therm. Sci., 2005, 44, 367–373

Duangthongsuk W., and Wongwises S., Measurement of temperature-dependent thermal conductivity and viscosity of TiO2-water nanofluids, Exp. Thermal Fluid Sci., 2009, 33, 706–714 Xie H., Wang J., Xi T., and Liu Y., Thermal conductivity of suspensions containing nanosized SiC particles, Int. J. Thermophys., 2002, 23(2), 571–580

Choi S.U.S., Yu W., Hull J.R., Zhang Z.G., and Lockwood F.E.. Nanofluids for vehicle thermal management, *SAE Technical Paper 2001-01-1706*; 2001, 139–144

Assael M.J., Metaxa I.N., Kakosimos K., and Constantinou D., Thermal conductivity of nanofluids—Experimental and theoretical, Int. J. Thermophys., 2006, 27, 999–1017 Eastman J.A., Choi S.U.S., Li S., Yu W., and Thompson L.J., Anomalously increased effective thermal conductivities of ethylene glycol-based nanofluids containing copper nanoparticles, Appl. Phys. Lett., 2001, 78, 718–720

Xuan Y., and Li Q., Heat transfer enhancement of nanofluids, Int. J. Heat Fluid Flow, 2005, 2, 58–64

Hong T.K., Yang H.S., and Choi C.J., Study of the enhanced thermal conductivity of Fe nanofluids, J. Appl. Phys., 2005, 97, 064311

Patel H.E., Das S.K., and Sundararajan T., Thermal conductivities of naked and monolayer protected metal nanoparticle based nanofluids: manifestation of anomalous enhancement and chemical effects, Appl. Phys. Lett., 2003, 83, 2931–2933

Putnam S.A., Cahill D.G., and Braun P.V., Thermal conductivity of nanoparticle suspensions, J. Appl. Phys., 2006, 99, 084308

Murshed S.M.S., Leong K.C., and Yang C., Investigations of thermal conductivity and viscosity of nanofluids, Int. J. Therm. Sci., 2008, 47, 560–568

Assael M.J., Chen C.F., Metaxa I., and Wakeham W.A., Thermal conductivity of suspensions of carbon nanotubes in water, Int. J. Thermophys., 2005, 25(4), 971–985

Liu M.S., Lin M.C.C., Huang I.T., and Wang C.C., Enhancement of thermal conductivity with carbon nanotube for nanofluids, Int. Commun. Heat Mass Transfer, 2005, 32, 1202–1210 Biercuk M.J., Llaguno M.C., Radosavljevic M., Hyun J.K., Johnson A.T., and Fischer J.E., Carbon nanotube composites for thermal management, Appl. Phys. Lett., 2002, 80, 2767–2769 Chopkar M., Sudarshan S., Das P.K., and Manna I., Effect of particle size on thermal conductivity of nanofluids, Metall. Mater. Trans. A, 2008, 39, 1535–1542

Suresh S., Venkitaraj K.P., Selvakumar P., and Chandrasekar M., Synthesis of Al2O3-Cu/water hybrid nanofluids using two step method and its thermo physical properties, Colloids and Surfaces A: Physicochemical Engineering Aspects, 2011, 388, 41–48

Madhesh D., Parameshwaran R., and Kalaiselvam S., Experimental investigation on convective heat transfer and rheological characteristics of Cu–TiO2 hybrid nanofluid, Exp. Thermal Fluid Sci., 2014, 52, 104–115

Syam Sundar L., Singh M.K., and Sousa A.C.M., Enhanced heat transfer and friction factor of MWCNT–Fe3O4/water hybrid nanofluids, Int. Commun. Heat Mass Transfer, 2014, 52, 73–83

Mechiri S.K., Vasu V., and Venu Gopal A., Thermal conductivity and viscosity of vegetable oil–based Cu, Zn, and Cu–Zn hybrid nanofluids, J. Test. Eval., 2015, 44, 1–7, doi:10.1520/JTE20140286

Mechiri S.K., Vasu V., and Venu Gopal A., Investigation of thermal conductivity and rheological properties of vegetable oil based hybrid nanofluids containing Cu-Zn hybrid nanoparticles, Exp. Heat Transf., 2016, 30, 205–217, doi:10.1080/08916152.2016.1233147

Shen B., Shih A.J., Tung S.C., and Hunter M., Application of nanofluids in minimum quantity lubrication grinding, Tribol. Trans., 2008, 51(6), 730–737

Malkin S., and Sridharan U., Effect of minimum quantity lubrication (MQL) with nanolfuids on grinding behavior and thermal distortion, Trans. NAMRI/SME, 2009, 37, 629–636

Nam J.S., Lee P.H., and Lee S.W., Experimental characterization of micro-drilling process using nanofluid minimum quantity lubrication, Int. J. Mach. Tools Manuf., 2011, 51(7–8), 649–652 Lee P.-H., Nam T.-S., Li C., and Lee S.W., Experimental study on meso-scale milling process using nanofluid minimum quantity lubrication and compressed chilly air, in Proceedings of the International Conference on Mechanical, Inductrial, and Manufacturing Technologies (MIMT 2010), Sanya, China, 2010, pp. 457–463

Shen B., Shih A.J., and Tung S.C., Application of nanofluids in minimum quantity lubrication grinding, Tribol. Trans., 2008, 51, 730–737

Sanchez J., Pombo I., Alberdi R., Izquierdo B., Ortega N., Plaza S., and Martinez Toledano J., Machining evaluation of a hybrid MQL-CO2 grinding technology, J. Clean. Prod., 2010, 18, 1840–1849

Vasu V., and Kumar K.M., Analysis of nanofluids as cutting fluid in grinding EN-31 steel, Nano-Micro Lett., 2011, 3, 209–214

Setti D., Ghosh S., and Rao P.V., Application of nano cutting fluid under minimum quantity lubrication (MQL) technique to improve grinding of Ti–6Al–4V alloy, Proc. World Acad. Sci. Eng. Tech., 2012, 70, 512–516

Byers J.P., Metalworking Fluids, CRC Press, USA, 2012

Ghani J., Choudhury I., and Hassan H., Application of Taguchi method in the optimization of end milling parameters, J. Mater. Process. Technol., 2004, 145, 84–92

Razfar M.R., Principles of Machining, Amirkabir University of Technology Press, Tehran, Iran, 2010

Srikant R., Rao D., Subrahmanyam M., and Krishna V.P., Applicability of cutting fluids with nanoparticle inclusion as coolants in machining, Proc. Ins. Mech. Eng., Part J: J. Eng. Trib., 2009, 223, 221–225

Rahmati B., Sarhan A.A., and Sayuti M., Investigating the optimum molybdenum disulfide (MoS2) nanolubrication parameters in CNC milling of AL6061-T6 alloy, Int. J. Adv. Manuf. Technol., 2014, 70, 1143–1155

Lawal S., Choudhury I., and Nukman Y., Application of vegetable oil-based metalworking fluids in machining ferrous metals—A review, Int. J. Mach. Tools Manuf., 2012, 52, 1–12

Alinejad G., and Ghafari M., Principles of Universal Machining Machine Tool, Babol University Technology Press, Babol, Iran, 2011

Vasu V., and Reddy G.P.K., Effect of minimum quantity lubrication with Al2O3 nanoparticles on surface roughness, tool wear and temperature dissipation in machining Inconel 600 alloy, Proc. Inst. Mech. Eng., Part N: J. Nanoeng. Nanosys., 2011, 225, 3–16

Amrita M., Srikant R., Sitaramaraju A., Prasad M., and Krishna P.V., Experimental investigations on influence of mist cooling using nanofluids on machining parameters in turning AISI 1040 steel, Proc. Ins. Mech. Eng., Part J: J. Eng. Trib., 2013, 227, 1334–1346 Marcon A., Melkote S., Kalaitzidou K., and Debra D., An experimental evaluation of graphite nanoplatelet based lubricant in micro-milling, CIRP Ann. Man. Tech., 2010, 59, 141–144 Nam J.S., Lee P.H., and Lee S.W., Experimental characterization of micro-drilling process using nanofluid minimum quantity lubrication, Int. J. Mach. Tools Manuf., 2011, 51, 649–652 Mao C., Tang X., Zou H., Huang X., and Zhou Z., Investigation of grinding characteristic using nanofluid minimum quantity lubrication, Int. J. Precision Eng. Man., 2012, 13, 1745–1752 Dahmus J.B., and Gutowski T.G., An environmental analysis of machining, in ASME 2004 International Mechanical Engineering Congress and Exposition, American Society of Mechanical Engineers, 2004. (BOOK or JOURNAL (give VOLUME and ISSUE) Lee P.H., Nam J.S., Li C., and Lee S.W., An experimental study on micro-grinding process with nanofluid minimum quantity lubrication (MQL), Int. J. Precision Eng. Man., 2012, 13, 331–338

Mao C., Zou H., Huang X., Zhang J., and Zhou Z., The influence of spraying parameters on grinding performance for nanofluid minimum quantity lubrication, Int. J. Adv. Manuf. Technol., 2013 64, 1791–1799

Prabhu S., and Vinayagam B., Analysis of surface characteristics by electrolytic inprocess dressing (ELID) technique for grinding process using single wall carbon nano tube-based nanofluids, Arab. J. Sci. Eng., 2013, 38, 1169–1178

Prabhu S., and Vinayagam B., Nano surface generation of grinding process using carbon nano tubes, Sadhana, 2010, 35, 747–760

Mao C., Zou H., Zhou X., Huang Y., Gan H., and Zhou Z., Analysis of suspension stability for nanofluid applied in minimum quantity lubricant grinding, Int. J. Adv. Manuf. Technol., 2014, 71, 2073–2081

Mao C., Zhang J., Huang Y., Zou H., Huang X., and Zhou Z., Investigation on the effect of nanofluid parameters on MQL grinding, Mater. Manuf. Process., 2013, 28, 436–442

Mao C., Zou H., Huang Y., Li Y., and Zhou Z., Analysis of heat transfer coefficient on workpiece surface during minimum quantity lubricant grinding, Int. J. Adv. Manuf. Technol., 2013, 66, 363–370

Lee P.H., Nam T.S., Li C., and Lee S.W., Environmentally-friendly nano-fluid minimum quantity lubrication (MQL) meso-scale grinding process using nanodiamond particles, in Manufacturing Automation (ICMA), International Conference on IEEE, The University of Hong Kong, China, 2010

Rahman M.M., Kadirgama K., and Aziz A., Artificial neural network modeling of grinding of ductile cast iron using water based SiO2 Nanocoolant, Int. J. Autom. Mech. Eng., 2013, 9, 1649–1661

Sodavadia K., and Makwana A., Experimental investigation on the performance of coconut oil based nano fluid as lubricants during turning of AISI 304 austenitic stainless steel, Int. J. Adv. Mech. Eng., 2014, 4, 55–60

Shabgard M.R., Jafarian Zenjanab M., and Azarafza R., Experimental study on the influence of CuO nanofluid on surface roughness and machining force in turning of AISI 4340 steel, Modares Mech. Eng., 2014, 14, 27–33

Rao D.N., Srikant R., Krishna P.V., and Subrahmanyam M., Nano cutting fluids in minimum quantity lubrication, in International Multi-Conference on Engineering and Technological Innovation (IMETI 2008), Orlando, FL, USA, 2008

Vamsi Krishna P., Srikant R., and Nageswara Rao D., Experimental investigation on the performance of nanoboric acid suspensions in SAE-40 and coconut oil during turning of AISI 1040 steel, Int. J. Mach. Tools Manuf., 2010, 50, 911–916

Saravanakumar N., Prabu L., Karthik M., and Rajamanickam A., Experimental analysis on cutting fluid dispersed with silver nano particles, J. Mech. Sci. Technol., 2014, 28, 645–651 Rao S.N., Satyanarayana B., and Venkatasubbaiah B., Experimental estimation of tool wear and cutting temperatures in MQL using cutting fluids with CNT inclusion, Int. J. Eng. Sci. Technol., 2011, 3, 2928–2931

Prasad M., and Srikant R., Performance evaluation of nano graphite inclusions in cutting fluids with MQL technique in turning of AISI 1040 steel, Int. J. Res. Eng. Tech., 2013, 2, 381–393 Kumar M.S., and Kumar K.S., An investigation multi walled carbon nanotubes based nano cutting fluids in turning of martensitic stainless steel by using Taguchi and Anova analysis, IOSR J. Mech. Civil Eng., 2010, 7, 8–15

Roy S., and Ghosh A., High-speed turning of AISI 4140 steel by multi-layered TiN top-coated insert with minimum quantity lubrication technology and assessment of near tool-tip temperature using infrared thermography, Proc. Ins. Mech. Eng., Part B: J. Eng. Man., 2014, 228(9), 1058–1067

Poorazari S., Effect of alumina nanoparticles on the surface finish of the gear lubricant oil contains the simple production process, Hobbing Mech. Eng., 2014, 93, 38–47

Rahman M., and Kadirgama K., Performance of water-based zinc oxide nanoparticle coolant during abrasive grinding of ductile cast iron, J. Nanoparticles, 2014, 2014, 1–7

Weinert K., Inasaki I., Sutherland J., and Wakabayashi T., Dry machining and minimum quantity lubrication, CIRP Annals-Manuf. Tech., 2004, 53, 511–537

Nouari M., List G., Girot F., and Coupard D., Experimental analysis and optimization of tool wear in dry machining of aluminum alloys, Wear, 2003, 255, 1359–1368

Kuram E., Ozcelik B., Cetin M.H., Demirbas E., and Askin S., Effects of blended vegetablebased cutting fluids with extreme pressure on tool wear and force components in turning of Al 7075-T6, Lubr. Sci., 2013, 25, 39–52

Cetin M.H., Ozcelik B., Kuram E., and Demirbas E., Evaluation of vegetable based cutting fluids with extreme pressure and cutting parameters in turning of AISI 304L by Taguchi method, J. Clean. Prod., 2011, 19, 2049–2056

Bartz W.J., Ecological and environmental aspects of cutting fluids, Lubrication Eng. Illinois, 2001, 57, 13–16

Shokoohi Y., Khosrojerdi E., and Rassolian Shiadhi B.H., Machining and ecological effects of a new developed cutting fluid in combination with different cooling techniques on turning operation, J. Clean. Prod., 2015, 94, 330–339

Shekarian E., Tarighaleslami A.H., and Khodaverdi F., Review of effective parameters on the nanofluid thermal conductivity, J. Middle East App. Sci. Tech., 2014, 15, 776–780

Nanofluids for Machining in the Era of Industry 4.0

Bai, X., Li, C., Dong, L., & Yin, Q. (2019). Experimental evaluation of the lubrication performances of different nanofluids for minimum quantity lubrication (MQL) in milling Ti-6Al-4V. The International Journal of Advanced Manufacturing Technology, 101(9), 2621–2632. Balandin, A. A. (2011). Thermal properties of graphene and nanostructured carbon materials. Nature Materials, 10(8), 569–581.

Berman, D., Erdemir, A., & Sumant, A. V. (2013). Few layer graphene to reduce wear and friction on sliding steel surfaces. Carbon, 54, 454–459.

Calvert, G. M., Ward, E., Schnorr, T. M., & Fine, L. J. (1998). Cancer risks among workers exposed to metalworking fluids: A systematic review. American Journal of Industrial Medicine, 33(3), 282–292.

Chaudhari, S., Chakule, R., & Talmale, P. (2019). Experimental study of heat transfer characteristics of Al2O3 and CuO nanofluids for machining application. Materials Today: Proceedings, 18, 788–797.

Chen, J., & Collier, C. P. (2005). Noncovalent functionalization of single-walled carbon nanotubes with water-soluble porphyrins. The Journal of Physical Chemistry B, 109(16), 7605–7609.

Comyn, J. (1995). Surface characterization of pentaerythritol rosin ester. International Journal of Adhesion and Adhesives, 15(1), 9–14.

Debnath, S., Reddy, M. M., & Yi, Q. S. (2014). Environmental friendly cutting fluids and cooling techniques in machining: A review. Journal of Cleaner Production, 83, 33–47.

Howes, T., Toenshoff, H., Heuer, W., & Howes, T. (1991). Environmental aspects of grinding fluids. CIRP Annals, 40(2), 623–630.

Ibrahim, A. M. M., Li, W., Xiao, H., Zeng, Z., Ren, Y., & Alsoufi, M. S. (2020). Energy conservation and environmental sustainability during grinding operation of Ti–6Al–4V alloys via eco-friendly oil/graphene nano additive and Minimum quantity lubrication. Tribology International, 150, 106387.

Ibrahim, A. M. M., Shi, X., Zhai, W., & Yang, K. (2015). Improving the tribological properties of NiAI matrix composites via hybrid lubricants of silver and graphene nano platelets. RSC Advances, 5(76), 61554–61561.

Ibrahim, A. M. M., Shi, X., Zhai, W., Yao, J., Xu, Z., Cheng, L., ... Wang, Z. (2014). Tribological behavior of NiAl–1.5 wt% graphene composite under different velocities. Tribology Transactions, 57(6), 1044–1050.

Ibrahim, A. M. M., Shi, X., Zhang, A., Yang, K., & Zhai, W. (2015). Tribological characteristics of NiAl matrix composites with 1.5 wt% graphene at elevated temperatures: An experimental and theoretical study. Tribology Transactions, 58(6), 1076–1083.

Ilyas, S. U., Pendyala, R., & Marneni, N. (2014). Preparation, sedimentation, and agglomeration of nanofluids. Chemical Engineering & Technology, 37(12), 2011–2021.

Jailani, S., Franks, G. V., & Healy, T. W. (2008). ζ potential of nanoparticle suspensions: Effect of electrolyte concentration, particle size, and volume fraction. Journal of the American Ceramic Society, 91(4), 1141–1147.

Johnson, D. W., Dobson, B. P., & Coleman, K. S. (2015). A manufacturing perspective on graphene dispersions. Current Opinion in Colloid & Interface Science, 20(5–6), 367–382. Kalita, P., Malshe, A. P., Kumar, S. A., Yoganath, V., & Gurumurthy, T. (2012). Study of specific energy and friction coefficient in minimum quantity lubrication grinding using oil-based nanolubricants. Journal of Manufacturing Processes, 14(2), 160–166.

Kipling, M. (1977). Health hazards from cutting fluids. Tribology International, 10(1), 41–46. Klocke, F., & Eisenblätter, G. (1997). Dry cutting. Cirp Annals, 46(2), 519–526.

Kristiansen, K., Zeng, H., Wang, P., & Israelachvili, J. N. (2011). Microtribology of aqueous carbon nanotube dispersions. Advanced Functional Materials, 21(23), 4555–4564.

Kumar Singh, R., Sharma, A. K., Mandal, V., Gaurav, K., Nag, A., Kumar, A., ... Das, A. K. (2018). Influence of graphene-based nanofluid with minimum quantity lubrication on surface roughness and cutting temperature in turning operation. Materials Today: Proceedings, 5(11), 24578–24586.

Lee, P.-H., Lee, S. W., Lim, S.-H., Lee, S.-H., Ko, H. S., & Shin, S.-W. (2015). A study on thermal characteristics of micro-scale grinding process using nanofluid minimum quantity lubrication (MQL). International Journal of Precision Engineering and Manufacturing, 16(9), 1899–1909.

Li, Y., Tung, S., Schneider, E., & Xi, S. (2009). A review on development of nanofluid preparation and characterization. Powder Technology, 196(2), 89–101.

Mackerer, C. R. (1989). Health effects of oil mists: A brief review. Toxicology and Industrial Health, 5(3), 429–440.

Mao, C., Zhang, J., Huang, Y., Zou, H., Huang, X., & Zhou, Z. (2013). Investigation on the effect of nanofluid parameters on MQL grinding. Materials and Manufacturing Processes, 28(4), 436–442.

Moyano, M. A., París, R., & Martín-Martínez, J. M. (2016). Changes in compatibility, tack and viscoelastic properties of ethylene n-butyl acrylate (EBA) copolymer–pentaerythritol rosin ester blend by adding microcrystalline wax, Fischer–Tropsch wax and mixture of waxes. International Journal of Adhesion and Adhesives, 65, 47–53.

Nam, J. S., Lee, P.-H., & Lee, S. W. (2011). Experimental characterization of micro-drilling process using nanofluid minimum quantity lubrication. International Journal of Machine Tools and Manufacture, 51(7–8), 649–652.

Pal, A., Chatha, S. S., & Sidhu, H. S. (2021). Performance evaluation of the minimum quantity lubrication with Al2O3-mixed vegetable-oil-based cutting fluid in drilling of AISI 321 stainless steel. Journal of Manufacturing Processes, 66, 238–249.

Pathak, Y., & Dorle, A. (1987). Study of rosin and rosin derivatives as coating materials for controlled release of drug. Journal of Controlled Release, 5(1), 63–68.

Pei, X., Hu, L., Liu, W., & Hao, J. (2008). Synthesis of water-soluble carbon nanotubes via surface initiated redox polymerization and their tribological properties as water-based lubricant additive. European Polymer Journal, 44(8), 2458–2464.

Pop, E., Varshney, V., & Roy, A. K. (2012). Thermal properties of graphene: Fundamentals and applications. MRS Bulletin, 37(12), 1273–1281.

Rahman, M. A., Lokupitiya, H. N., Ganewatta, M. S., Yuan, L., Stefik, M., & Tang, C. (2017). Designing block copolymer architectures toward tough bioplastics from natural rosin. Macromolecules, 50(5), 2069–2077.

Sahu, N. K., Andhare, A. B., & Raju, R. A. (2018). Evaluation of performance of nanofluid using multiwalled carbon nanotubes for machining of Ti–6AL–4V. Machining science and Technology, 22(3), 476–492.

Sarkar, S., & Datta, S. (2021). Machining performance of inconel 718 under dry, MQL, and nanofluid MQL conditions: Application of coconut oil (base fluid) and multi-walled carbon nanotubes as additives. Arabian Journal for Science and Engineering, 46(3), 2371–2395.

Scharrer, R. P., & Epstein, M. (1979). Oxygen-stable rosin-primary polyhydric aliphatic alcohol esters and a method for preparing the same utilizing arylsulfonic acid catalysis: Google Patents. Setti, D., Sinha, M. K., Ghosh, S., & Rao, P. V. (2015). Performance evaluation of Ti–6Al–4V grinding using chip formation and coefficient of friction under the influence of nanofluids. International Journal of Machine Tools and Manufacture, 88, 237–248.

Shabgard, M., Seyedzavvar, M., & Mohammadpourfard, M. (2017). Experimental investigation into lubrication properties and mechanism of vegetable-based CuO nanofluid in MQL grinding. The International Journal of Advanced Manufacturing Technology, 92(9–12), 3807–3823.

Sheorey, D., & Dorle, A. (1994). Preparation and study of release kinetics of rosin pentaerythritol ester microcapsules. Journal of Microencapsulation, 11(1), 11–17. Shokrani, A., Dhokia, V., & Newman, S. T. (2012). Environmentally conscious machining of difficult-to-machine materials with regard to cutting fluids. International Journal of Machine Tools and Manufacture, 57, 83–101.

Singh, R., Dureja, J. S., Dogra, M., Gupta, M. K., & Mia, M. (2019). Influence of grapheneenriched nanofluids and textured tool on machining behavior of Ti-6AI-4V alloy. The International Journal of Advanced Manufacturing Technology, 105(1), 1685–1697.

Singh, R., Dureja, J., Dogra, M., Gupta, M. K., Mia, M., & Song, Q. (2020). Wear behavior of textured tools under graphene-assisted minimum quantity lubrication system in machining Ti-6AI-4V alloy. Tribology International, 145, 106183.

Singh, H., Sharma, V. S., & Dogra, M. (2020). Exploration of graphene assisted vegetables oil based minimum quantity lubrication for surface grinding of TI-6AL-4V-ELI. Tribology International, 144, 106113.

Singh, H., Sharma, V. S., Singh, S., & Dogra, M. (2019). Nanofluids assisted environmental friendly lubricating strategies for the surface grinding of titanium alloy: Ti6Al4V-ELI. Journal of Manufacturing Processes, 39, 241–249.

Sinha, M. K., Madarkar, R., Ghosh, S., & Rao, P. V. (2017). Application of eco-friendly nanofluids during grinding of Inconel 718 through small quantity lubrication. Journal of Cleaner Production, 141, 1359–1375.

Şirin, Ş., & Kıvak, T. (2019). Performances of different eco-friendly nanofluid lubricants in the milling of Inconel X-750 superalloy. Tribology International, 137, 180–192.

Uysal, A., Demiren, F., & Altan, E. (2015). Applying minimum quantity lubrication (MQL) method on milling of martensitic stainless steel by using nano MoS2 reinforced vegetable cutting fluid. Procedia-Social and Behavioral Sciences, 195, 2742–2747.

Wang, Y., Li, C., Zhang, Y., Li, B., Yang, M., Zhang, X., ... Zhai, M. (2017). Comparative evaluation of the lubricating properties of vegetable-oil-based nanofluids between frictional test and grinding experiment. Journal of Manufacturing Processes, 26, 94–104.

Wang, Y., Li, C., Zhang, Y., Yang, M., Li, B., Dong, L., & Wang, J. (2018). Processing characteristics of vegetable oil-based nanofluid MQL for grinding different workpiece materials. International Journal of Precision Engineering and Manufacturing-Green Technology, 5(2), 327–339.

Wang, Y., Li, C., Zhang, Y., Yang, M., Li, B., Jia, D., ... Mao, C. (2016). Experimental evaluation of the lubrication properties of the wheel/workpiece interface in minimum quantity lubrication (MQL) grinding using different types of vegetable oils. Journal of Cleaner Production, 127, 487–499.

Wang, Y., Li, C., Zhang, Y., Yang, M., Zhang, X., Zhang, N., & Dai, J. (2017). Experimental evaluation on tribological performance of the wheel/workpiece interface in minimum quantity lubrication grinding with different concentrations of Al2O3 nanofluids. Journal of Cleaner Production, 142, 3571–3583.

Winer, W. O. (1967). Molybdenum disulfide as a lubricant: A review of the fundamental knowledge. Wear, 10(6), 422–452.

Witharana, S., Palabiyik, I., Musina, Z., & Ding, Y. (2013). Stability of glycol nanofluids—the theory and experiment. Powder Technology, 239, 72–77.

Wong, E. W., Sheehan, P. E., & Lieber, C. M. (1997). Nanobeam mechanics: Elasticity, strength, and toughness of nanorods and nanotubes. Science, 277(5334), 1971–1975.

Xu, Z., Lou, W., Zhao, G., Zhang, M., Hao, J., & Wang, X. (2019). Pentaerythritol rosin ester as an environmentally friendly multifunctional additive in vegetable oil-based lubricant. Tribology International, 135, 213–218.

Yao, K., Wang, J., Zhang, W., Lee, J. S., Wang, C., Chu, F., ... Tang, C. (2011). Degradable rosin-ester–caprolactone graft copolymers. Biomacromolecules, 12(6), 2171–2177.

Young, R. J., Kinloch, I. A., Gong, L., & Novoselov, K. S. (2012). The mechanics of graphene nanocomposites: A review. Composites Science and Technology, 72(12), 1459–1476.

Yu, C., Chen, C., Gong, Q., & Zhang, F. A. (2012). Preparation of polymer microspheres with a rosin moiety from rosin ester, styrene and divinylbenzene. Polymer International, 61(11), 1619–1626.

Yu, W., & Xie, H. (2012). A review on nanofluids: Preparation, stability mechanisms, and applications. Journal of nanomaterials, 2012, 1–17.

Zhang, Y., Li, C., Jia, D., Zhang, D., & Zhang, X. (2015). Experimental evaluation of MoS2 nanoparticles in jet MQL grinding with different types of vegetable oil as base oil. Journal of Cleaner Production, 87, 930–940.

Zhang, J. Z., Rao, P., & Eckman, M. (2012). Experimental evaluation of a bio-based cutting fluid using multiple machining characteristics. Wear, 12, 13–14.

Zhu, H.-T., Lin, Y.-S., & Yin, Y.-S. (2004). A novel one-step chemical method for preparation of copper nanofluids. Journal of Colloid and Interface Science, 277(1), 100–103.

Ionic Liquids as a Potential Sustainable Green Lubricant for Machining in the Era of Industry 4.0

Ghosh, A.K., Sharif Ullah, A.M.M., Kubo, A., Akamatsu, T., D'Addona, D.M.: Machining phenomenon twin construction for industry 4.0: A case of surface roughness. J. Manuf. Mater. Process. 4, (2020). https://doi.org/10.3390/jmmp4010011

Zheng, P., Wang, H., Sang, Z., Zhong, R.Y., Liu, Y., Liu, C., Mubarok, K., Yu, S., Xu, X.: Smart manufacturing systems for Industry 4.0: Conceptual framework, scenarios, and future perspectives. Front. Mech. Eng. 13, 137–150 (2018). https://doi.org/10.1007/s11465-018-0499-5

Aoyama, T., Kakinuma, Y., Yamashita, M., Aoki, M.: Development of a new lean lubrication system for near dry machining process. CIRP Ann. - Manuf. Technol. 57, 125–128 (2008). https://doi.org/10.1016/j.cirp.2008.03.094

Denkena, B., Helmecke, P., Hülsemeyer, L.: New production technologies in aerospace industry - 5th machining innovations conference (MIC 2014) energy efficient machining with optimized coolant lubrication flow rates. Procedia CIRP. 24, 25–31 (2014).

https://doi.org/10.1016/j.procir.2014.07.140

Claudin, C., Mondelin, A., Rech, J., Fromentin, G.: Effects of a straight oil on friction at the toolworkmaterial interface in machining. Int. J. Mach. Tools Manuf. 50, 681–688 (2010). https://doi.org/10.1016/j.ijmachtools.2010.04.013

Mondelin, A., Claudin, C., Rech, J., Dumont, F.: Effects of lubrication mode on friction and heat partition coefficients at the tool-work material interface in machining. Tribol. Trans. 54, 247–255 (2011). https://doi.org/10.1080/10402004.2010.538489

Skerlos, S.J., Hayes, K.F., Clarens, A.F., Zhao, F.: Current advances in sustainable Metalworking Fluids research. Int. J. Sustain. Manuf. 1, 180–202 (2008). https://doi.org/10.1504/IJSM.2008.019233

Tawakoli, T., Hadad, M.J., Sadeghi, M.H.: Investigation on minimum quantity lubricant-MQL grinding of 100Cr6 hardened steel using different abrasive and coolant-lubricant types. Int. J. Mach. Tools Manuf. 50, 698–708 (2010). https://doi.org/10.1016/j.ijmachtools.2010.04.009 Peng, Y., Miao, H., Peng, Z.: Development of TiCN-based cermets: Mechanical properties and wear mechanism. Int. J. Refract. Met. Hard Mater. 39, 78–89 (2013). https://doi.org/10.1016/j.ijmmh.2012.07.001

Xu, C.H., Feng, Y.M., Zhang, R.B., Zhao, S.K., Xiao, X., Yu, G.T.: Wear behavior of Al2O3/Ti(C,N)/SiC new ceramic tool material when machining tool steel and cast iron. J. Mater. Process. Technol. 209, 4633–4637 (2009). https://doi.org/10.1016/j.jmatprotec.2008.10.017 Biksa, A., Yamamoto, K., Dosbaeva, G., Veldhuis, S.C., Fox-Rabinovich, G.S., Elfizy, A., Wagg, T., Shuster, L.S.: Wear behavior of adaptive nano-multilayered AlTiN/MexN PVD coatings during machining of aerospace alloys. Tribol. Int. 43, 1491–1499 (2010). https://doi.org/10.1016/j.triboint.2010.02.008

Zhang, S., Guo, Y.B.: An experimental and analytical analysis on chip morphology, phase transformation, oxidation, and their relationships in finish hard milling. Int. J. Mach. Tools Manuf. 49, 805–813 (2009). https://doi.org/10.1016/j.ijmachtools.2009.06.006

Fox-Rabinovich, G.S., Kovalev, A.I., Aguirre, M.H., Beake, B.D., Yamamoto, K., Veldhuis, S.C., Endrino, J.L., Wainstein, D.L., Rashkovskiy, A.Y.: Design and performance of AlTiN and TiAlCrN PVD coatings for machining of hard to cut materials. Surf. Coatings Technol. 204, 489–496 (2009). https://doi.org/10.1016/j.surfcoat.2009.08.021

Rahman Rashid, R.A., Sun, S., Wang, G., Dargusch, M.S.: An investigation of cutting forces and cutting temperatures during laser-assisted machining of the Ti-6Cr-5Mo-5V-4Al beta

titanium alloy. Int. J. Mach. Tools Manuf. 63, 58–69 (2012).

https://doi.org/10.1016/j.ijmachtools.2012.06.004

Schubert, A., Nestler, A., Pinternagel, S., Zeidler, H.: Influence of ultrasonic vibration assistance on the surface integrity in turning of the aluminium alloy AA2017. Materwiss. Werksttech. 42, 658–665 (2011). https://doi.org/10.1002/mawe.201100834

Sun, S., Brandt, M., Dargusch, M.S.: Machining Ti-6Al-4V alloy with cryogenic compressed air cooling. Int. J. Mach. Tools Manuf. 50, 933–942 (2010).

https://doi.org/10.1016/j.ijmachtools.2010.08.003

Umbrello, D., Micari, F., Jawahir, I.S.: The effects of cryogenic cooling on surface integrity in hard machining: A comparison with dry machining. CIRP Ann. - Manuf. Technol. 61, 103–106 (2012). https://doi.org/10.1016/j.cirp.2012.03.052

Dhar, N.R., Kamruzzaman, M.: Cutting temperature, tool wear, surface roughness and dimensional deviation in turning AISI-4037 steel under cryogenic condition. Int. J. Mach. Tools Manuf. 47, 754–759 (2007). https://doi.org/10.1016/j.ijmachtools.2006.09.018

Kuzu, A.T., Berenji, K.R., Ekim, B.C., Bakkal, M.: The thermal modeling of deep-hole drilling process under MQL condition. J. Manuf. Process. 29, 194–203 (2017).

https://doi.org/10.1016/j.jmapro.2017.07.020

Sultan, A.A., Okafor, A.C.: Effects of geometric parameters of wavy-edge bull-nose helical endmill on cutting force prediction in end-milling of Inconel 718 under MQL cooling strategy. J. Manuf. Process. 23, 102–114 (2016). https://doi.org/10.1016/j.jmapro.2016.05.015

Weinert, K., Inasaki, I., Sutherland, J.W., Wakabayashi, T.: Dry machining and minimum quantity lubrication. CIRP Ann. Manuf. Technol. 53, 511–537 (2004).

https://doi.org/10.1016/S0007-8506(07)60027-4

Emami, M., Sadeghi, M.H., Sarhan, A.A.D.: Investigating the effects of liquid atomization and delivery parameters of minimum quantity lubrication on the grinding process of Al2O 3 engineering ceramics. J. Manuf. Process. 15, 374–388 (2013).

https://doi.org/10.1016/j.jmapro.2013.02.004

Liew, W.Y.H.: Low-speed milling of stainless steel with TiAIN single-layer and TiAIN/AICrN nano-multilayer coated carbide tools under different lubrication conditions. Wear. 269, 617–631 (2010). https://doi.org/10.1016/j.wear.2010.06.012

Wang, Z.Y., Rajurkar, K.P.: Cryogenic machining of hard-to-cut materials. Wear. 239, 168–175 (2000). https://doi.org/10.1016/S0043-1648(99)00361-0

Aramcharoen, A., Chuan, S.K.: An experimental investigation on cryogenic milling of inconel 718 and its sustainability assessment. Proceedia CIRP. 14, 529–534 (2014).

https://doi.org/10.1016/j.procir.2014.03.076

Bermingham, M.J., Palanisamy, S., Kent, D., Dargusch, M.S.: A comparison of cryogenic and high pressure emulsion cooling technologies on tool life and chip morphology in Ti-6AI-4V cutting. J. Mater. Process. Technol. 212, 752–765 (2012).

https://doi.org/10.1016/j.jmatprotec.2011.10.027

Kenda, J., Pusavec, F., Kopac, J.: Analysis of residual stresses in sustainable cryogenic machining of nickel based alloy - Inconel 718. J. Manuf. Sci. Eng. Trans. ASME. 133, 1–7 (2011). https://doi.org/10.1115/1.4004610

Shokrani, A., Dhokia, V., Muñoz-Escalona, P., Newman, S.T.: State-of-the-art cryogenic machining and processing. Int. J. Comput. Integr. Manuf. 26, 616–648 (2013). https://doi.org/10.1080/0951192X.2012.749531

Zisman, W.A.: Relation of the Equilibrium Contact Angle to Liquid and Solid Constitution. Adv. Chem. 43, 1–51 (1964). https://doi.org/10.1021/ba-1964-0043.ch001

Madanchi, N., Winter, M., Thiede, S., Herrmann, C.: Energy Efficient Cutting Fluid Supply: The Impact of Nozzle Design. Procedia CIRP. 61, 564–569 (2017).

https://doi.org/10.1016/j.procir.2016.11.192

Bay, N., Azushima, A., Groche, P., Ishibashi, I., Merklein, M., Morishita, M., Nakamura, T., Schmid, S., Yoshida, M.: Environmentally benign tribo-systems for metal forming. CIRP Ann. - Manuf. Technol. 59, 760–780 (2010). https://doi.org/10.1016/j.cirp.2010.05.007

Attanasio, A., Gelfi, M., Giardini, C., Remino, C.: Minimal quantity lubrication in turning: Effect on tool wear. Wear. 260, 333–338 (2006). https://doi.org/10.1016/j.wear.2005.04.024 Walker, T.: The MOL handbook-A guide to maching with Minimum Quantity Lubrication 1–41

Walker, T.: The MQL handbook-A guide to machiing with Minimum Quantity Lubrication. 1–41 (2011)

da Silva, L.R., Bianchi, E.C., Fusse, R.Y., Catai, R.E., França, T.V., Aguiar, P.R.: Analysis of surface integrity for minimum quantity lubricant-MQL in grinding. Int. J. Mach. Tools Manuf. 47, 412–418 (2007). https://doi.org/10.1016/j.ijmachtools.2006.03.015

Goindi, G.S., Sarkar, P., Jayal, A.D., Chavan, S.N., Mandal, D.: Investigation of ionic liquids as additives to canola oil in minimum quantity lubrication milling of plain medium carbon steel. Int. J. Adv. Manuf. Technol. 94, 881–896 (2018). https://doi.org/10.1007/s00170-017-0970-1

Goindi, G.S., Jayal, A.D., Sarkar, P.: Application of ionic liquids in interrupted minimum quantity lubrication machining of plain medium carbon steel: Effects of ionic liquid properties and cutting conditions. J. Manuf. Process. 32, 357–371 (2018).

https://doi.org/10.1016/j.jmapro.2018.03.007

Pham, M.Q., Yoon, H.S., Khare, V., Ahn, S.H.: Evaluation of ionic liquids as lubricants in micro milling - Process capability and sustainability. J. Clean. Prod. 76, 167–173 (2014).

https://doi.org/10.1016/j.jclepro.2014.04.055

Abdul Sani, A.S., Rahim, E.A., Sharif, S., Sasahara, H.: Machining performance of vegetable oil with phosphonium- and ammonium-based ionic liquids via MQL technique. J. Clean. Prod. 209, 947–964 (2019). https://doi.org/10.1016/j.jclepro.2018.10.317

Hallett, J.P., Welton, T.: Room-temperature ionic liquids: Solvents for synthesis and catalysis. 2. Chem. Rev. 111, 3508–3576 (2011). https://doi.org/10.1021/cr1003248

Welton, T.: Room-Temperature Ionic Liquids. Solvents for Synthesis and Catalysis. Chem. Rev. 99, 2071–2084 (1999). https://doi.org/10.1021/cr980032t

Welton, T.: Ionic liquids: a brief history. Biophys. Rev. 10, 691–706 (2018).

https://doi.org/10.1007/s12551-018-0419-2

Rogers, R.D., Seddon, K.R.: Ionic Liquids - Solvents of the Future? Science (80-). 302,

792–793 (2003). https://doi.org/10.1126/science.1090313

Singh, S.K., Savoy, A.W.: Ionic liquids synthesis and applications: An overview. J. Mol. Liq. 297, 112038 (2020). https://doi.org/10.1016/j.molliq.2019.112038

Davis, B., Schueller, J.K., Huang, Y.: Study of ionic liquid as effective additive for minimum quantity lubrication during titanium machining. Manuf. Lett. 5, 1–6 (2015). https://doi.org/10.1016/j.mfglet.2015.04.001

Karthikey, V.V.: Investigations on the effect of Ionic Liquid (BMIMBF4) based Metal Working Fluids applied using Minimum Quantity Lubrication on Surface Roughness. Int. J. Pure Appl. Math. 118, 2283–2293 (2018)

Abdul Sani, A.S., Rahim, E.A., Sharif, S., Sasahara, H.: The influence of modified vegetable oils on tool failure mode and wear mechanisms when turning AISI 1045. Tribol. Int. 129, 347–362 (2019). https://doi.org/10.1016/j.triboint.2018.08.038

Pandey, A., Kumar, R., Sahoo, A.K., Paul, A., Panda, A.: Performance analysis of trihexyltetradecylphosphonium chloride ionic fluid under MQL condition in hard turning. Int. J. Automot. Mech. Eng. 17, 7629–7647 (2020). https://doi.org/10.15282/IJAME.17.1.2020.12.0567 Naresh Babu, M., Anandan, V., Dinesh Babu, M.: Performance of ionic liquid as a lubricant in turning inconel 825 via minimum quantity lubrication method. J. Manuf. Process. 64, 793–804 (2021). https://doi.org/10.1016/j.jmapro.2021.02.011

Saurín, N., Minami, I., Sanes, J., Bermúdez, M.D.: Study of the effect of tribo-materials and surface finish on the lubricant performance of new halogen-free room temperature ionic liquids. Appl. Surf. Sci. 366, 464–474 (2016). https://doi.org/10.1016/j.apsusc.2016.01.127

Cai, M., Yu, Q., Liu, W., Zhou, F.: lonic liquid lubricants: When chemistry meets tribology. Chem. Soc. Rev. 49, 7753–7818 (2020). https://doi.org/10.1039/d0cs00126k

Gusain, R., Mungse, H.P., Kumar, N., Ravindran, T.R., Pandian, R., Sugimura, H., Khatri, O.P.: Covalently attached graphene-ionic liquid hybrid nanomaterials: Synthesis, characterization and tribological application. J. Mater. Chem. A. 4, 926–937 (2016). https://doi.org/10.1039/c5ta08640j

Upendra, M., Vasu, V.: Synergistic Effect Between Phosphonium-Based Ionic Liquid and Three Oxide Nanoparticles as Hybrid Lubricant Additives. J. Tribol. (2020). https://doi.org/10.1115/1.4045769

Nagendramma, P., Khatri, P.K., Thakre, G.D., Jain, S.L.: Lubrication capabilities of amino acid based ionic liquids as green bio-lubricant additives. J. Mol. Liq. 244, 219–225 (2017). https://doi.org/10.1016/j.molliq.2017.08.115

Rohlmann, P., Munavirov, B., Furó, I., Antzutkin, O., Rutland, M.W., Glavatskih, S.: Nonhalogenated ionic liquid dramatically enhances tribological performance of biodegradable oils.

Front. Chem. 7, 1-8 (2019). https://doi.org/10.3389/fchem.2019.00098 Saurín, N., Sanes, J., Bermúdez, M.D.: New Graphene/Ionic Liquid Nanolubricants, Mater. Today Proc. 3. S227–S232 (2016). https://doi.org/10.1016/i.matpr.2016.02.038 Fernando, W.L.R., Sarmilan, N., Wickramasinghe, K.C., Herath, H.M.C.M., Perera, G.I.P.: Experimental investigation of Minimum Quantity Lubrication (MOL) of coconut oil based Metal Working Fluid. Mater. Today Proc. 23, 23-26 (2019). https://doi.org/10.1016/j.matpr.2019.06.079 Belluco, W., De Chiffre, L.: Performance evaluation of vegetable-based oils in drilling austenitic stainless steel, J. Mater. Process. Technol. 148, 171–176 (2004). https://doi.org/10.1016/S0924-0136(03)00679-4 Abdul Sani, A.S., Rahim, E.A., Sharif, S., Sasahara, H.: Machining performance of vegetable oil with phosphonium- and ammonium-based ionic liquids via MOL technique. J. Clean. Prod. 209, 947-964 (2019). https://doi.org/10.1016/j.jclepro.2018.10.317 Gusain, R., Singh, R., Sivakumar, K.L.N., Khatri, O.P.: Halogen-free imidazolium/ammoniumbis(salicylato)borate ionic liquids as high performance lubricant additives. RSC Adv. 4, 1293-1301 (2014). https://doi.org/10.1039/c3ra43052a Fu, X., Sun, L., Zhou, X., Li, Z., Ren, T.: Tribological Study of Oil-Miscible Ouaternary Ammonium Phosphites Ionic Liquids as Lubricant Additives in PAO. Tribol. Lett. 60, 1–12 (2015). https://doi.org/10.1007/s11249-015-0596-0

Sustainable Electrical Discharge Machining Process

Mohd N., Solomon D.G., Bahari M.F. (2007) "A review on current research trends in electrical discharge machining (EDM)," Int J Mach Tools Manuf 47: 1214–1228.

DiBitonto D.D. (1969) Theoretical models of the electrical discharge machining process. I. A simple cathode erosion model. J Appl Phys 66(9): 4095–4103.

Das S., Paul S., Doloi B. (2020) Feasibility assessment of some alternative dielectric mediums for sustainable electrical discharge machining: a review work. J Braz Soc Mech Sci Eng 42(148): 1–21.

Jain V.K., Sidpara A., Balasubramaniam R., Lodha G.S., Dhamgaye V.P. and Shukla R. (2014) "Micromanufacturing: A review - Part I," Proc IMechE Part B: J Eng Manuf 228(9): 973–994. Richard J., Demellayer R. (2013) Micro-EDM-milling development of new machining technology for micro-machining. Procedia CIRP 6: 292–296.

Uhlmann E., Perfilov I. (2018) Machine tool and technology for manufacturing of microstructures by micro dry electrical discharge milling. Procedia CIRP 68: 825–830.

Gotoh H., Tani T., Okada M., Goto A., Masuzawa T., Mohri N. (2013) Wire electrical discharge milling using a wire guide with reciprocating rotation. Procedia CIRP 6: 199–202.

Wang J., Qian J., Ferraris E., Reynaerts D. (2016) Precision Micro-EDM milling of 3D cavities by incorporating *in-situ* pulse monitoring. Procedia CIRP 42: 656–661.

Schulze V., Ruhs C. (2010) On-Machine Measurement for the Micro-EDM-Milling Process Using a Confocal White Light Sensor. In Proc. 10th Int. Conf. Eur. Soc. Precis. Eng. Nanotechnology, EUSPEN 2: 37–40.

Chakraborty S., Dey V., Ghosh S.K. (2015) A review on the use of dielectric fluids and their effects in EDM characteristics. Precis Eng 40: 1–6.

Valaki J.B., Rathod P.P., Sidpara A.M. (2019) Sustainability issues in electric discharge machining. In: Gupta K (ed) Innovations in Manufacturing for Sustainability. Materials forming, machining and tribology. Springer, Cham.

Das S., Paul S., Doloi B. (2019) Investigation of the machining performance of neem oil as a dielectric medium of EDM: a sustainable approach. In: IOP conference series: materials science and engineering, vol 653, no 1. IOP Publishing, p 012017.

Valaki J.B., Rathod P.P., Sankhavara C.D. (2016) Investigations on technical feasibility of Jatropha curcas oil based bio dielectric fluid for sustainable electric discharge machining (EDM). J Manuf Process 22: 151–160.

Leão F.N., Pashby I.R. (2004) A review on the use of environmentally-friendly dielectric fluids in EDM. J Mater Process Technol 149: 341–346.

Zhu G., Bai J., Guo Y., Cao Y., Huang Y. (2014) A study of the effects of working fluids on micro-hole arrays in micro-electrical discharge machining. Proc IMechE, Part B: J Eng Manuf 228(11): 1381–1392.

Chen S.L., Yan B.H., Huang F.Y. (1999) Influence of kerosene and distilled water as dielectrics on the electric discharge machining characteristics of Ti–6A1–4V. J Mater Process Technol 87: 107–111.

Jeswani M.L. (1981) Effect of the addition of graphite powder to kerosene used as the dielectric fluid in electrical discharge machining. Wear 70: 133–139.

Jahan M.P., Rahman M., Wong Y.S. (2010) Modelling and experimental investigation on the effect of nano powder mixed dielectric in micro-electro discharge machining of tungsten carbide. Proc IMechE, Part B: J Eng Manuf 224: 1725–1739.

Ndaliman M.B., Khan A.A., Ali M.Y. (2013) Influence of dielectric fluids on surface properties of electrical discharge machined titanium alloy. Proc IMechE, Part B: J Eng Manuf 227(9): 1310–1316.

Zhang Y., Liu Y., Shen Y., Ji R., Li Z., Zheng C. (2014) Investigation on the influence of the dielectrics on the material removal characteristics of EDM. J Mater Process Technol 214: 1052–1061.

Mariprasath T., Kirubakaran V. (2016) A critical review on the characteristics of alternating liquid dielectrics and feasibility study on pongamia pinnata oil as liquid dielectrics. Renew Sustain Energy Rev 65: 784–799.

Valaki J.B., Rathod P.P. (2016) Investigating feasibility through performance analysis of green dielectrics for sustainable electric discharge machining. Mater Manuf Process 31(4): 541–549. Yan B.H., Tsai H.C., Huang F.Y. (2005) The effect in EDM of a dielectric of a urea solution in water on modifying the surface of titanium. Int J Mach Tools Manuf 45(2): 194–200.

Zhang Y., Liu Y., Ji R., Cai B. (2011) Study of the recast layer of a surface machined by sinking electrical discharge machining using water-in-oil emulsion as dielectric. Appl Surf Sci 257(14): 5989–5997.

Valaki J.B., Rathod P.P. (2016) Assessment of operational feasibility of waste vegetable oil based bio-dielectric fluid for sustainable electric discharge machining (EDM). Int J Adv Manuf Technol 87(5–8): 1509–1518.

Das S., Paul S., Doloi B. (2020) Feasibility investigation of neem oil as a dielectric for electrical discharge machining. Int J Adv Manuf Technol 106: 1179–1189.

Valaki J.B., Rathod P.P., Khatri B.C., Vaghela J.R. (2016) Investigations on palm oil based biodielectric fluid for sustainable electric discharge machining. In: Proceedings of the international conference on advances in materials and manufacturing, India.

Kao C.C., Tao J., Shih A. (2007) Near-dry electrical discharge machining. Int J Mach Tools Manuf 47: 2273–2281.

Sustainable Abrasive Jet Machining

Kovacevic, R., Hashish, M., Mohan, R., Ramulu, M., Kim, J.T. and Geskin, S.E. (1997). State of the art of research and development in abrasive waterjet machining. Journal of Manufacturing Science and Engineering, 119, 776–785.

Zhu, T.H., Huang, Z.C., Wang, J., Li, L.Q. and Che, L.C. (2009). Experimental study on abrasive waterjet polishing for hard-brittle materials. International Journal of Machine Tools and Manufacture, 49, 569–578.

Davim, J.P., Rubio, J.C. and Abrao, A.M. (2007). A novel approach based on digital image analysis to evaluate the delamination factor after drilling composite laminates. Composites Science and Technology, 67(9), 1939–1945.

Bayraktar, Ş. and Hekimoğlu, A.P. (2020). Effect of zinc content and cutting tool coating on the machinability of the AI-(5–35) Zn alloys. Metals and Materials International, 26(4), 477–490. Bayraktar, Ş. and Afyon, F. (2020). Machinability properties of AI–7Si, AI–7Si–4Zn and AI–7Si–4Zn–3Cu alloys. Journal of the Brazilian Society of Mechanical Sciences and Engineering, 42(4), 1–12.

Bayraktar, Ş. and Demir, O. (2020). Processing of T6 heat-treated Al-12Si-0.6 Mg alloy. Materials and Manufacturing Processes, 35(3), 354–362.

Bayraktar, S. (2021). Dry Cutting: A Sustainable Machining Technology. In Sustainable Manufacturing, London, Elsevier.

Bayraktar, S. (2020). Cryogenic Cooling-Based Sustainable Machining. In High Speed Machining, London, Academic Press.

Ramulu, M. and Arora, D. (1993). Water jet and abrasive water jet cutting of undirectional graphite/epoxy composite. Composites, 24(4), 299–308.

Hoogstrate, A.M., and Van Luttervelt, C.A. (1997). Opportunities in abrasive water-jet machining. Annals CIRP, 46(2), 697–714.

Liu, H.T., Cutler, V., Raghavan, C., Miles, P., Schubert, E. and Webers, N. (2018). Advanced Abrasive Waterjet for Multimode Machining, Abrasive Technology-Characteristics and Applications, London, IntechOpen.

Folkes, J. (2009). Waterjet-an innovative tool for manufacturing. Journal of Materials Processing Technology, 209(20), 6181–6189.

Momber, A.W. and Kovacevic, R. (2012). Principles of Abrasive Water Jet Machining. USA, Springer.

Meng, H.C. and Ludema, K.C. (1995). Wear models and predictive equations: their form and content. Wear, 181, 443–457.

Miller, D.S. (2004). Micromachining with abrasive waterjets. Journal of Materials Processing Technology, 149, 37–42.

Youssef, H.A. and El-Hofy, H. (2008). Nontraditional Machine Tools and Operations, In: Machining Technology. Machine Tools and Operations, USA, CRC Press.

Hemanth, P., Naqash, P.V. and Babu, N.R. (2005). A simplified model for predicting the penetration of an abrasive water jet in a piercing operation. International Journal of Manufacturing Technology and Management, 72(4), 366–380.

Jain, V. (2008). Advanced Non-traditional Machining Processes. In: Machining. London, Springer.

Hlaváč, L.M., Hlaváčová, I.M., Gembalová, L., Kaličinský, J., Fabian, S., Měšťánek, J. and Mádr, V. (2009). Experimental method for the investigation of the abrasive water jet cutting quality. Journal of Materials Processing Technology, 209(20), 6190–6195.

Liu, H.T. (2017). "7M" advantage of abrasive waterjet for machining advanced materials. Jornal of Manufacturing and Material Processing, 1(1), 11.

Zaring, K. (1991). Procedure Optimization and Hardware Improvements in Abrasive Waterjet Cutting Systems. 6th American Water Jet Conference. Houston, TX, USA.

Selvan, M.C. and Raju, N.M. (2012). Analysis of surface roughness in abrasive waterjet cutting of cast iron. International Journal of Science Environment and Technology, 1(3), 174–182. Kolahan, F. and Khajavi, H. (2009). A statistical approach for predicting and optimizing depth of

Kolahan, F. and Khajavi, H. (2009). A statistical approach for predicting and optimizing depth of cut in AWJ machining for 6063-T6 al alloy. International Journal of Mechanical and Mechatronics Engineering, 3(11), 1395–1398.

Todkar, M. and Patkure, J. (2014). Fuzzy modelling and ga optimization for optimal selection of process parameters to maximize MRR in abrasive water jet machining. International Journal on Theoretical and Applied Research in Mechanical Engineering, 3(1), 9–16.

Selvan, M.C.P., Raju, N.M.S. and Sachidananda, H.K. (2012). Effects of process parameters on surface roughness in abrasive water jet cutting of aluminium. Frontiers of Mechanical Engineering, 7(4), 439–444.

Babu, M.N. and Muthukrishnan, N. (2014). Investigation on surface roughness in abrasive water-jet machining by the response surface method. Materials and Manufacturing Processes, 29 (11–12), 1422–1428.

Uthayakumar, M., Khan, M.A., Kumaran, S.T., Slota, A. and Zajac, J. (2016). Machinability of nickel-based Superalloy by abrasive water jet machining. Materials and Manufacturing Processes, 31, 1733–1739.

Ay, M., Çaydaş, U. and Hasçalik, A. (2010). Effect of traverse speed on abrasive waterjet machining of age hardened Inconel 718 nickel based Superalloy. Materials and Manufacturing Processes, 25, 1160–1165.

Goutham, U., Hasu, B.S., Chakraverti, G. and Kanthababu, M. (2016). Experimental investigation of pocket milling on Inconel 825 using abrasive water jet machining. International Journal of Current Engineering and Technology, 6(1), 295–302.

Holmberg, J., Berglund, J., Wretland, A. and Beno, T. (2019). Evaluation of surface integrity after high energy machining with EDM, laser beam machining and abrasive water jet machining

of alloy 718. International Journal of Advanced Manufacturing Technology, 1005(8), 1575–1591. Karakurt, I., Aydin, G. and Aydiner, K. (2010). Effect of the abrasive grain size on the cutting performance of concrete in awj technology. Technology, 133, 145–150.

Sharma, A. and Lalwani, D.I. (2013). Experimental Investigation of Process Parameters Influence on Surface Roughness in Abrasive Water Jet Machining of AISI H13 Die Steel. Proceeding of Conference on Intelligent Robotic Automation and Manufacturing (pp. 333–341). Deepak, D., Anjaiah, D. and Shetty, S. (2017). Optimization of Process Parameters in Abrasive Water Jet Drilling of D2 Steel to Produce Minimum Surface Roughness Using Taguchi Approach. 6th International Conference on Electronics, Computer and Manufacturing Engineering (pp. 236–239).

Wang, J., Kuriyagawa, T. and Huang, C.Z. (2003). An experimental study to enhance the cutting performance in abrasive waterjet machining. Machining Science and Technology, 72, 191–207.

Shipway, P.H., Fowler, G. and Pashby, I.R. (2005). Characteristics of the surface of a titanium alloy following milling with abrasive waterjets. Wear, 258(1–4), 123–132.

Kumar, K.R., Sreebalaji, V.S. and Pridhar, T. (2018). Characterization and optimization of Abrasive Water Jet Machining parameters of aluminium/tungsten carbide composites. Measurement, 117, 57–66.

Srivastava, A.K., Nag, A., Dixit, A.R., Scucka, J., Hloch, S., Klichová, D., Hlaváček, P. and Tiwari, S. (2019). Hardness measurement of surfaces on hybrid metal matrix composite created by turning using an abrasive water jet and WED. Measurement, 131, 628–639.

Wang, S., Hu, D., Yang, F., Tang, C. and Lin, P. (2021). Exploring cutting front profile in abrasive water jet machining of aluminum alloys. International Journal of Advanced Manufacturing Technology, 112, 845–851.

Pradhan, S., Das, S.R., Nanda, B.K., Jena, P.C. and Dhupal, D. (2020). Experimental investigation on machining of hardstone quartz with modified AJM using hot silicon carbide abrasives. Journal of the Brazilian Society of Mechanical Sciences and Engineering, 42(11), 1–22.

Sabarinathan, P., Annamalai, V.E. and Rajkumar, K. (2020). Sustainable application of grinding wheel waste as abrasive for abrasive water jet machining process. Journal of Cleaner Production, 261, 121225.

Saravanan, K., Sudeshkumar, M.P., Maridurai, T., Suyamburajan, V. and Jayaseelan, V. (2021). Optimization of SiC Abrasive Parameters on Machining of Ti-6AI-4V Alloy in AJM Using Taguchi-Grey Relational Method. *Silicon*, 1–8.

Paul, T. and Mandal, I. (2020). Development and Parametric Optimization of Abrasive Jet Machining Setup. Materials Today: Proceedings, 22, 2306–2315.

Chaitanya, A.K., Babu, D.K. and Kumar, K.G. (2020). Experimental study on surface roughness by using abrasive jet machine. Materials Today: Proceedings, 23, 453–457.

Kang, C., Liang, F., Shen, G., Wu, D. and Fang, F. (2021). Study of micro-dimples fabricated on alumina-based ceramics using micro-abrasive jet machining. Journal of Materials Processing Technology, 297, 117181.

Tomy, A. and Hiremath, S.S. (2021). Machining and characterization of multidirectional hybrid silica glass fiber reinforced composite laminates using abrasive jet machining. Silicon, 13, 1151–1164.

Artificial Neural Networks for Machining

Eberbach, E., Goldin, D. and Wegner, P. (2004). Turing's ideas and models of computation. In Alan Turing: Life and Legacy of a Great Thinker. Berlin: Springer. ISBN 978-3-642-05744-1. Naresh, C., Bose, P.S.C. and Rao, C.S.P. (2020). Artificial neural networks and adaptiand neuro-fuzzy models for predicting WEDM machining responses of Nitinol alloy: comparatiand study. SN Applied Sciences, 2, 314.

McCulloch, W.S. and Pitts, W.A. (1943). A logical calculus of the ideas immanent in nervous activity. Buttetin of Mathematics and Biophysics, 5, 115–133.

McClelland, J.L. and Rumelhart, D.E. (1986). Parallel Distributed Processing, Explorations in the Microstructure of Cognition. Psychological and biological models. Cambridge: MIT Press.

Basheer, I.A. and Hajmeer, M. (2000). Artificial neural networks: fundamentals, computing, design, and application. Journal of Microbiological Methods, 43, 3–31.

Azimi, P. and Soofi, P. (2017). An ANN-based optimization model for facility layout problem using simulation technique. Scientia Iranica E, 24(1), 364–377.

Efendigil, T., Önüt, S. and Kahraman, C. (2009). A decision support system for demand forecasting with artificial neural networks and neuro-fuzzy models: A comparatiand analysis. Expert Systems with Applications, 36, 6697–6707.

Benardos, P.G. and Vosniakos, G.C. (2003). Predicting surface roughness in machining: A review. International Journal of Machine Tools and Manufacture, 43(8), 833–844.

Beltratti, A., Margarita, S. and Terna, P. (1996). Neural Networks for Economic and Financial Modelling. London: International Thomson Computer Press.

Davim, J.P., Gaitonde, V.N. and Karmik, S.R. (2008). Investigations into the effect of cutting conditions on surface roughness in turning of free machining steel by ANN models. Journal of Material Processing Technology, 205(1–3), 16–23.

Singh, A.K., Panda, S.S., Pal, S.K. and Chakraborty, D. (2006). Prediction drill wear using an artificial neural network. International Journal of Advanced Manufacturing Technology, 28, 456–462.

Wang, T., Gao, H. and Qiu, J. (2016). A combined adaptiand neural network and nonlinear model predictiand control for multirate networked industrial process control. IEEE Transactions on Neural Networks and Learning Systems, 27(2), 416–425.

Mitrea, C.A., Lee, C.K.M. and Wu, Z. (2009). A Comparison between Neural Networks and Traditional Forecasting Methods: A Case Study. International Journal of Engineering Business Management, 1, 11.

Janikova, D. and Bezak, P. (2016). Prediction of production line performance using neural Networks. In: 3rd International Conference on Artificial Intelligence and Pattern Recognition (AIPR) (pp. 1–5).

Ding, L. and Matthews, J. (2009). A contemporary study into the application of neural network techniques employed to automate CAD/CAM integration for die manufacture. Computers and Industrial Engineering, 57(4), 1457–1471.

Fast, M. and Palme, T. (2010). Application of artificial neural networks to the condition monitoring and diagnosis of a combined heat and power plant. Energy, 35(2), 1114–1120. Mukherjee, I. and Ray, P.K. (2006). A review of optimization techniques in metal cutting processes. Computers & Industrial Engineering, 50, 15–34.

Jawahir, I.S., Balaji, A.K., Rouch, K.E. and Baker, J.R. (2003). Towards integration of hybrid models for optimized machining performance in intelligent manufacturing systems. Journal of Materials Processing Technology, 139 (1–3), 488–498.

Zain, A.M., Haron, H. and Sharif, S. (2010). Prediction of surface roughness in the end milling machining using artificial neural network. Expert Systems with Applications, 37(2), 1755–1768. Rashid, M.F.A. and Abdul Lani, M.R. (2010). Surface roughness prediction for CNC milling process using artificial neural network. In: Proceedings of the World Congress on Engineering (WCE '10), London, United Kingdom.

Razfar, M.R., Zinati, R.F. and Haghshenas, M. (2011). Optimum surface roughness prediction in face milling by using neural network and harmony search algorithm. International Journal of Advanced Manufacturing Technology, 52(5–8), 487–495.

Levin, A.U. and Narendra, K.S. (1993). Control of nonlinear dynamical systems using neural networks: Controllability and Stabilization. IEEE Transactions on Neural Networks, 4(2), 192–206.

Mukhopadhyay, A., Barman, T.K., Sahoo, P. and Davim, J.P. (2019). Modeling and optimization of fractal dimension in wire electrical discharge machining of EN 31 steel using the ANN-GA approach. Materials, 12(3), 454.

Khatir, S., Tiachacht, S., Thanh, C.L., Bui, T.Q. and Wahab, M.A. (2019). Damage assessment in composite laminates using ANN-PSO-IGA and Cornwell indicator. Composite Structures, 230, 111509.

Joshi, S.N. and Pande, S.S. (2011). Intelligent process modeling and optimization of die-sinking electric discharge machining. Applied soft computing, 11(2), 2743–2755.

Hakimpoor, H., Arshad, K.A.B., Tat, H.H., Khani, N. and Rahmandoust, M. (2011). Artificial Neural Networks' Applications in Management. World Applied Sciences Journal, 14(7), 1008–1019.

Kosko, B. (1994). Neural Networks and Fuzzy Systems. Prentice-Hall of India. New Delhi: MIT Press.

Yusoff, Y., Zain, M.A., Sharif, S., Sallehuddin R. and Ngadiman, M.S. (2016). Potential ANN prediction model for multiperformances WEDM on Inconel 718. Neural Computing and Applications, 30, 2113–2127.

Jain, S.P., Ravindra, H.V., Ugrasen, G., Prakash, G.V.N. and Rammohan, Y.S. (2017). Study of Surface Roughness and AE Signals while Machining Titanium Grade-2 Material using ANN in WEDM. Materials Today: Proceedings, 4(9), 9557–9560.

Agrawal, D.P. and Kamble D.N. (2017). GRA and ANN Integrated Approach for Photochemical Machining of Al/SiC Composite. Materials Today: Proceedings, 4(8), 7177–7188.

Abbas, A. T., Pimenov, D. Y., Erdakov, I. N., Taha, M. A., Soliman, M. S. and El Rayes, M. M. (2018). ANN Surface Roughness Optimization of AZ61 Magnesium Alloy Finish Turning: Minimum Machining Times at Prime Machining Costs. Materials, 11(5), 808.

Badiger, P.V., Desai, V., Ramesh, M.R., Prajwala B.K. and Raveendra K. (2019). Cutting Forces, Surface Roughness and Tool Wear Quality Assessment Using ANN and PSO Approach During Machining of MDN431 with TiN/AIN-Coated Cutting Tool. Arabian Journal for Science and Engineering, 44, 7465–7477.

Zerti, A., Yallese M. A., Zerti O., Nouioua M. and Khettabi R. (2019). Prediction of machining performance using RSM and ANN models in hard turning of martensitic stainless steel AISI 420. SAGE Journals, 233(13), 4439–4462.

Sada, S.O. and Ikpeseni, S.C. (2021). Evaluation of ANN and ANFIS modeling ability in the prediction of AISI 1050 steel machining performance. Heliyon, 7(2), e06136.

Bagga, P. J., Makhesana, M.A., Patel, H.D. and Patel, K.M. (2021). Indirect method of tool wear measurement and prediction using ANN network in machining process. Materials Today: Proceedings, 44(1), 1549–1554.

Xu, L., Huang, C., Niu, J., Li, C., Wang, J., Liu, H. and Wang, X. (2021). An improved casebased reasoning method and its application to predict machining performance. Soft Computing, 25, 5683–5697.

Paturi, U.M.R., Cheruku, S., Pasunuri, V.P.K. and Salike, S. (2021). Modeling of tool wear in machining of AISI 52100 steel using artificial neural networks. Materials Today: Proceedings, 38(5), 2358–2365.

Quarto, M., D'Urso, G., Giardini, C., Maccarini, G. and Carminati, M. (2021). A Comparison between Finite Element Model (FEM) Simulation and an Integrated Artificial Neural Network (ANN)-Particle Swarm Optimization (PSO) Approach to Forecast Performances of Micro Electro Discharge Machining (Micro-EDM) Drilling. Micromachines, 12(6), 667.

Trinath, K., Aepuru, R., Biswas, A., Viswanathan, M.R. and Manu, R. (2021). Study of self lubrication property of Al/SiC/Graphite hybrid composite during Machining by using artificial neural networks (ANN). Materials Today: Proceedings, 44, 3881–3887.

Lalwani, V., Sharma, P., Pruncu, C.I. and Unune, D.R. (2020). Response surface methodology and artificial neural network-based models for predicting performance of wire electrical discharge machining of inconel 718 alloy. Journal of Manufacturing and Materials Processing, 4(2), 44.

Kaviarasan, V., Elango, S., Gnanamuthu, E.M.A. and Durairaj, R. (2021). Smart Machining of Titanium Alloy Using ANN Encompassed Prediction Model and GA Optimization. In Futuristic Trends in Intelligent Manufacturing. Cham: Springer.

Lin, Y.C., Wu, K.D., Shih, W.C., Hsu, P.K. and Hung, J.P. (2020). Prediction of surface roughness based on cutting parameters and machining vibration in end milling using regression method and artificial neural network. Applied Sciences, 10(11), 3941.

Nagaraj, Y., Jagannatha, N., Sathisha, N. and Niranjana, S.J. (2020). Prediction of material removal rate and surface roughness in hot air assisted hybrid machining on soda-lime-silica glass using regression analysis and artificial neural network. *Silicon*, 13 (11), 4163–4175.

Machining and Vibration Behavior of Ti-TiB Composites Processed through Powder Metallurgy Techniques

Klocke, F., Mohammadnejad, M., Holsten, M., Ehle, L., Zeis, M., Klink, A. (2018). A Comparative study of polarity-related effects in single discharge EDM of Titanium and Iron alloys, Procedia CIRP, 68, 52–57.

Kou, Z., Han, F. (2018). On sustainable manufacturing titanium alloy by high-speed EDM milling with moving electric arcs while using water-based dielectric, Journal of Cleaner Production, 189, 78–87.

Bhaumik, M., Maity, K. (2018). Effect of different tool materials during EDM performance of titanium grade 6 alloy, Engineering Science and Technology, an International Journal, 21 (3), 507–516.

Pecas, P., Henriques, E. (2003). Influence of silicon powder-mixed dielectric on conventional electrical discharge machining, Internatioanl Journal of Machine Tools Manufacture, 43, 1465–1471.

Lee, S.H., Li, X.P., (2001). Study of the effect of machining parameters on the machining characteristics in electrical discharge machining of tungsten carbide, Journal of Materials Processing and Technology, 115, 344–358.

Boujelbene, M., Bayraktar, E., Tebni, W., Salemn, S.B. (2009). Influence of machining parameters on the surface integrity in electrical discharge machining, Archives of Materials Science and Engineering, 37, 110–116.

Selvakumar, M., Ramkumar, T., Mohanraj, M., Chandramohan, P., Narayanasamy, P. (2020). Experimental investigations of reciprocating wear behavior of metal matrix (Ti/TiB) composites, Archives of Civil and Mechanical Engineering, 20, 1–9.

Ramkumar, T., Selvakumar, M., Mohanraj, M., Chandrasekar, P. (2019). Experimental investigation and analysis of drilling parameters of metal matrix (Ti/TiB) composites, Journal of the Brazilian Society of Mechanical Sciences and Engineering, 41, 1–12.

Torres, A., Luis, C.J., Puertas, I. (2017). EDM machinability and surface roughness analysis of TiB2 using copper electrodes, Journal of Alloys and Compounds, 690, 337–347.

Sales, W.F., Oliveira, A.R.F., Raslan, A.A. (2016). Titanium perovskite (CaTiO3) formation in Ti6Al4V alloy using the electrical discharge machining process for biomedical applications, Surface and Coatings Technology, 307, 1011–1015.

Rao, P.S., Ramji, K., Satyanarayana, B. (2016). Effect of wire EDM conditions on generation of residual stresses in machining of aluminum 2014 T6 alloy, Alexandria Engineering Journal, 55, 1077–1084.

Li, L., Feng, L., Bai, X., Li, Z.Y. (2016). Surface characteristics of Ti–6Al–4V alloy by EDM with Cu–SiC composite electrode, Applied Surface Science, 388, 546–550.

Urso, G.D., Ravasio, C. (2017). Material-Technology Index to evaluate micro-EDM drilling process, Journal of Manufacturing Processes, 26, 13–21.

Hou, J., Zhao, Z., Fu, Y., Qian, N. (2021). Machining stability enhancement in multi-axis milling of titanium hollow blade by introducing multiple damping and rigid supporters, Journal of Manufacturing Processes, 64, 198–208.

Sajadifar, S.V., Atli, C., Yapici, G.G. (2019), Effect of severe plastic deformation on the damping behavior of titanium, Materials Letters, 244, 100–103.

Numerical Analysis of Machining Forces and Shear Angle during Dry Hard Turning

Bouacha, K., Yallese, M.A., Khamel, S., Belhadi, S. (2014). Analysis and optimization of hard turning operation using cubic boron nitride tool, International Journal of Refractory Metals and Hard Materials 45, 160–178.

Gajrani, K.K., Suvin, P.S., Kailas, S.V., Rajurkar, K.P., Sankar, M.R. (2021). Machining of hard materials using textured tool with minimum quantity nano-green cutting fluid, CIRP Journal of Manufacturing Science and Technology 35, 410–421.

Bartarya, G., Choudhury, S.K. (2012). Effect of Cutting Parameters on Cutting Force and Surface Roughness during Finish Hard Turning AISI52100 Grade Steel, Procedia CIRP 1,

651–656.

Alok, A., Das, M. (2018). Cost-effective way of hard turning with newly developed HSN2-coated tool, Material and Manufacturing Process 33, 1003–1010.

Zhang, Q., Zhang, S., Li, J. (2017). Three dimensional finite element simulation of cut-ting forces and cutting temperature in hard milling of AISI H13 steel, Procedia Manufacturing 10, 37–47.

Gajrani, K.K., Divse, V., Joshi, S.S. (2021). Burr reduction in drilling titanium using drills with peripheral slits, Transactions of the Indian Institute of Metals 74(5), 1155–1172.

Akbar, F., Mativenga, P.T., Sheikh, M.A. (2010). An experimental and coupled thermomechanical finite element study of heat partition effects in machining, The International Journal of Advanced Manufacturing Technology 46(5–8), 491–507.

Zerilli, F.J. (2004). Dislocation mechanics-based constitutive equations, Metallurgical and Materials Transactions A 35(9), 2547–2555.

Johnson, G.R., Cook, W.H. (1983). A Constitutive Model and Data for Metals Subjected to Large Strains, High Strain Rates, and High Temperatures Proceedings 7th International Symposium on Ballistics, *The Hague*, 19–21, April 1983, 541–547. - References - Scientific Research Publish, (n.d.).

Ucun, İ., Aslantas, K. (2010). Numerical simulation of orthogonal machining process using multilayer and single-layer coated tools, The International Journal of Advanced Manufacturing Technology 46(5–8), 491–507.

Machining Performance Evaluation of Titanium Biomaterial, Ti6Al4V in CNC cylindrical turning Using CBN Insert

Degarmo, E. P., Black, J. T. (2003). Materials and Processes in Manufacturing (9th ed.), Wiley, ISBN:0-471-65653-4.

Schneider, G. (2002). Cutting Tool Applications, Tooling and Production, Chapter 3, Nelson Publishing LLC.

Bhaumik, S. K., Divakar, C., Singh, A. K. (1995). Machining Ti-6AI-4V alloy with a WBN-CBN composite tool. Materials & Design, 16, 221–226.

Huang, Y., Chou, Y. K., Steven Y. L. (2007). CBN tool wear in hard turning: A survey on research progresses. The International Journal of Advanced Manufacturing Technology, 35, 443–453.

Zareena, A. R. (2002). High-speed machining of titanium alloys. *Master Thesis, National University of Singapore.*

Hirosaki, K., Kazuhiro S., Hideharu K. (2004). High speed machining of bio-titanium alloy with a binder-less PCBN tool. JSME International Journal Series C Mechanical Systems, Machine Elements and Manufacturing, 47, 14–20.

Wang, Z. G., Wong, Y. S., Rahman M. (2005). High-speed milling of titanium alloys using binderless CBN tools. International Journal of Machine Tools and Manufacture, 45, 105–114. Wang, Z. G. (2005). Modelling of Cutting Forces during Machining of Ti-6Al-4V with Different Coolant Strategies. In 8th CIRP International Workshop on Modelling in Machining Operations, Chemnitz, Germany.

Ezugwu, E. O., Da Silva R. B., Machado A. R. (2005). The effect of argon-enriched environment in high-speed machining of titanium alloy. Tribology Transactions, 48, 18–23.

Ozel, T., Sima M., Srivastava, A. K., Kaftanoglu, B. (2010). Investigations on the effects of multilayered coated inserts in machining Ti–6Al–4V alloy with experiments and finite element simulations. CIRP Annals, 59, 77–82.

Bhaumik, S. K., Divakar, C., Singh, A. K. (1995). Machining Ti6Al4V alloy with a WBN-CBN composite tool. Materials & Design, 16, 221–226.

Burhanuddin, Y., Ghani J. A., Ariffin, A. K. (2008). The effects of CBN cutting tool grades on the tool life and wear mechanism when dry turning of titanium alloy. *Asian Int. Journal of Science and Technology*, 105–110.

Zoya, Z. A., Krishnamurthy, R. (2000). The performance of CBN tools in the machining of titanium alloys. Journal of Materials Processing Technology, 100, 80–86.

Liang, X., Wang, B. (2019). Investigation of surface topography and its deterioration resulting from tool wear evolution when dry turning of Titanium alloy Ti-6AI-4V. Tribology International, 135, 2019.

Li, A., Zhao, J., Hou, G. (2017). Effect of cutting speed on chip formation and wear mechanisms of coated carbide tools when ultra-high-speed face milling titanium alloy Ti-6AI-4V. Advances in Mechanical Engineering, 9(7), 1–13.

Polini, W., Turchetta, S. (2016). Cutting force, tool life and surface integrity in milling of titanium alloy Ti-6AI-4V with coated carbide tools. Proceedings of the Institution of Mechanical Engineers, Part B, 230, 694–700.

Guo, K., Yang, B., Sivalingam, V. (2018). Investigation on the tool wear model and equivalent tool life in end milling titanium alloy TI6AL4V. ASME 13th International Manufacturing Science and Engineering Conference MSEC, Vol. 4, 1–8.

Jaffery, S. H. I., Mativenga, P. T. (2012). Wear mechanisms analysis for turning Ti-6Al-4V-towards the development of suitable tool coatings. The International Journal of Advanced Manufacturing Technology, 58, 479–493.

Dorlin, T., Fromentin, G., Costes, J. P. (2016). Generalised cutting force model including contact radius effect for turning operations on Ti6Al4V titanium alloy. The International Journal of Advanced Manufacturing Technology, 86, 3297–3313.

Celik, Y. H., Kilickap, E., Guney, M. (2017). Investigation of cutting parameters affecting on tool wear and surface roughness in dry turning of Ti-6Al-4V using CVD and PVD coated tools. The Journal of the Brazilian Society of Mechanical Sciences and Engineering, 39, 2085–2093.

Kamaruddin, A. M. N. A., Hosokawa, A., Ueda, T., Furumoto, T. (2016). Cutting characteristics of binderless diamond tools in high-speed turning of Ti-6AI-4V: Availability of single-crystal and nano-polycrystalline diamond. International Journal of Automotive Technology, 10, 411–419.

Wang, Z., Rahman, M., Wong, Y. S. (2005). Tool wear characteristics of binderless CBN tools used in high-speed milling of titanium alloys. Wear 258, 752–758.

Bhaumik, S. K., Divakar, C. (1995). Machining Ti-6Al-4V alloy with a WBN-CBN composite tool. Materials & Design 16, 221–226.

Shaw, M. C. (1997). Metal Cutting Principles (Oxford Series on Advanced Manufacturing), Oxford University Press.

Christian, U. (2008). Acoustic Emission Testing, Springer.

Pawade, R. S., Joshi, S. S. (2012). Analysis of Acoustic Emission Signals and Surface Integrity in High-speed Turning of Inconel 718. Journal of Engineering Manufacture, Institution of Engineers, Part B, 126(1), 3–27.

Industrial Internet of Things in Manufacturing

Botta, A., De Donato, W., Persico, V., Pescapé, A. (2016). Integration of cloud computing and internet of things: A survey. Future Gener. Comput. Syst. 56:684–700.

Gubbi, J., Buyya, R., Marusic, S., Palaniswami, M. (2013). Internet of things (IoT): A vision, architectural elements, and future directions. Future Gener. Comput. Syst. 29(7): 1645–1660. Wang, C., Bi, Z., Da Xu, L. (2014). IoT and cloud computing in automation of assembly modeling systems. IEEE Trans. Ind. Inf. 10(2):1426–1434.

Colombo, A.W., Karnouskos, S., Kaynak, O., Shi, Y., Yin, S. (2017). Industrial cyberphysical systems: A backbone of the fourth industrial revolution. IEEE Ind. Electron. Mag. 11(1): 6–16. Da Xu, L., He, W., Li, S. (2014). Internet of things in industries: A survey. IEEE Trans. Ind. Inf. 10(4): 2233–2243.

He, W., Da Xu, L. (2014). Integration of distributed enterprise applications: A survey. IEEE Trans. Ind. Inf. 10(1): 35–42.

Aazam, M., Zeadally, S., Harras, K.A. (2018). Deploying fog computing in industrial internet of things and industry 4.0. IEEE Trans. Ind. Inf. 14(10):4674–4682. doi:10.1109/TII.2018.2855198. Al-Gumaei, K., Schuba, K., Friesen, A., Heymann, S., Pieper, C., Pethig, F., et al. (2018). A survey of internet of things and big data integrated solutions for industrie 4.0. In: 2018 IEEE 23rd International Conference on Emerging Technologies and Factory Automation (ETFA), 1. IEEE; 2018, pp. 1417–1424.
Long, N.B., Tran-Dang, H., Kim, D. (2018). Energy-aware real-time routing for large-scale industrial internet of things. IEEE Internet Things J. 5(3):2190–2199. doi:10.1109/JIOT.2018.2827050.

Mumtaz, S., Alsohaily, A., Pang, Z., Rayes, A., Tsang, K.F., Rodriguez, J. (2017). Massive internet of things for industrial applications: addressing wireless IIoT connec- tivity challenges and ecosystem fragmentation. IEEE Ind. Electron. Mag. 11(1):28–33.

Perera, C., Liu, C.H., Jayawardena, S. (2015). The emerging internet of things marketplace from an industrial perspective: a survey. IEEE Trans. Emerg. Top Comput. 3(4):585–598. doi:10.1109/TETC.2015.2390034

Xu, H., Yu, W., Griffith, D., Golmie, N. (2018). A survey on industrial internet of things: a cyberphysical systems perspective. IEEE Access 6:78238–78259. doi:10.1109/ACCESS.2018.2884906.

Zhu, C., Rodrigues, J.J.P.C., Leung, V.C.M., Shu, L., Yang, L.T. (2018). Trust-based communication for the industrial internet of things. IEEE Commun. Mag. 56(2):16–22. doi:10.1109/MCOM.2018.1700592.

Khan, W.Z., Rehman, M.H., Zangoti, H.M., Afzal, M.K., Armi, N., Salah, K. (2020). Industrial internet of things: Recent advances, enabling technologies and open challenges. Comput. Elect.Eng. 81:106–522.

Lin, S.W., Miller, B., Durand, J., Bleakley, G., Chigani, A., Martin, R., Murphy, B., Crawford, M. (2017). The industrial internet of things volume G1: Reference architecture. Ind. Internet Consort. 1:10–46.

Campobello, G, Castano, M, Fucile, A, Segreto, A. (2017).Weva: A complete solution for industrial internet of things. In: International Conference on Ad-hoc Networks and Wireless. Springer, pp. 231–238

Lee, C.K.M., Zhang, S.Z., Ng, K.K.H. (2017). Development of an industrial internet of things suite for smart factory towards re-industrialization. Adv. Manuf. 5(4):335–343. doi:10.1007/s40436-017-0197-2.

Martinez, B., Vilajosana, X., Kim, I., Zhou, J., Tuset-Peiró, P., Xhafa, A., Poissonnier, D., Lu, X. (2017). I3mote: an open development platform for the intelligent industrial internet. Sensors 17(5):986. doi:10.3390/s17050986.

Tao, F., Cheng, J., Qi, Q. (2018). IIHub: An industrial internet-of-things hub toward smart manufacturing based on cyber-physical system. IEEE Trans. Ind. Inf. 14(5):2271–2280. doi:10.1109/tii.2017.2759178

Meng, Z., Wu, Z., Muvianto, C., Gray, J. (2017). A data-oriented M2M messaging mechanism for industrial IoT applications. IEEE Internet Things J. 4(1):236–246.

Katsikeas, S., Fysarakis, K., Miaoudakis, A., Van Bemten, A., Askoxylakis, I., Papaefstathiou, I., Plemenos, A. (2017).Lightweight & secure industrial IoT communications via the MQ telemetry transport protocol. In: Computers and communications (ISCC), 2017 IEEE symposium on. IEEE,pp. 1193–1200.

Kiran, M., Rajalakshmi, P. (2018). Performance analysis of CSMA/CA and PCA for time critical industrial IoT applications. IEEE Trans. Ind. Inf. 14(5):2281–2293.

Ferrari, P., Sisinni, E., Brandão, D., Rocha, M. (2017). Evaluation of communication latency in industrial IoT applications. In: Measurement and networking (M&N), 2017 IEEE international workshop on. IEEE; 2017,pp. 1–6.

Raptis, T.P., Passarella, A. (2017). A distributed data management scheme for industrial IoT environments. In: Wireless and mobile computing, networking and communications (WiMob), IEEE,pp. 196–203

Raptis, T.P., Passarella, A., Conti, M. (2018). Maximizing industrial IoT network lifetime under latency constraints through edge data distribution. In: 1st IEEE international conference on industrial cyber-physical systems, (ICPS) (May 2018). Available at http://cnd.iit.cnr.it/traptis/2018-raptis-icps.pdf.

Rao, S., Shorey, R. (2017). Efficient device-to-device association and data aggregation in industrial IoT systems. In: Communication systems and networks (COM-SNETS), 2017 9th international conference on. IEEE, pp. 314–321.

ur Rehman, M.H., Yaqoob, I., Salah, K., Imran, M., Jayaraman, P.P., Perera, C. (2019). The role of big data analytics in industrial internet of things. Fut. Gener. Comput. Syst.; 99:247–259. doi:10.1016/j.future.2019.04.020

Vyatkin, V. (2013). Software Engineering in Industrial Automation: State-of-the-Art Review. IEEE Trans. Ind. Inf. 9(3): 1234–1249.

Calvo, I., Pérez, F., Etxeberria, I., Morán, G. (2010).Control communications with DDS using IEC61499 service interface function blocks.In: Emerging Technologies and Factory Automation (ETFA), 2010 IEEE Conference on, IEEE, pp. 1–4.

Dubinin, V.N., Vyatkin, V. (2012). Semantics-robust design patterns for IEC 61499. IEEE Trans. Ind. Inf. 8(2): 279–290.

Hanisch, H.M. (2004).Closed-loop modeling and related problems of embedded control systems in engineering.In: International Workshop on Abstract State Machines, Springer, pp. 6–19. Jammes, F., Smit, H. (2005). Service-oriented paradigms in industrial automation. IEEE Trans. Ind. Inf. 1(1): 62–70.

Theiss, S., Vasyutynskyy, V., Kabitzsch, K. (2009). Software agents in industry: A customized framework in theory and praxis. IEEE Trans.Ind. Inf. 5(1):147–156.

Thramboulidis, K. (2005). Model-integrated mechatronics-toward a new paradigm in the development of manufacturing systems. IEEE Trans. Ind. Inf. 1(1): 54–61.

Obermeier, M., Braun, S., Vogel-Heuser, B.(2015). A model-driven approach on objectoriented PLC programming for manufacturing systems with regard to usability. IEEE Trans. Ind. Inf. 11(3): 790–800.

Vogel-Heuser, B., Schütz, D., Frank, T., Legat, C.(2014). Model-driven engineering of manufacturing automation software projects–a SysML-based approach. Mechatronics 24(7): 883–897.

Vinoski, S. (1997). CORBA: Integrating diverse applications within distributed heterogeneous environments. IEEE Commun. Mag. 35: 46–55.

Sessions, R.(1997)COM and DCOM: Microsoft's Vision for Distributed Objects. New York, NY: Wiley.

Crnkovic, I.,Larsson, M. (2000). A case study: Demands on component- based development. Proc. Int. Conf. Software Eng.23–31. https://www.computer.org/csdl/proceedingsarticle/icse/2000/00870393/12OmNgzu6TR

Szer-Ming, L., Harrison, R., West, A.A. (2004). A component-based distributed control system for assembly automation. In Proc. 2nd IEEEInt. Conf. Ind. Inf., pp. 33–38.

Strasser, T.et al. (2008). Structuring of large scale distributed control programs with IEC 61499 subapplications and a hierarchical plant structure model.In:Proc. IEEE Int. Conf. Emerging Technol. Factory Autom., Sep. 15–18, pp. 934–941.

Wooldridge, M. (2002). An Introduction to Multi-Agent Systems. New York, NY: Wiley. Herrera, V.V., Bepperling, A., Lobov, A., Smit, H., Colombo, A.W., Lastra, J. (2008). Integration of multi-agent systems and service-oriented architecture for industrial automation. *Proc. 6th IEEE Int. Conf.Ind. Inf.*, pp. 768–773

Erl, T. (2005).Service-Oriented Architecture: Concepts, Technology, and Design. Upper Saddle River, NJ: Prentice-Hall PTR.

Cannata, A., Gerosa, M., Taisch, M. (2008). SOCRADES: A framework for developing intelligent systems in manufacturing. Proc. IEEEInt. Conf. Ind. Eng. Eng. Manag., pp. 1904–1908.

Muthukumar, N., Srinivasan, S., Ramkumar, K., Pal, D., Vain, J., Ramaswamy, S. (2019). A model-based approach for design and verification of Industrial Internet of Things. Fut. Gene. Comp. Syst. 95: 354–363

Reineke, J., Tripakis, S. (2014).Basic problems in multi-view modeling.In: International Conference on Tools and Algorithms for the Construction and Analysis of Systems, Springer, pp. 217–232.

Balasubramaniyan, S., Srinivasan, S., Buonopane, F., Subathra, B., Vain, J., Ramaswamy, S. (2016). Design and verification of cyber-physical systems using truetime, evolutionary optimization and UPPAAL. Microprocess. Microsyst. 42: 37–48.

Srinivasan, S., Buonopane, F., Vain, J., Ramaswamy, S. (2015). Model checking response times in networked automation systems using jitter bounds. Comput. Ind. 74: 186–200.

Improvement in Forming Characteristics Resulted in Incremental Sheet Forming

S. Ai and H. Long, "A review on material fracture mechanism in incremental sheet forming," Int. J. Adv. Manuf. Technol., vol. 104, no. 1–4, pp. 33–61, 2019. doi:10.1007/s00170-019-03682-6. T. J. Kim and D. Y. Yang, "Improvement of formability for the incremental sheet metal forming process," Int. J. Mech. Sci., vol. 42, no. 7, pp. 1271–1286, 2000. doi:10.1016/S0020-7403(99)00047-8.

L. Filice, L. Fratini, and F. Micari, "Analysis of material formability in incremental forming," CIRP Ann. - Manuf. Technol., vol. 51, no. 1, pp. 199–202, 2002. doi:10.1016/S0007-8506(07)61499-1.

J. Jeswiet and D. Young, "Forming limit diagrams for single-point incremental forming of aluminium sheet," Proc. Inst. Mech. Eng. Part B J. Eng. Manuf., vol. 219, no. 4, pp. 359–364, 2005. doi:10.1243/095440505X32210.

G. Ambrogio, V. Cozza, L. Filice, and F. Micari, "An analytical model for improving precision in single point incremental forming," J. Mater. Process. Technol., vol. 191, no. 1–3, pp. 92–95, 2007. doi:10.1016/j.jmatprotec.2007.03.079.

A. Taherkhani, A. Basti, N. Nariman-Zadeh, and A. Jamali, "Achieving maximum dimensional accuracy and surface quality at the shortest possible time in single-point incremental forming via multi-objective optimization," Proc. Inst. Mech. Eng. Part B J. Eng. Manuf., vol. 233, no. 3, pp. 900–913, 2019. doi:10.1177/0954405418755822.

A. Attanasio, E. Ceretti, A. Fiorentino, L. Mazzoni, and C. Giardini, "Experimental tests to study feasibility and formability in incremental forming process," Key Eng. Mater., vol. 410–411, 2009. doi:10.4028/www.scientific.net/KEM.410-411.391.

Y. H. Kim and J. J. Park, "Effect of process parameters on formability in incremental forming of sheet metal," J. Mater. Process. Technol., vol. 130, no. 131, pp. 42–46, 2002. doi:10.1016/S0924-0136(02)00788-4.

G. Hussain, L. Gao, N. Hayat, and L. Qijian, "The effect of variation in the curvature of part on the formability in incremental forming: An experimental investigation," Int. J. Mach. Tools Manuf., vol. 47, no. 14, pp. 2177–2181, 2007. doi:10.1016/j.ijmachtools.2007.05.001.

G. Hussain, L. Gao, N. Hayat, and N. U. Dar, "The formability of annealed and pre-aged AA-2024 sheets in single-point incremental forming," Int. J. Adv. Manuf. Technol., vol. 46, no. 5–8, pp. 543–549, 2010. doi:10.1007/s00170-009-2120-x.

J. Jeswiet, F. Micari, G. Hirt, A. Bramley, J. Duflou, and J. Allwood, "Asymmetric single point incremental forming of sheet metal," CIRP Ann. - Manuf. Technol., vol. 54, no. 2, pp. 88–114, 2005. doi:10.1016/s0007-8506(07)60021-3.

J. M. Allwood, D. R. Shouler, and A. E. Tekkaya, "The increased forming limits of incremental sheet forming processes," Key Eng. Mater., vol. 344, pp. 621–628, 2007. doi:10.4028/www.scientific.net/kem.344.621.

G. Centeno, A. J. Martínez-Donaire, I. Bagudanch, D. Morales-Palma, M. L. Garcia-Romeu, and C. Vallellano, "Revisiting formability and failure of AISI304 sheets in SPIF: Experimental approach and numerical validation," Metals (Basel), vol. 7, no. 12, pp. 1–14, 2017. doi:10.3390/met7120531.

K. Jawale, J. F. Duarte, A. Reis, and M. B. Silva, "Characterizing fracture forming limit and shear fracture forming limit for sheet metals," J. Mater. Process. Technol., vol. 255, no. 2010, pp. 886–897, 2018. doi:10.1016/j.jmatprotec.2018.01.035.

S. Gatea, D. Xu, H. Ou, and G. McCartney, "Evaluation of formability and fracture of pure titanium in incremental sheet forming," Int. J. Adv. Manuf. Technol., vol. 95, no. 1–4, pp. 625–641, 2018. doi:10.1007/s00170-017-1195-z.

B. T. Araghi, G. L. Manco, M. Bambach, and G. Hirt, "Investigation into a new hybrid forming process: Incremental sheet forming combined with stretch forming," CIRP Ann. - Manuf. Technol., vol. 58, no. 1, pp. 225–228, 2009. doi:10.1016/j.cirp.2009.03.101.

L. Fratini, G. Ambrogio, R. Di Lorenzo, L. Filice, and F. Micari, "Influence of mechanical properties of the sheet material on formability in single point incremental forming applied to manufacturing of biocompatible polymer prostheses forming," CIRP Ann. - Manuf. Technol., vol. 53, no. 1, pp. 207–210, 2004.

M. B. Silva, M. Skjoedr, A. G. Atkins, N. Bay, and P. A. F. Martins, "Single-point incremental forming and formability-failure diagrams," J. Strain Anal. Eng. Des., vol. 43, no. 1, pp. 15–35, 2008. doi:10.1243/03093247JSA340.

W. C. Emmens and A. H. van den Boogaard, "An overview of stabilizing deformation mechanisms in incremental sheet forming," J. Mater. Process. Technol., vol. 209, no. 8, pp. 3688–3695, 2009. doi:10.1016/j.jmatprotec.2008.10.003.

R. Malhotra, L. Xue, T. Belytschko, and J. Cao, "Mechanics of fracture in single point incremental forming," J. Mater. Process. Technol., vol. 212, no. 7, pp. 1573–1590, 2012. doi:10.1016/j.jmatprotec.2012.02.021.

A. J. Martínez-Donaire, M. Borrego, D. Morales-Palma, G. Centeno, and C. Vallellano, "Analysis of the influence of stress triaxiality on formability of hole-flanging by single-stage SPIF," Int. J. Mech. Sci., vol. 151, pp. 76–84, 2019. doi:10.1016/j.ijmecsci.2018.11.006.

G. Hirt, J. Ames, M. Bambach, R. Kopp, and R. Kopp, "Forming strategies and process modelling for CNC incremental sheet forming," CIRP Ann. - Manuf. Technol., vol. 53, no. 1, pp. 203–206, 2004. doi:10.1016/S0007-8506(07)60679-9.

D. Nguyen and Y. Kim, "A Numerical Study on Establishing the Forming Limit Curve and Indicating the Formability of Complex Shape in Incremental Sheet Forming Process," Int. J. Precis. Eng., vol. 14, no. 12, pp. 2087–2093, 2013. doi:10.1007/s12541-013-0283-8.

S. Kurra and S. P. Regalla, "Experimental and numerical studies on formability of extra-deep drawing steel in incremental sheet metal forming," J. Mater. Res. Technol., vol. 3, no. 2, pp. 158–171, 2014. doi:10.1016/j.jmrt.2014.03.009.

M. J. Mirnia and M. Shamsari, "Numerical prediction of failure in single point incremental forming using a phenomenological ductile fracture criterion," J. Mater. Process. Tech., 2017. doi:10.1016/j.jmatprotec.2017.01.029.

L. Ben Said, J. Mars, M. Wali, and F. Dammak, "Numerical prediction of the ductile damage in single point incremental forming process," Int. J. Mech. Sci., vol. 131–132, pp. 546–558, 2017. doi:10.1016/j.ijmecsci.2017.08.026.

S. Deokar, P. K. Jain, and P. Tandon, "Formability assessment in single point incremental sheet forming through finite element analysis," Mater. Today Proc., vol. 5, no. 11, pp. 25430–25439, 2018. doi:10.1016/j.matpr.2018.10.348.

M. Bambach, M. Cannamela, M. Azaouzi, G. Hirt, and J. L. Batoz, "Computer-aided tool path optimization for single point incremental sheet forming," in Advanced Methods in Material Forming: With 264 Figures and 37 Tables, 2007. doi: 10.1007/3-540-69845-0 14.

M. Rauch, J. Y. Hascoet, J. C. Hamann, and Y. Plenel, "Tool path programming optimization for incremental sheet forming applications," CAD Comput. Aided Des., vol. 41, no. 12, pp. 877–885, 2009. doi:10.1016/j.cad.2009.06.006.

Z. Liu, W. J. T. Daniel, Y. Li, S. Liu, and P. A. Meehan, "Multi-pass deformation design for incremental sheet forming: Analytical modeling, finite element analysis and experimental validation," J. Mater. Process. Technol., vol. 214, no. 3, pp. 620–634, 2014. doi:10.1016/j.jmatprotec.2013.11.010.

Z. Liu, Y. Li, and P. A. Meehan, "Tool path strategies and deformation analysis in multi-pass incremental sheet forming process," Int. J. Adv. Manuf. Technol., vol. 75, no. 1–4, pp. 395–409, 2014. doi:10.1007/s00170-014-6143-6.

J. Wang, L. Li, P. Zhou, X. Wang, and S. Sun, "Improving formability of sheet metals in incremental forming by equal diameter spiral tool path," Int. J. Adv. Manuf. Technol., vol. 101, no. 1–4, pp. 225–234, 2019. doi:10.1007/s00170-018-2911-z.

T. J. Grimm and L. Mears, "Investigation of a radial toolpath in single point incremental forming," Procedia Manuf., vol. 48, pp. 215–222, 2020. doi:10.1016/j.promfg.2020.05.040.

J. R. Duflou, B. Callebaut, J. Verbert, and H. De Baerdemaeker, "Laser assisted incremental forming: Formability and accuracy improvement," CIRP Ann. - Manuf. Technol., vol. 56, no. 1, pp. 273–276, 2007. doi:10.1016/j.cirp.2007.05.063.

G. Ambrogio, L. Filice, and F. Gagliardi, "Formability of lightweight alloys by hot incremental sheet forming," Mater. Des., vol. 34, pp. 501–508, 2012. doi:10.1016/j.matdes.2011.08.024. D. Xu, B. Lu, T. Cao, J. Chen, H. Long, and J. Cao, "A comparative study on process potentials for frictional stir- and electric hot-assisted incremental sheet forming," Procedia Eng., vol. 81, no. October, pp. 2324–2329, 2014. doi:10.1016/j.proeng.2014.10.328.

A. Mohammadi, H. Vanhove, A. Van Bael, and J. R. Duflou, "Towards accuracy improvement in single point incremental forming of shallow parts formed under laser assisted conditions," Int. J. Mater. Form., vol. 9, no. 3, pp. 339–351, 2016. doi:10.1007/s12289-014-1203-x.

D. K. Xu et al., "Enhancement of process capabilities in electrically-assisted double sided incremental forming," Mat. Des., vol. 92, pp. 268–280, 2016. doi:10.1016/j.matdes.2015.12.009.

M. Vahdani, M. J. Mirnia, M. Bakhshi-Jooybari, and H. Gorji, "Electric hot incremental sheet forming of Ti-6Al-4V titanium, AA6061 aluminum, and DC01 steel sheets," Int. J. Adv. Manuf. Technol., vol. 103, no. 1–4, pp. 1199–1209, 2019. doi:10.1007/s00170-019-03624-2.

X. Xiao, C. Il Kim, X. D. Lv, T. S. Hwang, and Y. S. Kim, "Formability and forming force in incremental sheet forming of AA7075-T6 at different temperatures," J. Mech. Sci. Technol., vol. 33, no. 8, pp. 3795–3802, 2019. doi:10.1007/s12206-019-0722-2.

S. Zhang, G. H. Tang, W. Wang, and X. Jiang, "Evaluation and optimization on the formability of an AZ31B Mg alloy during warm incremental sheet forming assisted with oil bath heating," Meas. J. Int. Meas. Confed., vol. 157, p. 107673, 2020.

doi:10.1016/j.measurement.2020.107673.

H. Iseki, "Flexible and Incremental Bulging of Sheet Metal using High Speed Water Jet," Nihon Kikai Gakkai Ronbunshu, C Hen/Transactions of the Japan Society of Mechanical Engineers, Part C, vol. 65, no. 635. pp. 2912–2918, 1999. doi:10.1299/kikaic.65.2912.

B. Jurisevic, K. Kuzman, and M. Junkar, "Water jetting technology: An alternative in incremental sheet metal forming," Int. J. Adv. Manuf. Technol., vol. 31, no. 1–2, pp. 18–23, Oct. 2006. doi:10.1007/s00170-005-0176-9.

J. Li, K. He, S. Wei, X. Dang, and R. Du, "Modeling and experimental validation for truncated cone parts forming based on water jet incremental sheet metal forming," Int. J. Adv. Manuf. Technol., vol. 75, no. 9–12, pp. 1691–1699, 2014. doi:10.1007/s00170-014-6222-8.

R. Malhotra, J. Cao, F. Ren, V. Kiridena, Z. Cedric Xia, and N. V. Reddy, "Improvement of geometric accuracy in incremental forming by using a squeezing toolpath strategy with two forming tools," J. Manuf. Sci. Eng., vol. 133, no. 6, pp. 1–9, December 2011. doi:10.1115/1.4005179.

J. Smith, R. Malhotra, W. K. Liu, and J. Cao, "Deformation mechanics in single-point and accumulative double-sided incremental forming," Int. J. Adv. Manuf. Technol., vol. 69, no. 5–8, pp. 1185–1201, 2013. doi:10.1007/s00170-013-5053-3.

S. Ai, R. Dai, and H. Long, "Investigating formability enhancement in double side incremental forming by developing a new test method of tension under cyclic bending and compression," J. Mater. Process. Technol., vol. 275, no. July 2019, p. 116349, 2020. doi:10.1016/j.jmatprotec.2019.116349.

D. Möllensiep, T. Gorlas, P. Kulessa, and B. Kuhlenkötter, "Real-time stiffness compensation and force control of cooperating robots in robot-based double sided incremental sheet forming," *Prod. Eng.*, no. 0123456789, 2021. doi:10.1007/s11740-021-01052-4.

P. Li et al., "Evaluation of forming forces in ultrasonic incremental sheet metal forming," Aerosp. Sci. Technol., vol. 63, no. December 2016, pp. 132–139, 2017. doi:10.1016/j.ast.2016.12.028.
V. C. Do, D. C. Ahn, and Y. S. Kim, "Formability and effect of hole bridge in the single point incremental forming," Int. J. Precis. Eng. Manuf., vol. 18, no. 3, pp. 453–460, 2017. doi:10.1007/s12541-017-0054-z.

F. A. Lora, D. Fritzen, R. Alves de Sousa, and L. Schaffer, "Studying formability limits by combining conventional and incremental sheet forming process," Chin. J. Mech. Eng. (English Ed.), vol. 34, no. 1, 2021. doi:10.1186/s10033-021-00562-7.

Deformation Mechanism of Polymers, Metals, and Their Composites in Dieless Forming Operations

S. Gatea, H. Ou, and G. McCartney, "Review on the influence of process parameters in incremental sheet forming," Int. J. Adv. Manuf. Technol., vol. 87, no. 1–4, pp. 479–499, 2016, doi:10.1007/s00170-016-8426-6.

W. C. Emmens, G. Sebastiani, and A. H. van den Boogaard, "The technology of Incremental Sheet Forming-A brief review of the history," J. Mater. Process. Technol., vol. 210, no. 8, pp. 981–997, 2010, doi:10.1016/j.jmatprotec.2010.02.014.

M. Fiorotto, M. Sorgente, and G. Lucchetta, "Preliminary studies on single point incremental forming for composite materials," Int. J. Mater. Form., vol. 3, no. Suppl. 1, pp. 951–954, 2010, doi:10.1007/s12289-010-0926-6.

P. A. F. Martins, N. Bay, M. Skjoedt, and M. B. Silva, "Theory of single point incremental forming," CIRP Ann. - Manuf. Technol., vol. 57, no. 1, pp. 247–252, 2008,

doi:10.1016/j.cirp.2008.03.047.

T. J. Kim and D. Y. Yang, "Improvement of formability for the incremental sheet metal forming process," Int. J. Mech. Sci., vol. 42, no. 7, pp. 1271–1286, 2000, doi:10.1016/S0020-7403(99)00047-8.

H. Iseki, "An approximate deformation analysis and FEM analysis for the incremental bulging of sheet metal using a spherical roller," J. Mater. Process. Technol., vol. 111, no. 1–3, pp. 150–154, 2001, doi:10.1016/S0924-0136(01)00500-3.

M. S. Shim and J. J. Park, "The formability of aluminum sheet in incremental forming," J. Mater. Process. Technol., vol. 113, no. 1–3, pp. 654–658, 2001, doi:10.1016/S0924-0136(01)00679-3.
G. Hirt, J. Ames, M. Bambach, R. Kopp, and R. Kopp, "Forming strategies and process modelling for CNC incremental sheet forming," CIRP Ann. - Manuf. Technol., vol. 53, no. 1, pp. 203–206, 2004, doi:10.1016/S0007-8506(07)60679-9.

L. Filice, L. Fratini, and F. Micari, "Analysis of material formability in incremental forming," CIRP Ann. - Manuf. Technol., vol. 51, no. 1, pp. 199–202, 2002, doi:10.1016/S0007-8506(07)61499-1.

D. Young and J. Jeswiet, "Wall thickness variations in single-point incremental forming," Proc. Inst. Mech. Eng. Part B J. Eng. Manuf., vol. 218, no. 11, pp. 1453–1459, 2004, doi:10.1243/0954405042418400.

J. Jeswiet, F. Micari, G. Hirt, A. Bramley, J. Duflou, and J. Allwood, "Asymmetric single point incremental forming of sheet metal," CIRP Ann. - Manuf. Technol., vol. 54, no. 2, pp. 88–114, 2005, doi:10.1016/s0007-8506(07)60021-3.

J. Jeswiet and D. Young, "Forming limit diagrams for single-point incremental forming of aluminium sheet," Proc. Inst. Mech. Eng. Part B J. Eng. Manuf., vol. 219, no. 4, pp. 359–364, 2005, doi:10.1243/095440505X32210.

M. Ham and J. Jeswiet, "Single point incremental forming and the forming criteria for AA 3003," CIRP Ann. - Manuf. Technol., vol. 55, no. 1, pp. 241–244, 2006, doi:10.1016/S0007-8506(07)60407-7.

M. Ham and J. Jeswiet, "Forming limit curves in single point incremental forming," CIRP Ann. -Manuf. Technol., vol. 56, no. 1, pp. 277–280, 2007, doi:10.1016/j.cirp.2007.05.064.

W. C. Emmens and A. H. van den Boogaard, "Strain in Shear, and Material Behaviour in Incremental Forming," Key Eng. Mater., vol. 344, pp. 519–526, 2007, doi:10.4020/www.aciontific.net/loom.244.510

doi:10.4028/www.scientific.net/kem.344.519.

J. M. Allwood, D. R. Shouler, and A. E. Tekkaya, "The Increased Forming Limits of Incremental Sheet Forming Processes," Key Eng. Mater., vol. 344, pp. 621–628, 2007, doi:10.4028/www.scientific.net/kem.344.621.

J. M. Allwood and D. R. Shouler, "Generalised forming limit diagrams showing increased forming limits with non-planar stress states," Int. J. Plast., vol. 25, no. 7, pp. 1207–1230, 2009, doi:10.1016/j.ijplas.2008.11.001.

M. B. Silva, M. Skjoedt, P. A. F. Martins, and N. Bay, "Revisiting the fundamentals of single point incremental forming by means of membrane analysis," Int. J. Mach. Tools Manuf., vol. 48, no. 1, pp. 73–83, 2008, doi:10.1016/j.ijmachtools.2007.07.004.

M. B. Silva, M. Skjoedr, A. G. Atkins, N. Bay, and P. A. F. Martins, "Single-point incremental forming and formability-failure diagrams," J. Strain Anal. Eng. Des., vol. 43, no. 1, pp. 15–35, 2008, doi:10.1243/03093247JSA340.

K. Jackson and J. Allwood, "The mechanics of incremental sheet forming," J. Mater. Process. Technol., vol. 209, no. 3, pp. 1158–1174, 2009, doi:10.1016/j.jmatprotec.2008.03.025.
M. Bambach, "A geometrical model of the kinematics of incremental sheet forming for the prediction of membrane strains and sheet thickness," J. Mater. Process. Technol., vol. 210, no. 12, pp. 1562–1573, 2010, doi:10.1016/j.jmatprotec.2010.05.003.

Z. Cui, Z. Cedric Xia, F. Ren, V. Kiridena, and L. Gao, "Modeling and validation of deformation process for incremental sheet forming," J. Manuf. Process., vol. 15, no. 2, pp. 236–241, 2013, doi:10.1016/j.jmapro.2013.01.003.

Z. Liu, W. J. T. Daniel, Y. Li, S. Liu, and P. A. Meehan, "Multi-pass deformation design for incremental sheet forming: Analytical modeling, finite element analysis and experimental validation," J. Mater. Process. Technol., vol. 214, no. 3, pp. 620–634, 2014, doi:10.1016/j.jmatprotec.2013.11.010.

Y. Li, W. J. T. Daniel, and P. A. Meehan, "Deformation analysis in single-point incremental forming through finite element simulation," Int. J. Adv. Manuf. Technol., vol. 88, no. 1–4, pp.

255–267, 2017, doi:10.1007/s00170-016-8727-9.

S. Ai, B. Lu, J. Chen, H. Long, and H. Ou, "Evaluation of deformation stability and fracture mechanism in incremental sheet forming," Int. J. Mech. Sci., vol. 124–125, pp. 174–184, 2017, doi:10.1016/j.ijmecsci.2017.03.012.

S. Ai, R. Dai, and H. Long, "Investigating formability enhancement in double side incremental forming by developing a new test method of tension under cyclic bending and compression," J. Mater. Process. Technol., vol. 275, no. July 2019, p. 116349, 2020, doi:10.1010/j.imstructure.2010.116240

doi:10.1016/j.jmatprotec.2019.116349.

F. Maqbool and M. Bambach, "Revealing the Dominant Forming Mechanism of Single Point Incremental Forming (SPIF) by Splitting Plastic Energy Dissipation," Procedia Eng., vol. 183, pp. 188–193, 2017, doi:10.1016/j.proeng.2017.04.018.

K. U. Yazar, S. Mishra, K. Narasimhan, and P. P. Date, "Deciphering the deformation mechanism in single point incremental forming: experimental and numerical investigation," Int. J. Adv. Manuf. Technol., vol. 101, no. 9–12, pp. 2355–2366, 2019, doi:10.1007/s00170-018-3131-2.

V. Franzen, L. Kwiatkowski, P. A. F. Martins, and A. E. Tekkaya, "Single point incremental forming of PVC," J. Mater. Process. Technol., vol. 209, no. 1, pp. 462–469, 2009, doi:10.1016/j.jmatprotec.2008.02.013.

P. A. F. Martins, L. Kwiatkowski, V. Franzen, A. E. Tekkaya, and M. Kleiner, "Single point incremental forming of polymers," CIRP Ann. - Manuf. Technol., vol. 58, no. 1, pp. 229–232, 2009, doi:10.1016/j.cirp.2009.03.095.

M. B. Silva, L. M. Alves, and P. A. F. Martins, "Single point incremental forming of PVC: Experimental findings and theoretical interpretation," Eur. J. Mech. A/Solids, vol. 29, no. 4, pp. 557–566, 2010, doi:10.1016/j.euromechsol.2010.03.008.

M. B. Silva, T. Marques, and P. A. F. Martins, "Single-point incremental forming of polymers," Mechatronics Manuf. Eng., pp. 293–331, Jan. 2012, doi: 10.1533/9780857095893.293.

L. V. Sy and N. T. Nam, "A numerical simulation of incremental forming process for polymer sheets," Int. J. Model. Simul., vol. 32, no. 4, pp. 265–272, 2012,

doi:10.2316/Journal.205.2012.4.205-5752.

S. Alkas Yonan, C. Soyarslan, P. Haupt, L. Kwiatkowski, and A. E. Tekkaya, "A simple finite strain non-linear visco-plastic model for thermoplastics and its application to the simulation of incremental cold forming of polyvinylchloride (PVC)," Int. J. Mech. Sci., vol. 66, pp. 192–201, 2013, doi:10.1016/j.ijmecsci.2012.11.007.

S. Alkas Yonan, M. B. Silva, P. A. F. Martins, and A. E. Tekkaya, "Plastic flow and failure in single point incremental forming of PVC sheets," Express Polym. Lett., vol. 8, no. 5, pp. 301–311, 2014, doi:10.3144/expresspolymlett.2014.34.

I. Bagudanch, G. Centeno, C. Vallellano, and M. L. Garcia-Romeu, "Revisiting formability and failure of polymeric sheets deformed by Single Point Incremental Forming," Polym. Degrad. Stab., vol. 144, pp. 366–377, 2017, doi:10.1016/j.polymdegradstab.2017.08.021.

M. Durante, A. Formisano, and F. Lambiase, "Incremental forming of polycarbonate sheets," J. Mater. Process. Technol., vol. 253, no. September 2017, pp. 57–63, 2018, doi:10.1016/j.jmatprotec.2017.11.005.

M. Durante, A. Formisano, and F. Lambiase, "Formability of polycarbonate sheets in singlepoint incremental forming," Int. J. Adv. Manuf. Technol., vol. 102, no. 5–8, pp. 2049–2062, 2019, doi:10.1007/s00170-019-03298-w.

L. M. Lozano-Sánchez et al., "Mechanical and structural studies on single point incremental forming of polypropylene-MWCNTs composite sheets," J. Mater. Process. Technol., vol. 242, pp. 218–227, 2017, doi:10.1016/j.jmatprotec.2016.11.032.

A. AL-Obaidi, A. Kunke, and V. Kräusel, "Hot single-point incremental forming of glass-fiberreinforced polymer (PA6GF47) supported by hot air," J. Manuf. Process., vol. 43, no. May, pp. 17–25, 2019, doi:10.1016/j.jmapro.2019.04.036.

M. Hassan, G. Hussain, H. Wei, A. Qadeer, and M. AlKahtani, "Progress on single-point incremental forming of polymers," Int. J. Adv. Manuf. Technol., vol. 114, no. 1–2, 2021, doi:10.1007/s00170-021-06620-7.

Sustainable Polishing of Directed Energy Deposition–Based Cladding Using Micro-Plasma Transferred Arc

ISO/ASTM 52900:2015 (ASTM F2792) Additive manufacturing - General principles – Terminology. https://www.iso.org/standard/69669.html

Kumar, P., Jain, N. K., Sawant, M. S. (2020), Modeling of dimensions and investigations on geometrical deviations of metallic components manufactured by μ-plasma transferred arc additive manufacturing process, The International Journal of Advanced Manufacturing Technology, 107, 3155–3168. https://doi.org/10.1007/s00170-020-05218-9

Gharbi, M., Peyre, P., Gorny, C., Carin, M., Morville, S., Le Masson, P., Fabbro, R. (2013), Influence of various process conditions on surface finishes induced by the direct metal deposition laser technique on a Ti–6Al–4V alloy, Journal of Materials Processing Technology, 213, 791–800. https://doi.org/10.1016/j.jmatprotec.2012.11.015

DebRoy, T., Wei, H. L., Zuback, J. S., Mukherjee, T., Elmer, J. W., Milewski, J. O., Beese, A. M., Wilson-Heid, A. D., De, A., Zhang, W. (2018), Additive manufacturing of metallic components–process, structure and properties, Progress in Materials Science, 92, 112–224. https://doi.org/10.1016/j.pmatsci.2017.10.001

Thivillon, L., Bertrand, P., Laget, B., Smurov, I. (2009), Potential of direct metal deposition technology for manufacturing thick functionally graded coatings and parts for reactors components, Journal of Nuclear Materials, 385, 236–241.

https://doi.org/10.1016/j.jnucmat.2008.11.023

Kim, M. S., Park, S. H., Pyun, Y. S., Shim, D. S. (2020), Optimization of ultrasonic nanocrystal surface modification for surface quality improvement of directed energy depositd stainless steel 316L, Journal of Materials Research and Technology, 9, 15102–15122. https://doi.org/10.1016/j.jmrt.2020.10.092

Tan, K. L., Yeo, S. H. (2017), Surface modification of additive manufactured components by ultrasonic cavitation abrasive finishing, Wear, 378, 90–95.

http://dx.doi.org/10.1016/j.wear.2017.02.030

Bai, Y., Chaudhari, A., Wang, H. (2020), Investigation on the microstructure and machinability of ASTM A131 steel manufactured by directed energy deposition, Journal of Materials Processing Technology, 276, 116410. https://doi.org/10.1016/j.jmatprotec.2019.116410

Oyelola, O., Crawforth, P., M'Saoubi, R., Clare, A. T. (2018), Machining of functionally graded Ti6Al4V/WC produced by directed energy deposition, Additive Manufacturing, 24, 20–29. https://doi.org/10.1016/j.addma.2018.09.007

Jaritngam, P., Tangwarodomnukun, V., Qi, H., Dumkum, C. (2020), Surface and subsurface characteristics of laser polished Ti6Al4V titanium alloy, Optics and Laser Technology, 126, 106102. https://doi.org/10.1016/j.optlastec.2020.106102

Rosa, B., Mognol, P., Hascoët, J. Y. (2015), Laser polishing of additive laser manufacturing surfaces, Journal of Laser Applications, 27, S29102. https://doi.org/10.2351/1.4906385 Ma, C. P., Guan, Y. C., Zhou, W. (2017), Laser polishing of additive manufactured Ti alloys, Optics and Lasers in Engineering, 93, 171–177. https://doi.org/10.1016/j.optlaseng.2017.02.005 Proskurovsky, D. I., Rotshtein, V. P., Ozur, G. E., Markov, A. B., Nazarov, D. S., Shulov, V. A., Buchheit, R. G. (1998), Pulsed electron-beam technology for surface modification of metallic materials, Journal of Vacuum Science & Technology A: Vacuum, Surfaces, and Films, 16, 2480–2488. https://doi.org/10.1116/1.581369

Zhu, X. P., Lei, M. K., Ma, T. C. (2003), Surface morphology of titanium irradiated by highintensity pulsed ion beam, Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions with Materials and Atoms, 211, 69–79. https://doi.org/10.1016/S0168-583X(03)01124-8

Townsend, P. D. (1970), Ion beams in optics—an introduction, Optics Technology, 2, 65–67. https://doi.org/10.1016/0374-3926(70)90003-7

Jie, X. U., Wang, C. J., Bin, G. U. O., Shan, D. B., Sugiyama, Y., Ono, S. (2009), Surface finish of micro punch with ion beam irradiation, Transactions of Nonferrous Metals Society of China, 19, 526–530. https://doi.org/10.1016/S1003-6326(10)60102-1

Yamada, I. (1999), Low-energy cluster ion beam modification of surfaces, Nuclear Instruments and Methods in Physics Research Section B: Beam Interactions, Materials and Atoms, 148, 1–11. https://doi.org/10.1016/S0168-583X(98)00875-1

Kumar, P., Jain, N. K. (2020), Effect of material form on deposition characteristics in microplasma transferred arc additive manufacturing process, CIRP Journal of Manufacturing Science and Technology, 30, 195–205. https://doi.org/10.1016/j.cirpj.2020.05.008

Sawant, M. S., Jain, N. K. (2017), Characteristics of single-track and multi-track depositions of Stellite by micro-plasma transferred arc powder deposition process, Journal of Materials Engineering and Performance, 26, 4029–4039. https://doi.org/10.1007/s11665-017-2828-y